Designing Electricity Distribution Network Tariffs for Beneficial Electrification

by

Graham Turk

B.S.E. Computer Science, Princeton University, 2017

Submitted to the Institute for Data, Systems, and Society in partial fulfillment of the requirements for the degree of

MASTER OF SCIENCE IN TECHNOLOGY AND POLICY

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

May 2024

(c) 2024 Graham Turk. This work is licensed under a CC BY-NC-ND 4.0 license.

The author hereby grants to MIT a nonexclusive, worldwide, irrevocable, royalty-free license to exercise any and all rights under copyright, including to reproduce, preserve, distribute and publicly display copies of the thesis, or release the thesis under an open-access license.

Authored by:	Graham Turk Institute for Data, Systems, and Society May 10, 2024
Certified by:	Pablo Duenas-Martinez Research Scientist, MIT Energy Initiative, Thesis Supervisor
Certified by:	Paul L. Joskow Elizabeth and James Killian Professor of Economics, Emeritus MIT Department of Economics, Thesis Supervisor
Accepted by:	Frank R. Field III Senior Research Engineer, Sociotechnical Systems Research Interim Director, Technology and Policy Program

Designing Electricity Distribution Network Tariffs for Beneficial Electrification

by

Graham Turk

Submitted to the Institute for Data, Systems, and Society on May 10, 2024 in partial fulfillment of the requirements for the degree of

MASTER OF SCIENCE IN TECHNOLOGY AND POLICY

ABSTRACT

Decarbonizing the transportation and residential building sectors will require rapid electrification through the uptake of electric vehicles (EVs) and cold climate heat pumps (CCHPs), respectively. There is broad consensus that the flat volumetric electricity tariffs currently in place for residential customers in most of the US discourage electrification and do not reflect the underlying marginal costs of electricity delivery. Under flat volumetric tariffs, utilities are projecting sharp rises in distribution-level peak demand, which will necessitate network upgrades whose costs are recovered from all grid users. Alternative rate designs can help mitigate the need for these upgrades by shifting new demand away from peak periods. However, there is an emerging narrative that electricity tariff design is a zero-sum game: regulators can either protect vulnerable households or encourage electrification, but not both. In this thesis, we challenge that perception by asking whether well-designed distribution network tariffs can deliver a win-win in the long run, reducing operating costs for EVs and/or CCHPs and average network costs for households that cannot yet afford to electrify. We answer this question by running a series of bottom-up optimizations to simulate household's responses to alternative network tariff designs in two distinct geographies, then assessing cost impacts on different household groups. We use open-source data on household electricity consumption and travel behavior. We find that beyond very low adoption levels, time-of-use per-kWh network tariffs, which several states have adopted as the default, perform poorly on all metrics and lead to large increases in local peak demand. Per-kW capacity tariffs (subscription and demand charges) are effective at mitigating EV-driven peaks, especially when paired with TOU energy tariffs. We recommend that regulators separate network charges from energy charges and introduce a per-kW subscription network tariff to collect a portion of the network revenue requirement. This approach will reduce the total cost of ownership of electrified devices while mitigating the network upgrades needed to maintain reliability. Our recommendations offer a path towards rapid electrification that benefits all grid users.

Thesis supervisor: Pablo Duenas-Martinez Title: Research Scientist, MIT Energy Initiative

Thesis supervisor: Paul L. Joskow Title: Elizabeth and James Killian Professor of Economics, Emeritus MIT Department of Economics

Acknowledgments

This thesis is the culmination of two incredible years at MIT. I am indebted to my research advisors Pablo Duenas-Martinez and Tim Schittekatte for their expert guidance, support, and feedback every step of the way, and for handling the behind-the-scenes logistics so I could focus on research. A special thank you to Paul Joskow and Richard Schmalensee; our conversations and email threads have been an education in electricity regulation and rate design. Paul and Pablo provided invaluable comments on earlier drafts of this thesis. I could not have asked for better research collaborators in Abdelrahman Ayad and Zachary Schmitz. I am grateful to Elena Byrne, Barb DeLaBarre, Frank Field, and Noelle Selin for their tireless dedication to cultivating a close-knit TPP community. To Abby, Lauren, Mike, Luc, Peter, Ben, Johnattan, and Nirmal: thanks for letting this TPP elder feel young again. Finally, to my parents, Jon and Carolyn: thank you for encouraging me to go back to school and the constant love and support from grades K-18. Though I clearly did not heed your advice to omit needless words, this thesis is dedicated to you.

Contents

Ti	itle page	1		
Al	bstract	3		
Ac	Acknowledgments 5			
Li	st of Figures	11		
Li	st of Tables	13		
1	Introduction: Electrification's Role in Meeting Climate Targets1.1Current Trends in US Electricity Networks1.2How Rate Design Impacts Electrification1.3Central Question and Thesis Outline	15 16 16 20		
2	 Background on the Electric Power Sector and Cost of Service Regulation 2.1 The Electric Power Sector in the US 2.1.1 Ownership Structure 2.1.2 Level of Vertical Integration 2.2 Origins of Regulation 2.3 Cost of Service Regulation and Rate Cases 2.4 Limitations of Cost of Service Regulation and Alternatives 2.5 The Link Between Rate Design and Network Investments 	 23 23 24 25 26 27 28 		
3	Literature Review and Research Contribution 3.1 Impacts of Electrification Under Flat Volumetric Tariffs 3.1.1 Heat Pump Impacts 3.1.2 Electric Vehicle Impacts 3.1 Second Provide	31 31 32 33 35 36 37		
4	Distribution Network Tariff Design in Practice: Approaches from the USand Europe4.1Distribution Network Tariff Design in the US	39 39		

		4.1.1 Default Time-of-Use Tariffs	39
		4.1.2 Technology-Specific Tariffs	41
	4.2	Distribution Network Tariff Design in Europe	43
5	Met	thodology: Simulating Residential Electrification Under Distribution	
	Net	work Tariff Alternatives	45
	5.1	Data Sources	46
		5.1.1 Household Energy Consumption Profiles	46
		5.1.2 Vehicle Driving Profiles	47
		5.1.3 Distribution Network Topology	48
	5.2	Tariff Price Calculation	49
		5.2.1 Fixed Charges	49
		5.2.2 Per-kWh Charges	49
		5.2.3 Capacity Charges	50
	5.3	Analytical Methods	52
		5.3.1 Optimization Formulations	52
		5.3.2 Feedback Loop for Recovering Network Investments and Recomputing	
		Charges	54
	5.4	Case Study Outlines	57
		5.4.1 Greensboro (North Carolina): Price-Responsive EV Charging in a	
		Warm-Climate Geography with a Detailed Network Model	57
		5.4.2 Massachusetts: Price-Responsive EV and CCHP Operation in a Cold-	•
		Climate Geography with a Simplified Network Model	58
	5.5	Key Metrics	59
6	Res	sults	63
	6.1	Results for Greensboro Case Study	63
		6.1.1 Greensboro: Annual Peak	63
		6.1.2 Greensboro: Levelized EV Charging Cost	68
		6.1.3 Greensboro: Difference in Network Costs for Non-EV Households	70
	6.2	Results for Massachusetts Case Study	73
		6.2.1 Massachusetts: Annual Peak	73
		6.2.2 Massachusetts: Levelized EV/CCHP Costs	75
		6.2.3 Massachusetts: Difference in Network Costs for Non-Electrified House-	
		m holds	76
7	Dis	cussion	78
	7.1	Assessment of Model Results	78
	7.2	Tariff Design Considerations	79
		7.2.1 Addressing the Apparent Benefits of Flat Volumetric Tariffs	81
		7.2.2 Subscription versus Demand Charges	82
		7.2.3 Direct Load Control	84
		7.2.4 Hybrid Heating Systems and Gas Network Cost Recovery	85
		7.2.5 Balancing Stakeholder Interests for Efficiency and Solar	87
	7.3	Limitations of Methodology	88

8	Conclusion and Policy Recommendations	91
Re	ferences	95

List of Figures

1.1	Difference in simplified 5-year total ownership cost between an EV and a	1 7
1.2	Example residential electricity bill with major charge categories labeled	17 18
4.1	Illustration of rebound peak due to EV charging	41
$5.1 \\ 5.2$	Discrete distribution used to select departure and arrival hour for driving profiles Illustrative flowchart of methodology	$\begin{array}{c} 48\\ 56 \end{array}$
6.1	Annual peak demand under each tariff scenario at 5% EV adoption increments (Greensboro)	64
6.2	Aggregate average weekday demand at 50% EV adoption under the flat and TOU per-kWh tariff scenarios (Greensboro)	65
6.3	EV charging behavior under a TOU per-kWh tariff versus a subscription tariff at the individual household and aggregate levels (Greensboro)	66
6.4	Annual peak demand at 100% and 70% price signal compliance with tariff price signals (Greensboro)	67
$\begin{array}{c} 6.5 \\ 6.6 \end{array}$	Levelized charging costs for EV owners (Greensboro)	69
67	tariff scenario (Greensboro)	$70 \\ 71$
6.8	Distribution in change of network costs paid by non-EV households for the	/ I
6.9	Aggregate demand by load type for two representative weeks at 50% electri-	72
6.10	fication (Massachusetts)	$73 \\ 74$
6.11	Load duration curves at 50% electrification for the Greensboro and Mas-	75
6.12	2 Levelized cost of EV and CCHP operation at 5% electrification adoption in-	10
6.13	crements (Massachusetts)	76
	tion increments (Massachusetts)	77
7.1	Annual peak demand in the Massachusetts case study with gas backup $\ . \ . \ .$	86

List of Tables

1.1	Description of major categories of charges on residential electricity bills	19
3.1	Classification of network tariff design studies with an application to electrifi- cation	38
4.1 4.2 4.3	Default TOU rates approved in the US	40 42 43
$5.1 \\ 5.2 \\ 5.3$	Summary of distribution network tariff types with illustrative prices Tariffs tested in the Greensboro case study	50 61 62
7.1 7.2	Key metrics in the Greensboro case study for each tariff at 50% EV adoption for low and high LRMC cases	78 79
8.1	Summary of key insights and policy recommendations	94

Chapter 1

Introduction: Electrification's Role in Meeting Climate Targets

Transportation and buildings are responsible for 41% of direct greenhouse gas emissions in the US [1]. To decarbonize these sectors, there is widespread consensus that electrification – replacing fossil-fueled vehicles and heating equipment with electric alternatives – offers the most cost-effective and scalable pathway [2] [3]. In transportation, where 58% emissions come from light-duty passenger vehicles [4], electrification is synonymous with the purchase of electric vehicles (EVs). For residential buildings, where space heating is the number one source of direct emissions [1], electrification requires installing cold climate heat pumps (CCHPs), which provide both heating and cooling and replace or supplement conventional boilers and furnaces fueled by natural gas, propane, or oil.

While electrification does not eliminate emissions completely, both EVs and CCHPs offer a significant reduction in emissions compared to fossil-fueled equipment [5] [6]. This emissions gap will improve as the electricity generation mix becomes cleaner, which many states have mandated through renewable portfolio standards (RPS). In addition to RPS statutes, there are numerous policy drivers for electrification. The Inflation Reduction Act, passed in 2022, provides purchase incentives for both EVs and CCHPs that reduce consumers' up-front costs. Several states offer purchase rebates that can be stacked on top of IRA funds. Furthermore, in 2020 California's governor issued an executive order (N-79-20) that all new passenger cars, trucks, and SUVs sold after 2035 must be EVs, which was codified in regulation approved by the California Air Resources Board in 2022. Nine other states, which have historically followed California's vehicle emissions standard, plan to follow suit and prohibit the sale of new internal combustion engine vehicles after 2035 [7].

Both EVs and CCHPs consume a significant amount of electricity; a typical EV uses 2,363 kWh per year for at-home charging [8]. CCHP consumption varies widely depending on climate; in cold regions, electric heating is expected to increase a home's electricity consumption by over 6,000 kWh per year [9]. Taken together, electrifying the light duty vehicle fleet and residential housing stock is projected to nearly double annual residential electricity consumption [2]. As states aim to substitute electricity as the primary "fuel" to power our vehicles and condition our buildings, the electricity grid has emerged as the backbone for decarbonization efforts.

1.1 Current Trends in US Electricity Networks

While the cost of generating electricity in the US has consistently fallen over the past 20 years, driven by cheap natural gas and the declining cost of renewables, the cost of delivering electricity has steadily grown [10]. Electricity utilities build and maintain a complex network of poles, wires, and power electronics equipment to send electricity over long distances to every building connected to the grid. The size of this network is determined in large part by the maximum (or "peak") electricity demand at each network level. Peak demand is correlated with extreme temperatures and diurnal patterns; for example, the 2023 peak for the New England region occurred on September 7 at 5:00 PM, with an average temperature of 88°F and 70°F dew point [11], close to the end of the workday as millions of New Englanders returned home and turned on air conditioning units.

In many parts of the US, peak demand has tapered [12]. However, electrification – combined with growth in data center construction and clean energy manufacturing – is expected to reverse this trend [13]. If EV owners plug in their vehicles upon returning home and heat pumps operate unmitigated during the hottest and coldest days of the year, peak demand is projected to grow rapidly [14][15]. This is expected to impact various portions of the electricity grid, from aggregate peak demand at the generation and high-voltage transmission levels down to peak demand on various distribution feeders and "semicollective" groups of feeders [16].

To meet the expected growth in peak demand, electricity utilities are already making plans to expand delivery capacity. For example, in Massachusetts, the two largest investorowned utilities – Eversource and National Grid – recently filed plans to invest over \$10B for distribution infrastructure upgrades to support electrification [14][15]. Demand-driven investment costs will be passed through to grid users via electricity rates, which may exacerbate energy burdens and (paradoxically) disincentivize electrification [17].

In utility plans and in the modeling literature, a narrative is emerging that rapid peak demand growth is an inevitable consequence of electrification and that electrification, as a result, will increase electricity rates. However, this is only one possible path. Electrification, if managed to efficiently utilize existing infrastructure, can apply downward rate pressure. We refer to this phenomenon as "beneficial electrification." ¹ One study by the California Public Utility Commission found that "the increase in electricity sales from electrification may outweigh the costs of distribution investments," a benefit to not just those who adopt EVs and CCHPs but all grid users [18]. Yet the structure of electricity rates today is not conducive to this beneficial outcome.

1.2 How Rate Design Impacts Electrification

When households consider whether to adopt EVs and CCHPs, one determinant is the total cost of ownership compared to conventional fossil fuel appliances [19][20]. If a household cannot save money over a reasonable time period, they will have little incentive to electrify.

¹While "beneficial electrification" is often used to describe the broad set of environmental and cost benefits associated with electrifying various end uses, we define it narrowly to mean electrification that achieves a reduction in the average cost of electricity for all network users.

	Chevrolet Bolt (EV)	Chevrolet Equinox (ICE)
MSRP	\$33,000	\$24,000
IRA Tax Incentive	-\$7,500	
5-year Fuel/Electricity Cost	\$2,990	\$6,400
Total	\$28,490	\$30,400

Figure 1.1: Difference in simplified 5-year total ownership cost between an EV and a comparable internal combustion engine vehicle. Maintenance costs are omitted.

Total cost of ownership consists of the cost of the equipment itself (inclusive of rebates), installation, and the cost of operating the equipment over its lifetime. EVs today cost more up front than conventional gasoline vehicles but experience lower operating costs due to the positive gap between the cost of electricity and gasoline [21]. Figure 1.1 shows a simple comparison of the 5-year ownership costs of an EV versus a comparable internal combustion engine vehicle driven for 12,000 miles per year, assuming manufacturer-listed prices and efficiencies [22], using US nationwide average gas (\$3.66/gallon) [23] and electricity (\$0.15/kWh) [24] prices as of January 2024.

CCHP systems have already achieved cost parity with conventional heating and cooling solutions for *new* construction [25]. But the economics are less favorable for retrofit installations, especially in areas connected to natural gas infrastructure where electricity costs are relatively high. In Massachusetts, for example, at current electricity and gas prices, it is less expensive to heat with natural gas once the outdoor temperature falls below 35F [26], and a household switching to all-electric heating can expect to spend more per year in operating costs [27]. With a negative cost gap, heat pump installations are far behind state targets [28]. By contrast, in Maine, which has a similar (if not less favorable) climate but where the dominant heating fuel is oil, over 100,000 CCHP systems have been installed, exceeding the state's ambitious targets [29]. These examples and prior research (e.g., [19]) provide evidence that electricity cost is a key determinant for CCHP adoption.

If we accept that the cost of electricity will play a central role in influencing the speed of electrification, we need to consider how consumers pay for electricity. Figure 1.2 shows a standard residential bill. The costs are broken into several categories – distribution, transmission, supply, etc. - described in Table 1.1.

Distribution and transmission together represent the cost of delivering electricity, as opposed to generating electricity, which is represented by the generation service charge (Category 1), and other policy costs (Category 5), which pay for policy-driven initiatives like renewable energy deployment and energy efficiency.

In this thesis, we focus on the distribution portion of the bill, which accounts for 26% of customer costs on average [30]. For investor-owned utilities (IOUs), which serve 72% of US electricity customers [31], distribution costs are regulated by each state's public utility commission (PUC), detailed in Chapter 2. In most of the US, distribution network costs are recovered from residential customers through "flat volumetric" electricity rates, or tariffs.

Meter		
Generation Service Charge (1)	172 kWh X .14810	\$25.47
Subtotal Supplier Services		\$25.47
Delivery		
(Rate 01 R1 RESIDENTIAL)		
Meter		
Customer Charge (2)		\$10.00
Distribution Charge 3	172 kWh X .09434	\$16.23
Transition Charge	172 kWh X00037	-\$0.06
Transmission Charge (4)	172 kWh X .04052	\$6.97
Revenue Decoupling Charge	172 kWh X .00006	\$0.01
Distributed Solar Charge	172 kWh X .00800	\$1.38
Renewable Energy Charge $\left. \right\rangle$ (5)	172 kWh X .00050	\$0.09
Energy Efficiency	172 kWh X .02334	\$4.01
Subtotal Delivery Services		\$38.63
Total Cost of Electricity		\$64.10

Supplier (CAMBRIDGE COMM ELEC DIRECT ENERGY)

....

Figure 1.2: Example residential electricity bill (the author's bill), with major charge categories labeled. We omit consideration of the transition and revenue decoupling charges, which are not universal among US utilities and account for a very small portion of total costs.

Category	Charge(s)	What the charge pays for
1	Generation Service	The cost of generating electricity, sometimes called the supply charge
2	Customer	Fixed costs including overhead (salaries, of- fice space, etc) and metering equipment
3	Distribution	The cost of building and operating the low- voltage distribution grid
4	Transmission	The cost of building and operating the high- voltage transmission grid
5	Distributed Solar Renewable Energy Energy Efficiency	Public policy programs to promote conserva- tion and deployment of clean energy

Table 1.1: Description of major categories of charges on residential electricity bills

Flat meaning that the price is the same at all hours (i.e., time-invariant) and volumetric meaning that consumers pay based on their total consumption (per-kWh) over the course of a billing period.² Under flat volumetric rates, someone who consumes 200 kWh in a month will pay twice as much as someone who consumes 100 kWh. However, the costs underlying those charges depend on other factors besides total monthly consumption. Many utilities' distribution costs are fixed in the short run, reflecting past investments, and in the longer run are driven by the need to make incremental investments to handle periods of higher kW demand [32].

This misalignment between the drivers of increasing network costs and how those costs are recovered from consumers presents two major problems:

- 1. Under flat volumetric network tariffs, consumers do not receive any information about when the network is congested. This makes it more likely that newly electrified load (e.g., from EVs and CCHPs) will trigger network investments, whose costs are borne by all consumers in the form of higher rates.
- 2. Flat volumetric network tariffs provide no opportunity for consumers to reduce their costs by shifting the timing of consumption to periods when the network has excess capacity and marginal delivery costs are negligible. This has a negative impact on the economics of electrification.

For many years, a flat volumetric tariff was the only option for utilities to collect money from customers. Analog electricity meters measure only total consumption, not the timing of that consumption. But advanced meters can measure and communicate consumption at subhourly intervals, enabling "time-varying" rates where the price to consume electricity changes

 $^{^{2}}$ We use the terms "volumetric" and "per-kWh" interchangeably. Flat volumetric tariffs are also widely used to collect transmission and generation costs, but we focus only on the design of distribution network tariffs in this thesis.

depending on the hour, day, and/or season. Internet-connected devices, like EV charging stations and smart thermostats, make it simple for consumers to schedule consumption in response to these rates. Today, while approximately 73% of residential customers in the US have advanced meters [33], only 10% are enrolled on time-varying tariffs [34].

There is broad consensus that flat volumetric network tariffs are not cost-reflective [32] and discourage electrification [35].³ There is a significant body of research recommending alternatives that more accurately reflect the true cost of consumption; as technology has improved, these proposals have grown increasingly complex, relying on such mechanisms as real-time auctions and autonomous home energy management agents. Yet US utilities and regulators have largely ignored these proposals; flat volumetric rates continue to prevail. This gap between what economists agree to be theoretically optimal rate designs and how rates are designed in practice can be partially explained by the regulatory framework dominant in the US today. Under cost-of-service regulation, IOUs have no incentive to offer alternative rates (discussed in Chapter 2). And regulators are hesitant to impose changes that might confuse customers or create large winners and losers compared to the status quo. This barrier highlights the need for simple solutions that consumers understand, that offer protections against large bill increases, and that do not rely on utilities to be active participants in managing load (against their own financial interest).

PUCs in California, Hawaii, Missouri, Colorado, and Michigan have illuminated a path forward by mandating that utilities offer per-kWh time-of-use (TOU) rates as the default. We review these initiatives in Chapter 4. This is a step in the right direction towards reflecting the underlying marginal cost of electricity delivery. But while TOU rates can improve the short-to-medium term economics of electrification, they may actually exacerbate peak demand growth and subsequent network investment. Other initiatives in the US to reduce the cost of electrification, most notably California's Income-Graduated Fixed Charge (IGFC), have been met with fierce opposition by solar and energy efficiency advocates. If we look outside the US, primarily in Europe, we find alternative approaches to tariff design that have the potential to reduce the marginal cost of electricity for consumers electrifying their heating and transportation and help defer network upgrades, which delivers benefits to all grid users.

1.3 Central Question and Thesis Outline

Under flat volumetric rates prevalent today, utilities are projecting significant network investments due to electrification, especially for the distribution grid. These investments will be recovered from all grid users through electricity rates. An increase in the cost of electricity will not only raise energy burdens broadly, but also threaten the positive gap (or

³Unless otherwise stated, the term "network tariff" in this thesis refers exclusively to the distribution network tariff. Despite our narrow focus, many of the proposed tariff-based solutions to deferring or avoided distribution investments can also be leveraged for transmission and generation constraints. A recent report by DOE projects a shortfall in generating capacity of 200 GW by 2030 to meet nationwide peak load [36]. Demand flexibility (i.e., devices acting in response to well-design tariffs) can help meet this gap; a recent study by Lawrence Berkeley National Lab found that price-based demand response is the most cost effective planning option for bulk system reliability [37].

exacerbate the negative cost gap) between electricity and fossil fuels, which will risks slowing decarbonization progress. Attempts to reform electricity rates in the US to promote electrification have primarily addressed symptoms of poor rate design, not their underlying causes, and have faced opposition from ratepayer advocates and consumer protection organizations. This opposition stems in part from the concern that vulnerable households will pay more under new rates because they cannot shift the timing of their consumption. There is an emerging narrative that rate reform is a zero-sum game; regulators can either protect vulnerable customers or encourage electrification, but not both. In this thesis, we challenge that perception: can well-designed distribution network tariffs deliver a win-win, reducing costs for households that adopt EVs and/or CCHPs and households that cannot yet afford to electrify? To answer this question, we calibrate a series of case studies in which we model household responses to different electricity tariffs under increasing levels of electrification. We consider the flat volumetric rates prevalent today, time-of-use rates recently implemented as the default in several states, and alternative rate designs that have already been implemented widely outside the US and which can be imposed by PUCs without specific legislative action. In each scenario, we assess the total network costs, economics of electrification, and distributional impacts on different household groups.

While many previous studies focus on the impacts of electrification on transmission and generation needs (e.g., [38], [39]), there has been relatively little analysis on the distribution grid under alternative tariff scenarios. Yet distribution, especially residential feeders with low load diversity, is where the majority of upgrades related to electrification are expected to occur [40]. Among studies focusing on the distribution grid, many rely on direct load control to reduce coincident peak demand from EVs and CCHPs, but this has not been widely implemented and many utilities in the US have no financial incentive to pursue direct load control. We address these barriers in Chapter 7. We are concerned that as the technical capabilities of load management evolve, advocates and regulators will ignore simpler solutions to harvest the low-hanging fruit. We believe direct load control, along with distributed generation and storage, will serve an important purpose in mitigating the need for expensive capital investments. But these are not substitutes for sound tariff design, which can be implemented at very low cost due to the existing metering infrastructure.

The central thesis question is particularly relevant for public utility commissioners who have a statutory obligation to regulate utilities in such a way that provides reliable service at least cost, ratepayer advocates who represent the public interest, climate advocates working to accelerate electrification, and state lawmakers focused on achieving climate, energy equity, and affordability targets who can influence PUC decision-making through legislative action if PUCs do not take the initiative themselves. Electricity rates are an often-overlooked barrier to decarbonization that do not require permitting change, extensive funding, or technology evolution. Using existing technology and existing statutory authority, every US state can undertake rate reform today. Our recommendations offer a path to removing one key barrier to rapid electrification while protecting all grid users from higher bills.

This thesis is organized as follows. Chapter 2 provides a brief background on utility regulation. Chapter 3 reviews the academic literature on electricity rate design and electrification impacts. Chapter 4 explores initiatives in both the US and Europe to replace flat volumetric rates. Chapter 5 presents the methodology for two case studies that examine performance of alternative rates against evaluative criteria. Chapter 6 presents the results of

our case studies. Chapter 7 summarizes and discusses the policy implications of our results. Chapter 8 concludes and offers recommendations for public utility commissions.

Chapter 2

Background on the Electric Power Sector and Cost of Service Regulation

"Free competition, the very basis of the modern social organization, superseded almost completely medieval restrictions, but it has just come to be recognized that the process of free competition fails in some cases to secure the public good, and it has been at last admitted that some control is necessary over such lines of industry as are affected with a public interest."

- Bruce Wyman [41]

This chapter provides background context on the electric power industry and the cost of service regulatory model. This context helps inform our methodology and is important for understanding both past and future considerations for rate design.

2.1 The Electric Power Sector in the US

We start with a brief overview of the structure of the electric power sector in the US today. The physical infrastructure comprising the electricity sector can be divided into three functions: generation (converting various forms of energy into electricity), transmission (transporting electricity across long distances at high voltage), and distribution (delivering electricity to homes and businesses at low voltage). Among US states, there is considerable diversity in the type of entity that provides each function, with two categories of distinction outlined below: ownership structure and level of vertical integration. The term "electricity utility" is typically used to describe any entity involved in transmission and/or distribution, regardless of its involvement in generation. Entities that own only generation are typically called independent power producers (IPPs).

2.1.1 Ownership Structure

In the US, there are three models for utility ownership: investor-owned (IOU), publiclyowned (federal, state, or municipal), and cooperative member-owned. IOUs serve 72% of US electricity customers while public and cooperative utilities serve the remaining 28% [31]. Among IOUs, some are publicly traded while others are privately held. The vast majority of public utilities are municipally owned. Notable exceptions are the Tennessee Valley Authority (federally-owned), Bonneville Power Administration (federally-owned), New York Power Authority (state-owned), and Nebraska Public Power District (state-owned). Investor-owned utilities are regulated at the state level. Municipally- and cooperatively-owned utilities are often exempted from state regulation. In this thesis, we are concerned primarily with the the distribution function of investor-owned utilities. However, while most of the recommendations in this thesis are directed at state regulators, they could equally apply to municipal and cooperative utility board members.

2.1.2 Level of Vertical Integration

Fully vertically integrated utilities own and operate generation, transmission, and distribution infrastructure. They are the sole entities responsible for balancing supply and demand within their territories to ensure system reliability. Examples include Georgia Power (Georgia) and Xcel Energy (Colorado).

In the 1990s and early 2000s, several states passed electricity sector restructuring laws that required vertically integrated utilities to divest themselves of generation assets [42]. At the federal level, FERC Order No. 888 (issued in 1996) required utilities to file nondiscriminatory open access transmission tariffs, removing a competitive barrier for thirdparty generators to interconnect with the transmission network. Shortly thereafter, FERC Order No. 2000 (issued in 1999) led to the voluntary formation of independent non-profit entities responsible for operating the transmission network, planning transmission upgrades, and administering competitive wholesale electricity markets. There are seven of these independent system operators (ISOs) in the US, which may span one state (e.g., NYISO, CAISO) or several (e.g., MISO, PJM). With the exception of ERCOT, ISOs are regulated by the Federal Energy Regulatory Commission (FERC), which approves market design and transmission access tariffs. There are two types of utilities operating within ISO service territories: 1) "wires-only" utilities, which own only transmission and/or distribution infrastructure. Examples include National Grid (Massachusetts) and Commonwealth Edison (Illinois); 2) utilities that are permitted to own generation assets in addition to transmission and distribution, but whose generation assets are dispatched economically by the ISO. Green Mountain Power (Vermont) and PacifiCorp (California) are two such examples, which purchase some portion of their electricity supply from the wholesale market.

In 13 states and the District of Columbia, restructuring applied not only to generation but also the supply component of the bill. Customers in these states are able to select their electricity supplier [43].¹ Suppliers do not own or operate physical infrastructure but purchase electricity on behalf of their customers. In all states except Texas, the local distribution utility is designated as the default supplier [44]. Suppliers are not subject to price or profit regulation [45]. While all states with retail choice fall within ISO territories, not all states in ISO territories permit retail choice (e.g., Vermont). We turn now to the origins of regulation in the electric power sector.

 $^{^{1}}$ As retail choice is sometimes called energy deregulation, there is a common misconception that these states deregulated all utility functions. This is not true; transmission and distribution remain tightly regulated in states with retail choice.

2.2 Origins of Regulation

Electricity distribution, or delivery, is a geographical natural monopoly. The industry is characterized by high sunk capital costs and significant economies of scale; the more network users in a particular geographical area, the lower the average price because fixed infrastructure costs are spread over higher consumption [46].

Unlike most industries, competition in electricity distribution drives costs up. When there are multiple electricity providers in one area, geographic economies of scale cannot be fully captured due to duplication of distribution facilities in the same geographic area. The summary judgment in Idaho Power & Light Co. v. Blomquist et al. (from 1914) offers a helpful hypothetical example to illustrate this phenomenon:

"An existing utility has already expended, say, \$1,000,000 in the power plant and transmission lines and distribution system in a town. Another utility, coming in, must also provide a power plant and transmission lines and a distributing system. If there is to be unrestricted competition, then the later distribution system must cover the same area as that of the older one. If it costs the same money, then there is an additional \$1,000,000 expended in a town where a \$1,000,000 system would be amply sufficient. There would be two sets of poles and transmission wires in the streets, the construction and keeping in repair of which would necessarily interfere with and obstruct the free use of the streets by the people more than one set of poles and wires; and two sets of electrical wires in a city would necessarily increase the danger to the lives and limbs of the people, and thus interfere with the peace, health, and welfare of the community."²

In light of this fact, as early as 1905, state legislatures passed laws establishing quasijudicial statewide commissions (public utility commissions, or PUCs) to regulate investorowned electricity utilities [47].³ The legality of state regulation over private industries "affected with a public interest" (as quoted above in Wyman [41]) was enshrined in the landmark 1876 US Supreme Court case, Munn v. Illinois.⁴ Regulation over electricity prices existed prior to the creation of state PUCs but consisted primarily of municipal franchise contract agreements [45]. The formation of PUCs shifted oversight of private electricity utilities to the state level. As it relates to electricity utilities, PUCs have two primary roles: 1) controlling market entry by issuing certificates of public convenience that grant exclusive franchise rights for IOUs to serve a particular geographic area; and 2) determining the prices IOUs are allowed to charge.

Today, PUCs are comprised of 3-7 commissioners and professional staff. Commissioners serve terms between 4-6 years and are either appointed by the governor or state legislature or publicly elected. PUCs regulate not only electricity utilities but also those operating in the telecommunications, water, and waste water industries.

²Idaho Power & Light Co. v. Blomquist et al. 26 Idaho 222 (1914)

 $^{^{3}}$ We use PUC as a generic term to refer to the state agency that regulates privately-owned electricity distribution utilities. In some states, these agencies go by different names, e.g., Public Service Commission or Department of Public Utilities.

⁴Munn v. Illinois 94 U.S. 113 (1876)

2.3 Cost of Service Regulation and Rate Cases

In the US, regulators exercise price-setting authority over IOUs via a framework known as Cost of Service Regulation (COSR).⁵ Under COSR, a utility is permitted to recover prudently incurred capital and operating costs, plus a regulated rate of return on invested capital. The prudency standard was first proposed by Supreme Court Justice Louis Brandeis in a 1923 dissenting opinion.⁶ Cost recovery is determined in a formal public adjudicatory process known as a rate case.⁷

A rate case consists of two core components: 1) determining the amount of money the utility is allowed to recover (the revenue requirement) and 2) allocating the revenue requirement to customers via rates (rate design). Historically, rate cases are either initiated by IOUs or occur at regular intervals (e.g., 3-5 years).

In a rate case, the utility submits its target revenue requirement along with supporting evidence. The public interest is represented by a "ratepayer advocate," which may be the state attorney general's office, an independent state agency, or a division within the PUC. Other individuals and organizations (e.g., trade groups, large industrial customers, environmental advocates) can formally intervene. Both the ratepayer advocate and intervenors can submit testimony, cross-examine witnesses in public evidentiary hearings, and offer competing analysis to challenge utility expenditures and modeling assumptions.

Regulators consider all evidence entered into the record and determine which costs to include in the revenue requirement. Regulators can disallow costs that they determine are excessive or the result of imprudent managerial decisions, although disallowances are typically small [45]. Once the revenue requirement is determined, regulators decide how to allocate the amount among customer classes (residential, commercial, industrial) via rates (i.e., tariffs) [48]. In Federal Power Commission v. Hope, the US Supreme Court ruled that PUCs are not required to use any one specific formula for setting rates so long as those rates are "just, reasonable, and not unduly discriminatory".⁸ This granted regulatory commissions latitude to exercise discretion in ratemaking, balancing the public good (reliable service at least cost) with fair compensation that allows IOUs to operate successfully and attract capital.

In practice, regulators typically allocate the share of the total revenue requirement to each class commensurate with the cost to serve that class [45]. Additional principles including simplicity and stability have guided regulators for decades [49]. Rate cases conclude with a final decision issued by the PUC (often via simple majority vote among the commissioners), which sets the new tariffs starting at a specified future date.

⁵As this thesis deals with distribution costs, we do *not* distinguish between IOUs that have been restructured through deregulation of generation and/or retail supply (e.g., Eversource in Massachusetts) and IOUs that remain vertically integrated (e.g., Georgia Power in Georgia).

⁶Southwestern Bell Telephone Company v. Public Service Commission of Missouri 262 U.S. 276 (1923)

⁷The brief summary in this section was inspired by Joskow and Schmalensee [45], which provides a more complete discussion of COSR and the mechanics of rate cases in practice.

⁸Federal Power Commission v. Hope Natural Gas 320 U.S. 591 (1944)

2.4 Limitations of Cost of Service Regulation and Alternatives

With perfect information, COSR can replicate the pressures provided by market competition, allowing utilities to recover only efficiently-incurred costs and earn a return on investments equal to the cost of capital [50]. However, in practice, resource limitations and information asymmetry between the regulator and IOU can lead to imperfect regulation and performance problems. IOUs may leverage their superior information during rate cases to increase their profits [51]. This issue is even more prevalent if the utility can influence the regulatory process [52], [53]. Research by the Energy and Policy Institute has revealed a pattern of utility involvement in state elections and policy development through campaign contributions and lobbying, respectively [54]. In perhaps the most extreme case, a utility was fined \$230 million for a bribery scheme that led to multiple indictments on federal racketeering charges [55]. More commonly, however, utilities aim to influence regulation through legal means by donating (either directly or through their employees or political action committees) and by supporting industry associations and similar groups that engage in lobbying activity on their behalf [54]. These political activities may further compromise the regulator's effectiveness in protecting customers from excessive rent-seeking by the IOU.

One other limitation of COSR is that it prima facie biases IOUs towards capital expenditures (which are eligible for a rate of return) over operating expenditures (which are simply passed through). This was identified by Averch and Johnson studying an idealized form of COSR with no information asymmetry or regulatory lag (i.e., the gap between rate cases) [56]. Empirical studies, which focus mostly on generation investments, show mixed results on the presence of capital bias in practice ([57], [58], [59]). Joskow and Schmalensee provide a concise summary of the literature on capital bias: "We believe that it is likely that the capital bias is articulated most importantly as a bias toward owning capital assets rather than buying services from third parties, since the costs of many of these services are treated as cost pass-throughs in the regulatory process, are not impacted by regulatory lag, and provide no profit opportunities" [45]. Under COSR, there are also no profit opportunities presented by solutions that *avoid* the need for capital investments that expand delivery capacity, including rate design. An IOU has little incentive to reduce its costs if it believes it is unlikely those costs will be disallowed.

Incentive regulation, also known as Performance-Based Regulation (PBR) in the US, attempts to address the limitations of COSR, especially when there are large regulatory lags [45]. The goal of incentive regulation is to induce regulated entities to employ their superior knowledge to reduce costs [50]. In its most basic form, revenue-cap regulation, the utility's revenue is capped (determined based on historical costs and comparisons with other utilities in similar geographies) and the utility earns a profit based on the gap between the allowed revenue and actually-incurred costs [52], [60]. Variations on this basic idea include menus of contracts (where the utility is allowed to choose among several regulation plans), profit sharing mechanisms, and TOTEX-based approaches (in which there is no accounting distinction between capital and operating expenses). These are described in detail in Brown and Sappington [50] and Joskow and Schmalensee [61].

In the US, PBR has focused on three components: multi-year rate plans (which adjust

rates based on external indices), revenue decoupling mechanisms (which eliminate incentives against promoting energy efficiency), and performance incentive mechanisms that adjust upward or downward the utility's rate of return base on its performance against predetermined metrics (e.g., average outage duration and average time to interconnect customer generation) [45]. PUCs have adopted these measures to varying degrees, with Hawaii generally recognized as the leading state on PBR. However, even in states with some or all of the three PBR components, COSR (along with its associated incentives) remains the underlying regulatory framework. Under both PBR and COSR, IOUs are typically also subject to minimum performance standards to ensure that they do not underinvest or overinvest in the system.

While PBR is outside the scope of our analysis, we mention it because it can be an effective complement to the rate interventions we propose. Regulators should leverage all tools at their disposal to achieve state policy mandates, including PBR.

2.5 The Link Between Rate Design and Network Investments

During rate case proceedings, utilities typically submit forecasts of expected electricity demand for the subsequent 10-20 years. These forecasts inform future investments, which are recovered from consumers via rates. When demand is growing, utilities must expand delivery capacity to continue to provide reliable service. For regulators, forecasts help determine whether investments are prudent.

Utilities forecast both the expected annual electricity consumption and the instantaneous maximum consumption (i.e., peak demand) at each network level. The latter is more relevant for distribution network investments, where equipment (substations, transformers, etc) is sized to accommodate peak loads.

A comprehensive analysis by Carvallo et al. [62] shows that utilities tend to overforecast peak demand. Dyson and Engel [63] attribute this tendency (which they estimated to be +12% for 10-year-ahead forecasts) to two factors: asymmetric risk (over-forecasting demand leads to excessive costs, while under-forecasting demand may lead to service disruptions) and the link between demand forecasting and capital investments.

Historically the main driver for network investment was growing demand [32]. In the last two decades, peak demand growth in the US has slowed due in part to slowing population growth, increasingly strict appliance efficiency standards, the growth of rooftop solar, and a transition away from an energy-intensive manufacturing economy to a service economy [62]. Economy-wide electrification, the recent boom in clean energy manufacturing, and growth in data center demand fueled by artificial intelligence applications are expected to reverse this trend and become significant drivers for new network investment [64], [13]. While EV and CCHP penetration rates are low today, utilities are pointing towards existing government incentives and anticipated phase-out mandates to forecast rapidly increasing peak demand, which will necessitate network investments to maintain reliability and support further electrification [14], [15].

These demand forecasts are influenced indirectly by rate design. In most states, utilities

forecast demand from residential customers assuming that they continue to pay flat volumetric rates. Under flat volumetric rates, consumers have no reason to avoid consuming electricity during times when the network is congested, contributing to peak demand growth and necessitating network investments, which customers ultimately pay for. Carvallo and Schwartz [37] found that utility integrated resource plans systematically ignore rate design as a tool to avoid the need for building new delivery capacity.

There is ample evidence that consumers respond to price signals when available and that these responses can meaningfully reduce peak demand [65], [66]. Faruqui and Sergici [66] find that the response is consistent across income groups, and price signals are even more effective when consumers have enabling technology that can respond automatically without the need for behavioral intervention. If cost-reflective price signals existed, EV and CCHP users would likely change the timing of their consumption in a way that avoids or defers the need to expand delivery capacity, even as total consumption grows. Yet COSR does not provide incentives for utilities to promote rates that reflect the long-run cost of electricity delivery. In fact, to the extent these rates may reduce peak demand and the level of efficient investment [50], utilities have an implicit disincentive to transition away from the status quo in rate design.

Kavulla [33] proposes a fix to this shortcoming of COSR, recommending that regulators (not utilities) decide on rate structure and design rates in a way that consumers are exposed (at least partially) to network cost drivers, thereby "activating" flexible demand. Regulators in California, Colorado, Hawaii, Michigan, and Missouri have done just that, by implementing time-of-use rates as the *default*, which we outline in Chapter 4. These rates have had high retention rates even though customers are free to opt out and choose to pay flat volumetric rates [34]. Compared to other solutions that mitigate network investments, including direct load control and battery storage deployment, it is relatively simple for regulators to monitor that utilities are in compliance with rate design reforms.

This chapter outlined the structure of the electric power sector in the US and the cost of service regulatory framework. In summary, while generation and (to a lesser extent) supply have been deregulated in many states, distribution remains characterized by geographic monopolies regulated at the state level subject to cost of service regulation. Under cost of service regulation, utilities recover prudently incurred costs and earn a regulated rate of return on capital investments. While some states have adopted elements of incentive regulation, COSR remains the dominant framework in the US. Under COSR, utilities have no financial incentive to reduce the need for capital investments. This poses a key challenge to beneficial electrification; as demand-driven capital investments are allocated to all grid users, this risks eroding the positive cost gap between electricity and fossil fuels and increasing energy burdens for vulnerable households.

While incentive regulation has been shown to enhance the productivity of IOUs [50], achieving beneficial electrification does not necessarily require a shift away from COSR, which could take years to implement. Rate design offers a near-term solution that can address the urgency of both decarbonization and electricity affordability. Regulators have a statutory obligation to design rates that are just and reasonable. We believe flat volumetric rates, which are based on outdated assumptions about consumers' willingness and ability to respond to prices, no longer meet this standard. There is overwhelming evidence that time-varying rates (complemented by enabling technology) can reduce the need for incremental network

investment by reducing peak demand and making more efficient use of existing infrastructure. Without specific authorizing legislation, regulators have imposed time-varying rates in five states, offering a blueprint to implement the rate design recommendations presented in this thesis.

Chapter 3

Literature Review and Research Contribution

In this chapter, we provide a summary of the academic literature on the impact of electrification on distribution grids and distribution network tariff design. At the end of this chapter, we discuss our research contribution.

Much of the existing literature at the intersection of electrification and network tariff design falls into two broad categories:

- 1. Studies on the grid impacts of electrification under the assumption of flat volumetric tariffs (i.e., price-inelastic demand)
- 2. Studies on alternative tariff designs using historical consumption data or in the presence of customer adoption of rooftop solar and battery storage

We review these two categories below.

3.1 Impacts of Electrification Under Flat Volumetric Tariffs

3.1.1 Heat Pump Impacts

Studies on grid and cost impacts of CCHP adoption typically assume that the heating equipment is used to achieve a constant indoor setpoint temperature. Waite and Modi [67] study distribution grid capacity to support heating electrification, finding that the existing capacity could support 53% electric heating (by delivered units of energy). This figure climbs to 97% if homes maintain backup fossil heating for extreme temperature days. Vaishnav and Fatimah [68] consider the household economic and carbon impact of electrifying space heating across the entire US; with existing electricity rates, only in states where resistive electric heating is prominent are savings achieved. White et al. [69] analyze the generation and transmission scale impacts of heat pump adoption in Texas; using the same open-source residential electricity consumption database that we employ in this thesis, the authors find that full electrification increases residential peak demand by 36% (12 GW). Wilson et al. [5]

study the cost and benefits of CCHP installation across all 50 US states and conclude that the technology is cost-effective in 65 million homes (inclusive of up-front installation costs). Simulating distribution networks and building heating demand for a region in rural Belgium, Protopapadaki and Saelens [70] find overloading and voltage issues above 30% penetration. Lee and Zhang [71] find that smart thermostats can severely increase winter peak heating demand through load synchronization.

These studies reveal significant geographic variation. In summer-peaking areas, heating electrification may actually reduce peak loads because heat pumps (which can operate in reverse as air conditioners) are more efficient than existing window air conditioning units. Conversely, in colder climates, heating electrification is projected to increase both system-wide peak demand [72] and local distribution peak demand [70].

3.1.2 Electric Vehicle Impacts

In transportation, many analyses on EV charging envision that drivers charge immediately upon returning home or as needed at public and workplace charging stations. Muratori [73] discovers that even if system-wide impacts are small in the near term due to low overall EV penetration, clustered EV adoption may require widespread upgrades for distribution grid equipment like small-scale transformers. Huang and Infield [74] employ Monte Carlo simulation and find that even at 20% adoption, EV charging increases peak demand on a representative low voltage feeder by 30%. Elmallah et al. [75] study cost impacts of distribution grid upgrades triggered by residential electrification in Northern California, which are dominated by EV charging. Coignard et al. [76] find that 60% of the feeders in the San Francisco Bay area would reach or exceed their maximum loading limit under an uncontrolled charging scenario where each household has one EV.

More recently, Needell et al. [77] estimate electricity demand curves at high penetrations of EV and rooftop photovoltaic (PV) systems in New York, NY and Dallas, TX, finding that workplace charging and delayed overnight charging are effective strategies for reducing peak demand and the need for investment in distribution capacity. Gschwendtner et al. [78] assess the impact on local system peak of plug-in behavior – when and for how long drivers decide to charge. Powell et al. [79] use charging data from a charging network operator to show how charging behavior can either complement or exacerbate renewable generation deployment issues.

Nour et al. [80] review the negative impacts of uncontrolled EV charging, including increases in peak demand and overloading of network components, and how those impacts can be mitigated by delayed and controlled charging techniques. Kevala [17] performs a bottom-up load forecasting and system impact study for EV charging in California, estimating up to \$50 billion in traditional electricity distribution grid infrastructure investments by 2035 under unmitigated charging scenarios. Finally, Steinbach and Blaschke [81] show that higher-income individuals tend to have uncontrolled charging schedules that align more closely with local network peaks than medium- and low-income individuals, posing a serious equity concern (since investments are recovered indiscriminately among all customers with flat volumetric rates).

There is less geographic variation in EV charging demand compared to electric heating, but cold temperatures impact battery efficiency and auxiliary energy consumption for cabin conditioning [82]. Charging activity in urban areas where dedicated parking is uncommon is less well-understood, but there has been progress recently at simulating this segment [83].

Efforts to understand the grid impacts of residential electrification under flat volumetric tariffs have provided important "worst case" scenarios to motivate an examination of electricity rates. When households have no incentive to change the timing of their consumption, EV charging and CCHP-related demand is likely to increase peak demand (both locally and at a bulk-system level), requiring extensive network upgrades. One Massachusetts distribution utility forecasted that adoption of these two technologies will increase coincident peak demand by 8 kW per customer, an increase of 250% compared to current peak demand values [14].

Many studies in this category typically perform "snapshot" analyses, looking at the peak hour of the year or a representative high-demand week. With the exception of a few papers (e.g., [5], [17]) they do not examine the longitudinal impacts on customers' costs due to electrification. They also offer little insight into system impacts if we expand the window of rate design possibilities and consider time-varying tariffs that consumers can react to, either through programmatic or behavioral changes.

Finally, some papers extend their analysis of electrification impacts by introducing topdown load control of EVs [76], [80], [84], [85], [86], [87] and heat pumps [88], [89], [90]. Load control is used to reduce peak demand by curtailing consumption or shifting consumption to hours with excess delivery capacity. Heat pumps are generally considered to be less flexible due to thermal comfort limits, but both devices can be shifted programatically without direct behavioral intervention. These analyses do not consider price-based coordination but instead direct management of devices by a utility or aggregator. Though not in every case, they also generally do not consider the regulatory context, incentives paid to participants, or consumer willingness to engage in control programs, only the theoretical potential of load control. While there is promising evidence (outlined in Chapter 4) that consumers accept frequent EV load control, we do not believe that it is a contemporary substitute for sound tariff design. Section 7.2.3 elaborates on this position.

3.2 Network Tariff Design in Theory

While energy economists have long understood that flat volumetric tariffs are inefficient, until recently there was not much political interest in addressing the problem. The metering technology made it difficult or impossible to measure hourly consumption, there was a strong status quo bias among regulators, and consistent load growth kept retail electricity prices relatively low. Because volumetric electricity consumption is correlated with income [91], legacy rates were somewhat progressive, which some stakeholders may have viewed as a positive feature not to be altered. High volumetric rates also encourage adoption of energyefficient appliances, which was supported by environmental advocates.

Tariff reform (specifically distribution network tariff reform) has gained attention due to three major recent trends: the roll-out of smart meters, the adoption of rooftop solar PV under net metering programs, and residential electrification.

Aided by \$8 billion of funding from the American Recovery and Reinvestment Act, many US utilities began rolling out smart energy meters in 2009 [92]. Today, over 73% of US house-

holds are estimated to have a smart meter [93], which allows utilities to collect consumption data at hourly or sub-hourly intervals. However, there is significant variation between states, even within the same region: Vermont has 99% smart meter adoption while in Massachusetts the figure is less than 10% [94].

In net metering programs, solar customers are charged for their net consumption (total consumed from grid minus total exported to the grid), typically calculated monthly. Under flat volumetric rates, solar customers can avoid paying not just supply costs but also network costs, even though they remain connected to the network and may not be generating solar during peak periods. Because sunk network costs are by definition unaffected by solar production, this creates a cost shift to customers who do not have solar PV systems [95]. And because solar adoption has been highly correlated with household income [96], there is an emerging equity issue associated with net metering programs. This has attracted much research interest (e.g., [97], [98], [99], [100]) along with regulatory initiatives aimed at designing successor programs to net metering. For example, California's recently-approved Net Energy Metering 3.0 tariff compensates solar customers for exported energy at the avoided cost rate rather than the full retail rate. This provides a clear incentive to align consumption with generation to avoid exports, or to install battery storage and charge during solar hours.

As states and nations have turned to electrification as a core decarbonization strategy, there is also a growing awareness that flat volumetric rates are a barrier to EV and CCHP adoption. Under a flat volumetric tariff that recovers both supply and network cost, the per-kWh charge substantially exceeds the marginal cost of generating electricity [35]. In Massachusetts today, it is cheaper to run a natural gas boiler than a heat pump at outdoor temperatures below 35F [26], and a customer who replaces their natural gas space heating with a CCHP system will likely pay more in annual energy costs, independent of the installed cost of the system itself [27]. These issues have prompted regulators and the academic community to rethink the historically-accepted practice of collecting network costs via flat volumetric rates.

A central objective of tariff design is to provide incentives for consumption while producing adequate revenue to recover the utility's costs. For a tariff to be efficient, it should be structured so that network users are charged according to the cost they impose on the system. This will provide incentives that limit overinvestment in the network. In addition, distribution network tariffs should be simple and predictable, non-discriminatory towards certain customer groups, and recover all regulated costs to ensure financial sustainability for the utility [32]. Flat volumetric tariffs violate many of these principles; Passey et al. [101] show that under time-invariant network tariffs, the costs consumers pay rarely reflects the costs they impose on the system. This leads to inefficient incentives for grid usage and to cost shifts between grid users. Yet there is also evidence that perfect cost-reflexivity may not be the ideal design; Hofmann and Lindberg [102] demonstrate that consumer responses become muted under increasingly complex tariffs.

In theory, the most efficient distribution network tariff contains two parts: a forwardlooking charge that reflects the long-run marginal cost of upgrading the network at each location and a complementary fixed charge to recover the residual costs of past network investments [32], [103]. While there is consensus around this framework, there is considerable disagreement in the literature on how it should be interpreted to design distribution network tariffs in practice, with some focusing on increasing fixed charges and others focusing instead on reflecting the true cost of network upgrades in time-varying tariffs. Though few papers specifically address barriers to electrification, many make proposals that would reduce the operating cost of EVs and CCHPs by sending more efficient signals and/or shifting cost recovery away from volumetric charges.

Broadly speaking, the literature contains two categories of analysis on redesigning network tariffs: $^{\rm 1}$

- 1. Backwards-looking studies that apply novel rates to static historical data (retroactive)
- 2. Optimization-based studies in which grid users can make decisions in response to new tariffs (proactive)

3.2.1 Tariff Design with Retroactive Analysis

Among retroactive analyses, there is considerable diversity in approach. Using annual data from California grid users, Borenstein et al. [100] experiment with shifting non-marginal network costs to a fixed charge. To improve equity, the authors argue for an income-graduated fixed charge (IGFC) similar to a progressive income tax. Using smart meter data from 100,000 grid users in the Chicago area, Burger et al. [105] recommend a two-part tariff with a per-kWh charge set equal to the social marginal cost of energy and an income-based fixed charge. Acknowledging that utilities collecting income information would be both burdensome and potentially unacceptable from a privacy perspective, Battle et al. [106] propose a fixed charge based on historical consumption, reasoning that past usage is a relatively good proxy for past costs users imposed on the system. Simshauser [98] suggests a three-part tariff (with fixed, volumetric, and capacity components) to mitigate cost shifts between solar and non-solar households in Queensland, Australia. Simshauser and Downer [107] find that households in financial hardship are the most adversely affected by flat tariffs and have the most to gain from a transition to cost-reflective network tariffs. Bergaentzlé et al. [108] study the distributional impacts of tariff reform in Denmark on different household groups, including income-based and those with EVs and heat pumps. Sergici et al. [27] use actual metered gas and electricity data from a utility in Massachusetts and advocate for demand charges and time-of-use volumetric charges to improve the economics of heating electrification. Kahn-Lang et al. [109] calculate bill savings from efficiency and electrification measures under flat and several time-of-use rates with varying peak:off-peak ratios. Küfeoğlu and Pollitt 110 examine the impacts of EV and PV adoption, concluding that the technologies have opposite effects: EV adoption drives tariff prices down while PV adoption increases rates. These papers typically treat network costs as sunk and apply the constraint of "revenue neutrality," meaning that the total amount of money collected by the utility is constant and customer end-use consumption is static. They do not consider outcomes where additional network investments are required to meet peak demand but can be avoided, reduced, or

¹There is a third category that attempts to estimate consumer price-elasticity to time-varying rates using empirical evidence (e.g., [65], [66], [102], [104]). While these are valuable for understanding potential tariff impacts, they do not differentiate between load related to electrification and other household consumption and are not well suited for the new era of demand flexibility where consumers can respond perfectly with the aid of automated tools. Rather, they focus on behavioral changes.

deferred through appropriate tariff design. The assumption of zero marginal demand-driven distribution cost does not align with utility planning to support end-use electrification of transportation, buildings, and other sectors [14], [15].

Other "retroactive" studies consider the possibility of future network investment or load growth. Morell-Dameto et al. [111] apply a simple fixed charge to cover residual costs with a highly spatially- and temporally-granular per-kWh component that reflects forward looking network expansion costs, choosing Slovenia as a case study.² Using smart meter data from Danish consumers, Gunkel et al. [112] advocate for a two-part tariff that includes an individual peak charge and a system-coincident peak charge. Finally, Govaerts et al. [113] build a synthetic grid to investigate the "Long Run Incremental Cost" approach to calibrate network tariffs.

3.2.2 Tariff Design with Proactive Analysis

Papers in the proactive category use optimization techniques (typically single or bi-level linear programming) to calibrate tariffs based on consumer responses. These typically focus on adoption of solar or battery storage due to the interest in addressing equity impacts of policies like net metering. Abdelmotteleb et al. [114] consider four network tariff designs and simulate consumers' cost-minimizing responses, with the ability to purchase electricity from the grid or invest in onsite generation. The authors ultimately recommend a fixed charge plus a peak-coincident network charge based on each customer's demand during the annual peak hour. Schittekatte [115] uses a bi-level optimization to simulate how consumers with the option to invest in distributed energy resources (DERs) like solar PV may respond to three different network tariff designs. Nouicer et al. [116] similarly apply a bi-level optimization (whereby the utility maximizes aggregate social welfare and consumers maximize their own welfare subject to the network tariff) to explore curtailment as a form of demand flexibility. All three papers adhere to the European regulatory context, where utilities have an incentive to reduce operating costs. In contrast, US utilities operating under a cost-of-service regulatory framework would have no financial incentive to minimize capital or operating costs. The papers in this category also do not consider outcomes where customers increase net demand through EV and/or CCHP adoption.

There are a few recent exceptions, which consider the interaction between tariff design and electrification. Hoarau and Perez [99] study network tariff design in the UK in the presence of both DER (solar PV and batteries) and EV adoption using a non-cooperative game where network users seek to minimize their costs. The authors study fixed, volumetric, and capacity tariffs and conclude that EV adoption can mitigate the cost shift caused by PV and battery adoption. Similar to Küfeoğlu and Pollitt [110], Hoarau and Perez [99] find that tariffs that incentivize DER adoption stifle electrification and vice versa. Arlt et al. [117] simulate responses of HVAC systems in Austin, Texas to real-time tariffs during a full year. The authors find moderate savings potential (\$39/customer/year); when all HVAC systems react in a coordinated fashion to the dynamic price it causes local congestion issues, which can be mitigated partially by supplementary demand management. However, the authors

²The approach in Morell-Dameto et al. [111] adheres to the "efficient ideal" advanced in Pérez-Arriaga et al. [32], which posits that forward-looking costs should be calculated precisely at each network node.
calculate rates in a cost-neutral way and do not consider network reinvestment (instead using assumed congestion prices). Bergaentzlé et al. [118] explore different tariff structures for a district heating system in Denmark, considering capacity charges, peak time rebates, and dynamic volumetric prices. The authors conduct a case study by minimizing operating costs from the perspective of the distribution utility and ultimately recommending adopting a dynamic tariff that reflects real-time grid conditions. Finally, Hennig et al. [119] lay out a framework for assessing network tariff performance, focusing on cost efficiency, cost recovery, and implementation burden. In a case study, the authors test four network tariff designs under EV adoption, where the distribution utility upgrades distribution transformers once they reach 95% of their capacity. The case study illustrates how to use the tariff assessment framework in practice, but it uses a sample of only 50 consumers in Germany and considers network tariffs in isolation, with no analysis of the interaction between network and energy tariffs.

3.3 Research Contribution

There has been only limited research on the design of distribution network tariffs in the presence of electrification and the grid impacts of electrification under time-varying tariffs. This thesis fills an important gap in the literature by modeling electrification under alternative tariff scenarios and focusing on the long-term cost impacts related to distribution network investment. Critically, we assume that the utility has no incentive to minimize operating costs, which reflects cost-of-service regulation prevalent in the US. We consider only bottomup responses to tariffs published ex-ante by individual households acting independently to minimize costs (i.e., no coordination between households). We apply technology-agnostic whole-home tariffs that do not require additional metering. Our tariff choices are inspired by existing distribution network tariff designs in Europe (reviewed in Chapter 4), which seem largely to have escaped the attention of regulators and analysts in the US.

In our case studies, tariff prices evolve to collect the full revenue requirement as the network is expanded to accommodate growing load. Many existing studies consider tariff reform at a single point in time, which necessary creates winners and losers compared to the status quo as the same amount of money must be collected (i.e., there is a constraint of revenue neutrality). However, when we consider how a rate will impact future loads and network investment as customers respond to price signals, we can assess whether well-designed rates can deliver benefits for all customers, including those who have not yet electrified. Finally, we consider interactions between separately-designed energy and distribution network tariffs, which we have not encountered anywhere else in the literature.

The closest existing studies to this thesis are Satchwell et al. [120] and Li and Jenn [121], although neither study considers heating electrification. Satchwell et al. [120] look at the impacts of EV adoption on utility customers and shareholders. The authors model charging demand, estimate network upgrade and capacity expansion costs, and calculate the resulting average retail rates. They find that EVs reduce rates for all customers under a "low peak impact" scenario, but they consider only direct load control (not price-based coordination), assuming all customers continue to pay flat volumetric rates. Li and Jenn [121] use empirical EV charging data (including residential charging responding to TOU rates) to estimate that

California utilities will need to invest between \$6 and \$20 billion to upgrade distribution infrastructure between now and 2045, but that growth in consumption will outpace these investments and apply downward rate pressure of between \$0.01 and \$0.06/kWh. However, the authors do not consider tariff design, only the expected rate impacts under existing tariffs and the current revenue requirement.

In this chapter, we reviewed the academic literature on electrification impacts and network tariff design. We introduced our research contribution, which extends and addresses gaps in a small body of existing literature at the intersection of these fields. Table 3.1 shows how this thesis compares to existing research. This is not a complete list but a representative sample.

Author	Technology	Geography	Coordination Mechanism	Considers Long Term Rate Impacts?	Duration
Hoarau and Perez [99]	PV, EV	UK	Tariffs	No	1 year
Bergaentzlé et al. [118]	District heating	Denmark	Tariffs	No	1 year
Arlt et al. [117]	HVAC	Austin, TX	Tariffs	No	1 year
Hennig et al. [119]	EV	Germany	Tariffs	Yes	0-25 EVs
Satchwell et al. [120]	EV	Southern US	Control	Yes	30 years
Li and Jenn [121]	EV	CA	N/a	Yes	20 years
This thesis	EV, CCHP	US (MA, NC)	Tariffs	Yes	0-100% adoption

Table 3.1: Classification of network tariff design studies with an application to electrification

Chapter 4

Distribution Network Tariff Design in Practice: Approaches from the US and Europe

In this chapter, we first cover recent tariff reform efforts in the US, including "whole-home" and "electrification-specific" designs. Next, we review distribution network tariffs widely implemented in Europe, which serve as inspiration for our case studies.

4.1 Distribution Network Tariff Design in the US

In this section, we review recent tariff reform efforts in the US. While over 73% of residential consumers have smart meters capable of metering at sub-hourly intervals [93], only 10% are estimated to be subject to time-varying rates today [34]. Many utilities offer voluntary time-of-use rates, but these are not widely promoted. Recognizing that flat volumetric rates do not reflect underlying costs and that voluntary TOU rates are undersubscribed, several states have introduced alternatives. We highlight two types here: default TOU tariffs and electrification-specific tariffs.

4.1.1 Default Time-of-Use Tariffs

In 2020, California became the first US state to institute default TOU rates for all residential customers, with network and supply costs combined into a single, time-differentiated perkWh charge. Customers were transitioned to either a 2-part or 3-part TOU rate; the timing of each period and the ratio of on-peak to off-peak price varied according to location and distribution utility. Those who wanted to remain on a flat volumetric rate needed to explicitly request to opt out. All new residential accounts starting in October 2020 have been placed on the TOU rate.

Other states have followed California's lead. After a multi-year Advanced Rate Design proceeding, in May 2023 Hawaii's PUC ordered the utility (Hawaiian Electric) to implement a default TOU rate by July 1, 2024. The TOU roll-out will be paired with a fixed charge reform intended to shift some of the utility's fixed costs to a non-variable charge and the introduction

of a "grid access" fee proportional to each customer's monthly maximum demand. This grid access fee is first instance of a demand charge being used for a *default* residential tariff in the US. We are aware of two instances where a demand charge is used a component of a *voluntary* residential tariff: 1) in North Carolina, Duke Energy's RS tariff has a two-part demand charge, with one component calculated based on the maximum demand during the on-peak period (6:00 - 9:00 PM during the summer and 6:00 - 9:00 AM during the winter) and the other component calculated based on the maximum demand during all hours; 2) in Arizona, Arizona Public Service offers an opt-in TOU rate with a demand charge assessed based on the customer's highest usage hour between 4:00 PM and 7:00 PM during weekdays. Participants can request one "demand charge credit" per year, calculated using the difference between the bill month's kW demand and the kW demand from the same billing period the previous year.

In Missouri, regulators recently approved a default TOU rate with an unprecedented 5:1 peak to off-peak ratio for customers of the distribution utility Evergy. In Michigan, DTE Energy transitioned residential customers to a more modest default TOU rate in March 2023 (1.5:1 ratio), and Xcel Energy in Colorado was required to enact a similar reform in 2022. Table 4.1 shows all default TOU rates currently approved in the US. It is noteworthy that no state with retail choice (whereby consumers may select their electricity supplier for the generation portion of the bill) has yet implemented default TOU rates for network cost recovery. All examples except for Hawaii include network and supply costs bundled as a single per-kWh charge.

State	Utility	TOU Periods	Per-kWh charges	Status
ні	Hawaiian Electric	Super off-peak: 9am - 5pm Off-peak: 9pm - 9am On-peak: 5pm - 9pm	Specific prices will be determined in fu- ture rate cases but must adhere to a 3:2:1 ratio for on-peak: off-peak: su- per off-peak. There will also be a grid access chart similar to capacity charges in Europe, but the amount is as yet un- specified.	Approved by PUC; set to take effect by July 1, 2024
CA	Several	Off-peak: 8pm - 5pm On-peak: 5pm - 8am	\$0.38/kWh off-peak \$0.42/kWh on-peak (Oct – May) \$0.51/kWh on-peak (Jun – Sep)	Active
СО	Xcel Energy	Off-peak: 7pm – 1pm Mid-peak: 1pm – 3pm On-peak: 3pm – 7pm	\$0.11/kWh off-peak \$0.19/kWh mid-peak \$0.27/kWh on-peak	Active
МО	Evergy	Off-peak: 8pm – 4pm On-peak: 4pm – 8pm	\$0.09/kWh off-peak \$0.38/kWh on-peak	Active
MI	DTE Energy	Off-peak: 7pm – 3pm On-peak: 3pm – 7pm	\$0.15/kWh off-peak \$0.16/kWh on-peak (Oct – May) \$0.21/kWh on-peak (Jun – Sep)	Active

Table 4.1: Default TOU rates approved in the US^1

These new rates are undoubtedly an improvement over flat volumetric tariffs. Under default TOU rates, consumers receive price signals that, as one rate reform advocate put it, dispel "the basic lie to retail consumers that every kilowatt-hour costs the same regardless of the time of day or the season of the year" [122]. And even though all rates allow customers

¹For California, rates displayed are for PG&E's default TOU plan.



Figure 4.1: Illustration of rebound peak due to EV charging. The EIS curve depicts California EV charging demand in 2035 responding to a TOU rate with an off-peak period starting at 9:00 PM. (Source: Public Advocates Office [18])

to opt out and pay a flat volumetric charge, we can expect that only a small portion of customers will do so based on empirical evidence from opt-out TOU pilot programs [123] and utility resource planning [37].

However, by combining all costs into a single per-kWh charge, TOU rates create the potential for adverse impacts due to correlated demand responding in an automated manner in response to tariff price. This problem is particularly acute at the distribution level where there is relatively low diversity of customer loads compared to parts of the distribution network that host multiple housing types and customer classes (e.g., commercial and industrial). When a large population of EVs or heat pumps respond to a change in tariff price, it creates "rebound peaks" that may exceed existing peaks, shown in Figure 4.1. This phenomenon has been found empirically [124] and in simulation (e.g., [125], [126]) and limits the effectiveness of volumetric time-of-use rates at mitigating network impacts beyond low adoption levels.

4.1.2 Technology-Specific Tariffs

As an alternative or complement to default TOU rates, some US utilities have also begun to offer tariffs specifically tailored to EVs and CCHPs. Utilities and regulators recognize the unique operating characteristics of these devices, particularly their demand flexibility compared to other household appliances. These tariffs allow households to save money by shifting the timing of consumption to off-peak hours or allowing the utility to curtail consumption during critical periods.

Demand response programs that compensate customers (typically via a one-time or annual payment) in exchange for control of a smart thermostat have existed for many years. What is novel about some of these new programs is that they incorporate the expected savings into a discounted per-kWh charge. Households enrolling in electrification-specific tariffs are often not required to select a time-varying tariff for their remaining household demand. We review a selection of these dedicated electrification rates here, shown in Tables 4.2 and 4.3.

State	Utility	Details		
MN	Xcel Energy	Unlimited charging between 9pm and 9am for a flat charge of $42.50/\mathrm{month}$		
MA	Unitil	TOU rate for separately-metered EV charging: ² On-peak (M-F 3pm - 8pm): \$0.68/kWh Mid-peak (M-F 6am - 3pm): \$0.45/kWh Off-peak (all other hours): \$0.36/kWh		
NJ	PSEG	Ex-post credit of $0.105/kWh$ for off-peak charging (between 9pm – 7am M-F), as measured by a compatible smart Level 2 charger		
CA	Sonoma Clean Power	One-time \$250 enrollment bonus plus $5/month$ in exchange for EV owners' authorization to curtail charging for up to 120 hours per year		

Table 4.2: Selection of EV-specific residential tariffs in the US

A 2019 report by the Smart Electric Power Alliance estimated that across 20 utilities, 21% of eligible EV customers were enrolled on a specialized EV Rate [127]. The report demonstrated that participation increases significantly when the rate is marketed by utilities and connected to EV purchase incentives. The Vermont utility Green Mountain Power offers a free Level 2 charger with the purchase of any electric vehicle. Customers claiming this incentive are required to enroll on one of two EV rates: a TOU rate with an on-peak period between 1:00 PM – 9:00 PM on weekdays and a managed charging rate where the utility is allowed to curtail charging up to 5 times per month. Green Mountain Power reported that 80% of customers purchasing EVs claimed the free charger incentive and enrolled on one of the EV rates [128]. While Wong et al. [129] surveyed individuals and found that most people are in theory willing to participate in smart charging programs without incentives, Bailey et al. [130] use an actual field experiment to show that charging behavior reverts to uncontrolled (i.e., plug immediately upon returning home) when incentives are removed and only nudges provided.

For CCHPs, Central Maine Power (CMP) offers a seasonal heat pump rate with a significantly-reduced per-kWh rate during winter months, a higher per-kWh charge during summer months, and a slightly higher fixed charge all year round (shown in Table 4.3). Note that in contrast to the EV rates discussed, CMP's rate does not provide any incentive for shifting the timing of consumption; it is simply a discount for consumption during winter months for households who affirm that they have a heat pump (although no proof is required). One possible explanation for the lack of time-differentiation is that whereas some EV rates rely on metered data from the vehicle or charger that can be shared with the utility for billing purposes, there is no analogous source of data for heat pumps. The lack of integrated metering and Internet-connectivity is a barrier to offering time-differentiated heat pump rates without a secondary meter.

 $^{^{2}}$ These prices assume that the customer chooses Unitil for the supply portion of their bill.

Table 4.3: Central Maine Power's seasonal heat pump rate compared to its standard residential service rate

Tariff	Customer Charge	Per-kWh Charge
Seasonal Heat Pump Rate	\$37.71/mo	May – October: \$0.137508/kWh November – April: \$0.004317/kWh
Standard Residential Rate	\$21.55/mo	\$0.085717/kWh

4.2 Distribution Network Tariff Design in Europe

Since the advent of smart meters, Europe has been a pioneer in network tariff design. European regulators have implemented several innovative approaches for allocating network costs and sending efficient price signals. Until recently, these efforts had not been coordinated at an EU level but were instead spearheaded by individual member states.

Regulation (EU) 2019/943 mandates that EU member states implement cost-reflective distribution network tariffs, which implies that future network costs need to be reflected to the grid users through the tariff. The EU Agency for the Cooperation of Energy Regulators (ACER) reports that already 25 of the 27 assessed EU member States had some form of capacity distribution network charge in place (based on the maximum capacity in kW measured or kW contracted, with or without time-differentiation) [131].³ As there are many nuances, it is hard to verify where households, connected to the lowest voltage level, are still facing more simplified distribution network charges. These simplified rates typically consist of a flat volumetric charge and a fixed charge. At the time of writing, based on ACER's analysis, we estimate that at least in one third of the member states households are facing some sort of capacity-based and/or TOU per-kWh network charges as the default [131]. We highlight four examples below.

In Spain, the distribution network tariff for residential consumers consists of both TOU per-kWh and TOU capacity components. The per-kWh component has 3 periods per day, with tariff levels ranging from 0.1 euro/kWh (off-peak) to 2.9 euro/kWh (on-peak). For the capacity component, consumers subscribe ex-ante to a desired level during two periods: on-peak and off-peak. If consumption exceeds the ex-ante level, power is temporarily disconnected at the meter, providing a real-time feedback signal to reduce simultaneous consumption.

In France, residential consumers have access to a regulated rate with a volumetric and capacity component. For the capacity component, consumers subscribe ex-ante up to a capacity level, and the price per kW goes down with more kWs contracted (i.e., declining block). For the per-kWh component, consumers can choose between a flat volumetric tariff, a simple two-part TOU tariff, and a dynamic tariff that resembles a critical peak rate where the utility announces one day prior whether it is a critical, peak, or off-peak day. During critical days, the price at all hours is significantly higher than peak and off-peak days, discouraging consumption at all hours equally.

³Capacity charges are described in detail in Section 5.2.

Italy's regulated distribution network tariff for residential consumers is similar to France's except it does not have a per-kWh component and the price per contracted kW does not change as the contracted amount grows.

Finally, the Flanders region in Belgium recently reformed its previously flat volumetric distribution network tariff. The new tariff consists of a small flat volumetric charge plus a capacity charge based on each consumer's 15-minute maximum demand in each month.

These four regulated distribution network tariffs in Europe provide inspiration for possible alternatives to the flat volumetric tariffs prevalent today in the US. While it is difficult to assess which of the European designs performs the best - and each country faces a unique set of challenges related to its clean energy transition - it is telling that all tariffs include a deterrent to maximum power consumption in the form of a capacity charge (either ex-ante or ex-post).

Among other European countries, there has been interest in reforming distribution network tariffs to comply with the new EU regulation and improve efficient utilization of the existing network. In Norway, while network tariffs today are predominantly volumetric, the energy ministry proposed shifting to a three-part tariff consisting of time-differentiated per-kWh charge, a capacity charge, and a fixed charge [132].

In the UK, OFGEM (the UK energy regulatory agency) initiated a review of network tariffs in 2018 via a Significant Code Review process. Network access charges were modified to only recover "shallow costs" directly related to interconnection, and OFGEM has an ongoing effort to reform forward-looking charges since 2022 [133].

Looking beyond the distribution network component of the bill (which is typically determined by national regulators) Europe also has an advanced electricity retail supply industry. Suppliers have created innovative offerings that help customers reduce costs by adjusting the timing of consumption or allowing for direct load control, which also serve as inspiration for US tariff reform efforts (though outside the scope of this thesis). For example, in the UK, the retailer Octopus Energy offers a tariff where the price of electricity for EV charging is heavily discounted for four overnight hours. In Spain, Iberdrola's retail arm allows consumers to select their off-peak energy tariff hours based on their household schedules.

In this chapter, we reviewed recent tariff reform efforts in the US, including both wholehome default TOU and device-specific offerings. We also explored popular distribution network tariff designs in Europe that have been effective at reducing local peak demand and allowing grid users to reduce their costs by changing the timing of consumption. While most European countries use some form of incentive regulation for electric distribution companies, there is no reason why US regulators could not impose the same types of tariffs prevalent in European countries. In the next chapter, we introduce the methodology for our case studies in which we test tariff designs widely implemented in Europe.

Chapter 5

Methodology: Simulating Residential Electrification Under Distribution Network Tariff Alternatives

The grid impacts of electrification have been modeled extensively, but the interaction between electrification and distribution network tariff design is only a nascent field, as discussed in Chapter 3. In this thesis, we develop a modeling framework that allows us to understand how distribution network tariff design impacts network investment needs in the face of high electrification. This in turn allows us to assess the evolving economics of electrification for individual households and distributional impacts on different customer groups.

In this chapter, we describe the modeling framework and case study formulation, including key assumptions and performance metrics. We adhere to several principles in our analytical approach:

- 1. Leverage open-source data to the greatest extent possible
- 2. Isolate the impacts of tariff design (in contrast to other dynamics like consumer technology adoption)
- 3. Choose representative geographies
- 4. When making assumptions, try to approximate actual end-use behavior based on empirical data

Our approach combines open-source data and a previously-developed building optimization model with a novel rate design "engine" to translate network investments into long-term cost impacts, much in the same way that a rate case would allocate costs to consumers. The core research contribution is the creation of a tool that allows utilities and regulators to understand the long-term impacts of tariff design on customer costs at different points in the electrification adoption curve. In contrast, most research on the impacts of electrification look at a single point in time (e.g., the peak day or peak week) or focuses only on the costs for electrified households. While any tariff reform will create winners and losers in the short-term, it is important to consider how new proposed rates will impact *future* investment needs, which directly influence costs paid by grid users. Through this analysis, we can better understand under which scenarios electrification would apply downward or upward rate pressure, tradeoffs in tariff design (e.g. simplicity vs efficiency), and how to better ensure beneficial outcomes for all customer groups.

The modeling process consists of several components, listed here, and described in greater detail in the subsections below:

- 1. Select data sources
- 2. Define distribution network tariff options
- 3. Select geography, choose representative network investment cases, and configure distribution network model
- 4. Simulate residential electricity consumption in response to tariff design
- 5. Calculate network investments and changes to tariff prices

Using these steps, we conduct two cases studies:

- 1. Price-responsive EV charging in a warm-climate geography with a detailed network model (Greensboro)
- 2. Price-responsive EV and CCHP operation in a cold-climate geography with a simplified network model (Massachusetts)

The two case studies share many features. They differ in geography, type of end-use consumption that is price-responsive, type of electrification, and tariff scenarios tested.

5.1 Data Sources

5.1.1 Household Energy Consumption Profiles

We obtain residential energy consumption time series data from the ResStock Database created by the National Renewable Energy Laboratory. ResStock provides synthetic annual hourly electricity and fuel consumption for a collection of simulated residential building archetypes, calibrated against actual building performance [134]. Consumption is segmented by household appliance end use (e.g., lighting, clothes washer, heating, etc). We choose diverse samples of buildings using the *weight* key, which indicates how representative each building archetype is of the actual housing stock in a census region. We consider only single family detached houses, using measure package zero (baseline). We remove houses with solar photovoltaic and battery storage because total consumption reflects the net consumption inclusive of self-consumed and exported solar generation. We use load profiles for the actual meteorological year 2018. One limitation of our study is that as electrification proceeds, we do not use future forecasted temperature data.

5.1.2 Vehicle Driving Profiles

We use the National Household Travel Survey (NHTS) to generate vehicle driving profiles. NHTS is conducted every 5 years by the US Federal Highway Administration and asks respondents to log all trips taken within one 24-hour period, starting at 4:00 AM local time and ending at 3:59 AM local time the next day [135]. For each trip, the respondent includes the trip purpose, arrival and departure times, number of miles traveled, and start and end locations. For households with multiple vehicles, each vehicle is tracked separately. While there are longitudinal datasets on driving patterns, these are limited to small geographies. Using the NHTS dataset, which includes all 50 states, allows us to extend our approach to the entire US.

Starting with the raw NHTS trip data, we obtain, for each vehicle, a parameter profile that contains the earliest home departure hour, the latest home arrival hour, and the total number of miles driven in the 24-hour period. These parameters signify when the vehicle is expected to be plugged in at home and the amount of electricity required to restore the battery's state of charge. The NHTS data includes trip weights to indicate how representative each trip is of all trips in a region, which we use for sampling purposes.

Because the NHTS survey is not longitudinal (i.e., each survey response covers only a 24-hour period of travel behavior), we use the following procedure to translate the survey responses to annual driving profiles. We require an annual profile because we are interested in not only peak demand during certain representative days but also annual cost impacts. While ResStock is not linked to NHTS data explicitly, both datasets include household demographic data, including income level and number of occupants, allowing us to match vehicles with households that have similar features.

- 1. For each ResStock household, we filter the NHTS data for vehicles associated with households of the same income level and number of occupants.
- 2. Using the NHTS trip weights, we randomly select one weekend and one weekday parameter profile (containing the departure hour, arrival hour, and miles driven) to associate with the household. These profiles provide the inputs to convert from survey responses to pseudo-random travel profiles.
- 3. For each house, we create weekend and weekday normal distributions centered at the actual number of miles driven with a standard deviation equal to 10% of the number of miles driven. We create discrete distributions for the weekend and weekday arrival and departure hours, as shown in Figure 5.1.
- 4. For each day of the year, we sample a departure and arrival hours from the discrete distribution and sample the daily mileage from the normal distribution (weekend or weekday).
- 5. We convert daily miles driven to electricity consumption using the regression computed by Yuksel and Michalek [82] in which battery efficiency depends on outdoor air temperature. We use the average temperature during hours that the vehicle is not at home on the given travel day. The difference in energy efficiency throughout the year



Figure 5.1: Discrete distribution used to select departure and arrival hour for driving profiles each day. The numbers in the boxes represent the offset (in hours) from the hour in the NHTS parameter profile. The percentages below reflect the probability of drawing from each box.

is due to both the temperature impacts on battery chemistry and cabin heating and air conditioning.

After this procedure, each household has a unique and uncorrelated annual vehicle usage profile that indicates when the vehicle is home versus away, and the daily electricity consumption due to driving. We use this profile to assign an EV battery capacity. The available capacities, based on common EVs available for purchase today, are 12, 40, 60, 90, or 120 kWh (the 12 kWh represents a plug-in hybrid electric vehicle). We select the minimum capacity that fulfills the vehicle's maximum daily electricity consumption, plus a 20% buffer. This assignment captures the expectation that households that routinely drive short distances are more likely to buy EVs with smaller batteries, which have lower associated capital costs. For any house where meeting the annual charging need is infeasible even with the maximum available battery size (i.e., the car is not plugged in long enough at home to avoid depleting the battery), we assume that house does not purchase an EV. This design choice aligns with the approach by Wei et al. [136].

5.1.3 Distribution Network Topology

For the Greensboro case study, we use synthetic distribution network topologies to study the impacts of electrification. In the absence of real network data from utilities (which typically cannot be shared due to security and privacy concerns), one widely used open-source distribution grid dataset is SMART-DS, developed by the National Renewable Energy Laboratory jointly with MIT and Comillas Pontifical University in Madrid, Spain. ¹ SMART-DS builds synthetic grid topologies using the Reference Network Model [137], including all electrical

¹https://www.nrel.gov/grid/smart-ds.html

components (e.g., lines, substations, capacitors, regulators, fuses, etc). These components connect to buildings modeled in ResStock and ComStock (for commercial buildings). The dataset is structured as nested directories to reflect the distribution grid's hierarchy, from the sub-transmission network down to distribution feeders. SMART-DS can be used to run power flows (using either CYME or OpenDSS software) for studying network behavior under different technology adoption scenarios. The authors performed extensive validation against real-world distribution feeders in the US. Integration with ResStock makes SMART-DS an ideal tool for studying where thermal and voltage violations appear at higher electrification levels. SMART-DS has been developed for 3 US cities (Austin, TX, Greensboro, NC, and San Francisco, CA), in addition to the entire state of Texas as part of the Texas Combined Transmission-Distribution Test Case developed at Texas A&M University in joint partnership with the aforementioned institutions.

5.2 Tariff Price Calculation

We test variations on three standard formats of distribution network tariff design: fixed, perkWh, and capacity (based on per-kW demand). These designs are not mutually exclusive; they can be combined to create hybrid tariffs that balance different policy objectives. Table 5.1 shows the key details of each tariff type, along with illustrative prices.

Below, we describe how tariff prices are calculated in our case studies. We assume that energy and distribution network tariffs are designed separately (even if the utility is vertically integrated). We pair the distribution network tariffs under consideration with two types of energy tariffs: flat per-kWh and time-of-use per-kWh. The energy tariff prices are exogenous to the model. We consider only whole-home, technology-agnostic tariffs that have been implemented widely outside the US, with consumption measured using a single utility meter. In the equation descriptions below, "revenue requirement allocation" refers to the portion of the revenue requirement allocated to that charge type (i.e., fixed, per-kWh, or capacity).

5.2.1 Fixed Charges

The fixed tariff is simply the revenue requirement allocation divided by the number of grid users.

5.2.2 Per-kWh Charges

The 1-part per-kWh tariff is the revenue requirement allocation divided by aggregate annual consumption. The 2-part TOU per-kWh tariff (we do not consider 3- or 4-part TOU) is calculated using Equation 5.1. The on-peak price is set at two times the off-peak price, which is common among US utilities [34].

$$Rate_{off-peak} = RevReq_{per-kWh} / (Cons_{off-peak} + 2 * Cons_{on-peak})$$
(5.1)

 $Rate_{on-peak} = 2 * Rate_{off-peak}$ (5.2)

Tariff Type	Variant	Periods	Illustrative Price	Description
Fixed	-	-	\$1000/year	Each consumer pays a fixed amount regardless of consump- tion. May be the same for all customers or differentiated based on, e.g., income
Per-kWh	Non-Time- differentiated (Flat)	1	\$0.11/kWh all hours	Per-kWh charge with no time variation
	Time- differentiated (TOU)	2+	\$0.08/kWh off-peak \$0.16/kWh on-peak	Per-kWh charge where total cost is the consumption in each pe- riod multiplied by the per-kWh price in that period. The on- peak charge is typically set at least two times the off-peak charge
Capacity	Ex-post measured (demand charge)	1+	\$170/kW-year	Assessed based on the con- sumer's maximum measured de- mand during each billed period
	Ex-ante contracted (subscription charge)	1+	\$150/kW-year	Consumer subscribes in advance to maximum threshold that can- not be exceeded

Table 5.1: Summary of distribution network tariff types with illustrative prices

where $Cons_p$ is the aggregated off-peak consumption during period p (on-peak or off-peak), $Rate_p$ is the per-kWh charge (kWh) during period p, and $RevReq_{per-kWh}$ is the revenue requirement allocation.

5.2.3 Capacity Charges

We consider two types of capacity (per-kW) tariffs: ex-post measured (demand) and ex-ante contracted (subscription).

Ex-Post Measured (Demand Charges)

An ex-post measured capacity charge (also called a demand charge) involves measuring the maximum demand during the billing period. In practice, it is calculated either by: 1) averaging several instantaneous demand readings during each hour and taking the maximum of those averages; or 2) taking the maximum instantaneous reading. In our case studies, this implementation detail is not relevant because we work with hourly consumption data.

We test three different demand charges in our case studies: 1-part, 4-part (with seasonality), and monthly (i.e., 12-part). The 1-part demand charge (DC) is computed by dividing the revenue requirement allocation by the sum of each household's maximum annual demand.

$$DC = RevReq_{capacity} / \left(\sum_{j}^{n} max_{i \in \{1,\dots,8760\}} \left(demand_{j,i}\right)\right)$$
(5.3)

where n is the total number of households and $demand_{j,i}$ is household j's total demand in hour i. The resulting 1-part demand charge (DC) has the units of k; each customer's demand-based network cost is that value multiplied by their maximum demand across all hours of the year.

We elaborate on the simple 1-part demand charge by adding time differentiation, both intraday (on-peak and off-peak) and seasonal (winter and summer). The hours and months contained in each period are unique to each case study, explained in Section 5.4. Under the 4-part demand charge, households pay a separate charge for their maximum demand in each of the following periods:

- Winter On-peak (WN)
- Winter Off-peak (WF)
- Summer On-peak (SN)
- Summer Off-peak (SF)

To calculate the price for each period, we use a similar methodology as the 1-part demand charge but apportion equal parts of the revenue requirement allocation to each seasonal period (i.e., demand charges for the winter period collect one half of the revenue requirement allocation and demand charges for the summer period collect the other half). Within each season, we set the on-peak charge to be two times the off-peak charge.

$$DC_{WF} = \frac{1}{2} Rev Req_{capacity} / \left(\sum_{j}^{n} max_{i \in \{WF\}} \left(demand_{j,i} \right) + 2 \sum_{j}^{n} max_{i \in \{WN\}} \left(demand_{j,i} \right) \right)$$

$$(5.4)$$

$$DC_{SF} = \frac{1}{2} Rev Req_{capacity} / \left(\sum_{j}^{n} max_{i \in \{SF\}} \left(demand_{j,i} \right) + 2 \sum_{j}^{n} max_{i \in \{SN\}} \left(demand_{j,i} \right) \right)$$

$$(5.5)$$

$$DC_{WN} = 2 * DC_{WF} \tag{5.6}$$

$$DC_{SN} = 2 * DC_{SF} \tag{5.7}$$

where $i \in \{WF\}$ indicates hours *i* contained in the set of winter off-peak hours $\{WF\}$ and defined similarly for the other periods.

We design the monthly demand charge to be equal for all months of the year:

$$DC_{monthly} = RevReq_{capacity} / \left(\sum_{m=1}^{12} \sum_{j}^{n} max_{i \in \{m\}} \left(demand_{j,i}\right)\right)$$
(5.8)

where $i \in \{m\}$ indicates hours *i* contained in month *m*.

Note that while demand charges are typically implemented as monthly charges, this is done to align with traditional utility billing cycles. The drivers of network costs are not bound by calendar months.

Ex-Ante Contracted (Subscription Charges)

Unlike a demand charge, in a subscription charge, the consumer selects a capacity level in advance. This works similarly to an Internet plan where customers select their bandwidth from a menu of options. In some implementations, if the customer exceeds the subscribed level, their power is cut off temporarily; in other cases, a penalty charge is imposed. Customers can determine their contracted (i.e., subscribed) level by looking back at their historical consumption.

To calculate the subscription tariff charges, we run the optimization under the demand charge case, which provides the maximum demand in each period. We calculate the subscription level as the maximum usage in each period *plus a 1 kW buffer*, rounded to the nearest whole kW value. This is meant to mirror the exercise a household would do to determine its subscription level (albeit with historical consumption data rather than perfect foresight), with the buffer representing a small amount of aversion to exceeding the historical peak values and incurring a penalty or lost load. After the subscription level for the different periods is determined, we re-run the optimization but with a hard physical cap equal to the assigned subscription value per time period. The subscription charge (SC) in each seasonal period pis calculated in Equation 5.9 (with on-peak subscription charge set equal to two times the off-peak subscription charge in each period, as with demand charges and per-kWh charges):

$$SC_p = K_p * RevReq_{capacity} / \left(\sum_{j}^{n} subscription_{j,p,off-peak} + 2\sum_{j}^{n} subscription_{j,p,on-peak}\right)$$
(5.9)

where K is the cost allocation factor for period p (e.g., 1/2) and subscription_{j,p,off-peak} is the pre-subscribed capacity level for household j in seasonal period p during off-peak hours (and defined similarly for on-peak hours). Because it is not possible for a household to exceed its pre-subscribed value in our model, we do not define a penalty cost function.

In our case studies, we envision the network tariffs as mandatory whereby grid users cannot opt out and select other tariff designs. Note that consumers could still choose their electricity supply tariff, either from the local distribution utility (in vertically integrated states) or a retail supplier (in states with retail choice). In our description of the case studies below, we provide the full set of tariffs considered in each case.

5.3 Analytical Methods

5.3.1 Optimization Formulations

We simulate each household's electricity consumption using a mixed integer linear program. The objective is to minimize the annual electricity cost, inclusive of network and energy costs.² EV chargers can vary kW demand while cars are plugged in. We assume that households respond rationally to price signals (with perfect foresight) and that appliance loads unrelated to EV charging and CCHP systems are inelastic (i.e., not price responsive). While there is ample evidence that residential consumers will make behavioral changes in response to electricity tariffs (e.g., [102]), these are relatively small compared to the flexible loads we consider in this thesis. In this way, our analysis envisions a "lower bound" scenario, with no change in consumer behavior outside of electrification-driven load.

In our first case study, we use a simplified optimization model where only EV charging is price-responsive. The model is implemented in the Python programming language using the Gurobi solver. The objective function for each household is defined as:

$$min\sum_{i}^{8760} \{c_i * (N_i + E_i)\} + \sum_{p} \{max_i ((c_i + A_i) * B_{i,p}) * DC_p\} + \sum_{i}^{8760} (maxsoc - soc_i) * 1e^{-4}$$
(5.10)

where:

- c_i is the charging value for each hour *i* (the decision variable)
- A_i is the non-EV load for each hour i
- N_i is the per-kWh network charge (in k/kWh) for each hour *i*
- E_i is the per-kWh energy charge (in k/kWh) for each hour *i*
- DC_p is the demand charge for each demand period p
- $B_{i,p}$ is a binary variable that indicates whether an hour *i* is part of a demand period *p*
- *maxsoc* is the battery's capacity
- soc_i is the energy in the EV battery for each hour i

The last part of the objective function ensures that EVs charge as early as possible such that the cost of charging does not increase. This is achieved by assessing a small penalty function (several orders of magnitude smaller than the per-kWh or capacity charges) for any hour when the battery is not fully charged. This reflects the empirically-observed behavior that EV owners overwhelmingly schedule charging to happen at the beginning of the low-price off-peak period [124], [128].

The model is subject to the following constraints:

$$minsoc < soc_i < maxsoc \tag{5.11}$$

$$0 \le c_i \le V S_i * 7.2 \tag{5.12}$$

$$soc_i = soc_{i-1} + (c_i - D_i)$$
 (5.13)

 $soc_0 = maxsoc$ (5.14)

$$soc_{8760} = minsoc \tag{5.15}$$

²For simplicity, we omit customer charges and policy-related charges. Assuming these would be continue to be collected via flat volumetric rates, this omission has no impact on optimal load profiles.

The first constraint ensures that the battery's state of charge never goes above the battery's capacity or below a minimum state of charge, set at 20% of the capacity. The second constraint limits EV charging to the power of a standard Level 2 residential charger, prohibits negative charging,³ and guarantees that the car can only charge when the vehicle is at home; VS_i is a binary status variable equal to 1 when the vehicle is at home and equal to 0 when the vehicle is away from home. The third constraint specifies that at each hour, the state of charge of the battery is equal to its state of charge in the previous time step plus the net charging value; D_i is the amount of energy drawn from the battery, set equal to the full day's electricity consumption in the hour the vehicle departs home (in order to ensure that the vehicle is sufficiently charged before departure). Finally, the fourth and fifth constraints set the vehicle's initial and final battery states of charge. For subscription charge scenarios, the objective function omits the middle summation of Equation 5.10; the ex-ante subscription level becomes a constraint.

Our second case study uses a model developed by several researchers at the MIT Energy Initiative [138], [139]: Distributed Energy Consumption in Actively Responsive Buildings (DECARB). The DECARB model is a mixed integer linear program that minimizes a building's total energy costs. DECARB is written in the Julia programming language and can be configured to allow the building owner to invest in new technology, for example solar photovoltaic and battery storage. However, in this thesis, we define technology adoption exogenously and specify each household's appliances ex-ante based on the level of electrification. The key inputs to DECARB are building metadata obtained from ResStock (including square footage, thermal properties of the walls and floors, and time series consumption for lights and appliances), outdoor air temperature, a catalog of installed equipment, price data, and thermal comfort bands.

DECARB minimizes cost by shifting the timing of electricity consumption within comfort limits. For HVAC systems, this might mean, e.g., pre-heating or pre-cooling the building during low-price hours and "riding through" high-price hours as the temperature approaches the comfort limit. As part of this thesis, we developed a new EV module for DECARB to allow EV charging to be co-optimized alongside other household demands. This is particularly important for capacity-based tariffs, where optimal charging behavior depends on the household's aggregate consumption.

5.3.2 Feedback Loop for Recovering Network Investments and Recomputing Charges

In this thesis, we employ an iterative approach to study the interaction between distribution network tariff design, network investment, and resulting cost impacts on different customer groups. The approach is meant to roughly approximate how a regulator would set rates based on forward-looking demand forecasts. We define electrification exogenously; the order in which houses are selected for electrification is random, cumulative, and proceeds in 5% increments. In other words, all EV/CCHP households in the 5% electrification scenario remain EV/CCHP households in the 10% scenario.

³While "vehicle-to-grid" (V2G) holds promise, the technology is immature and it remains to be seen whether automakers will allow it under battery warranty programs.

At each step in the electrification adoption curve, we simulate (in a bottom-up fashion) price-responsive household electricity consumption as described above and calculate demanddriven network investments. ⁴ We assume network investments are linearly correlated with aggregate annual peak demand, a limitation we discuss in Section 7.3. Network investments add to the revenue requirement, the amount recovered from consumers via the network tariff. For simplicity, we assume that there is no return on investment. Equation 5.16 describes the linear relationship between the aggregate peak demand and revenue requirement.

$$RevReq = BNC + LRMC * (SP_t - SP_0)$$
(5.16)

LRMC is the long-run marginal cost of expanding the network. We consider two cases: low-cost network expansion (\$100/kW) and high-cost network expansion (\$500/kW).⁵ These values are obtained from Lee [139], which takes an average of marginal distribution cost studies in various parts of the US. We treat these values as representative endpoints of a cost spectrum, rather than exact predictions. The baseline network cost (BNC) is defined as the product of the annual aggregated consumption at 0% EV/CCHP adoption times the current flat per-kWh network charge for the distribution utility in each region. We assume that all distribution network costs are allocated to the existing flat per-kWh network charge (i.e., no distribution-related costs are recovered in other line items). SP_t is the aggregate system peak for the electrification level t and SP_0 is the aggregate system peak at 0% electrification. The tariff price levels are then updated (via the equations in the Section 5.2) to collect the new revenue requirement.

For each incremental level of electrification, we first compute a naive solution where the tariff price levels are set to collect the revenue requirement at 0% EV adoption. Using the optimized load profiles from the naive run, we obtain the annual peak demand and new revenue requirement (from 5.16) We then recalculate the tariff prices to recover the new revenue requirement and re-run the optimization. Equilibrium is reached when the household responses do not deviate from the previous iteration and the full revenue requirement is collected. Figure 5.2 provides a visual overview of our methodology.

For all case studies, tariffs are recomputed only at discrete time steps (5% adoption intervals). This is meant to approximate the "regulatory lag" that describes the interval between utility rate cases. Unlike some studies(e.g., [116], [117]), we do not conduct a bilevel optimization where the utility minimizes its total costs while consumers minimize their bills. In the US, distribution utilities have no explicit financial incentive to minimize costs.

⁴While we consider distribution at only one layer, in reality it is a cascade of layers [111], with impacts aggregating up, correcting for increasing diversity through the entire distribution hierarchy [16]. The problem of correlated EV and CCHP operation is especially apparent at the lowest distribution layers, especially residential feeders with low load diversity. If we considered more layers, the impact of correlated consumption would be less pronounced; the higher you aggregate up the more you can leverage load diversity. In this way, our approach considers the "weakest link" of the cascade.

⁵Both values are annualized over the lifetime of the equipment entered into service.



Figure 5.2: Illustrative flowchart of methodology; ResStock is a database of synthetic hourly load profiles for representative US homes. NHTS is the National Household Travel Survey, in which respondents log their travel behavior.

5.4 Case Study Outlines

In this section we provide key assumptions, design decisions, and tariffs considered for our two case studies. The following design details are the same in both case studies:

- For the TOU energy tariff, we set the on- and off-peak prices to be revenue neutral with a flat per-kWh tariff at 0% EV/CCHP adoption. The on-peak price is two times the off-peak price (as in Equation 5.1). The TOU energy tariff prices remain the same at all levels of EV adoption (a limitation we discuss in Section 7.3).
- We use the same on and off peak hours for both network and energy tariffs.
- We assume electrical panel capacities are sufficiently large to accommodate new demand from EVs and CCHPs.

5.4.1 Greensboro (North Carolina): Price-Responsive EV Charging in a Warm-Climate Geography with a Detailed Network Model

For this case study, we select one distribution feeder from the Greensboro, NC Smart-DS dataset (feeder rhs2_1247-rdt1262). The feeder has 376 buildings, 339 of which are residential. We omit the non-residential buildings from this analysis. The feeder has an annual peak demand of 1.6 MW. We take the ResStock time series profiles assigned to each building as the base load and, at 5% adoption increments, assign to each household an EV (from North Carolina's NHTS responses) whose driving profile is calculated using the approach outlined in Section 5.1.2. EV charging is optimized under the network and energy tariffs listed in Table 5.2.⁶ All simulations were run on a 2.3 GHz Dual-Core Intel Core i5 laptop computer.

The Smart-DS dataset does not provide metadata or thermal time series data for the ResStock profiles assigned to each building. Because we do not have information on existing heating equipment, square footage, or total heating load, it is not possible to "convert" each building to electric heating. However, North Carolina has a warm climate and many buildings already heat with electricity [140], [8]. We believe it is reasonable to assume that the feeder will not see a significant increase in demand due to CCHP adoption, making EVs the dominant source of new residential load. One area for future work is to match the ResStock timeseries profiles in Smart-DS with timeseries data for buildings in public ResStock releases for which metadata is available; this would allow consideration of both heating and transportation electrification on realistic network topologies.

To set the initial tariff prices, we use Duke Energy's standard residential tariff (RS), which has a customer charge of \$14.00 per month and a per-kWh charge of \$0.114311/kWh. Duke Energy does not differentiate between network and energy charges on the bill; we assume that 50% of the per-kWh charge recovers network costs. We use Duke's residential TOU (RT) tariff to set the on- and off-peak hours. On-peak includes weekdays 6:00 AM - 9:00 AM and 4:00 PM - 9:00 PM whereas all other hours as off-peak. For tariffs with

⁶Abbreviations in Tables 5.2 and 5.3: DC = Demand Charge, SC = Subscription Charge, WN = Winter On-Peak, WF = Winter Off-Peak, SN = Summer On-Peak, SF = Summer Off-Peak

seasonal periods, we use the RT tariff's seasonal delineations: May - September (Summer) and October - April (Winter).

We consider two types of per-kWh energy tariff (flat and TOU) and six types of distribution network tariff (fixed, flat per-kWh, TOU per-kWh, 1-part demand charge, 4-part demand charge, and 4-part subscription charge). For the fixed and capacity tariffs, we collect one half of the total network revenue requirement via a flat per-kWh charge and the other half via the fixed or capacity charge. This design decision is meant to temper extreme distributional impacts and ease the transition away from the status quo.

Table 5.2 shows the combinations of energy and distribution network tariffs considered in the Greensboro case study. We omit the *eflat_ntou* and *etou_nflat* tariffs because they are duplicative of results from other scenarios.

5.4.2 Massachusetts: Price-Responsive EV and CCHP Operation in a Cold-Climate Geography with a Simplified Network Model

For this case study, we select 300 ResStock profiles from suburban counties in the Greater Boston Area, where heating is dominated by natural gas and both EV and CCHP penetrations are low today. In this case study, we have metadata for all households, which we use to filter for ResStock house archetypes with natural gas heating and floor area between 1500 -2500 square feet. We select driving profiles for households by matching occupant and income information between the ResStock archetype and NHTS survey responses filtered for Massachusetts. For households selected for electrification at each step in the adoption curve, we assign both an EV and a CCHP system, then use DECARB to co-optimize their operation.⁷ The CCHP system is an air source ductless heat pump sized to meet the full annual heating load without backup heating. This design choice adheres to a Massachusetts whole-home CCHP rebate program, which requires participants to remove their backup heating system in order to claim the full rebate [143]. The thermal comfort band (i.e., setback temperature range) is 19C - 24C, regardless of occupancy.⁸ We use the same linear coefficient of performance curve for CCHPs as in Khorramfar et al. [72]. We allow the heat pump to be used for cooling regardless of whether the ResStock archetype had existing air conditioning, which reflects behavior from actual heat pump customers [144]. We assume that the 300 households are electrically connected in the same neighborhood and distribution grid. The annual baseline peak demand is 729 kW.

Unlike the Greensboro case study, in Massachusetts network and energy charges are separated on the bill (shown in Figure 1.2). We use Eversource's network tariff price (\$0.13486/kWh) and the author's energy supply charge (\$0.1481/kWh) as of April 1, 2024. Eversource does not currently offer TOU rates for residential customers. We set the onand off-peak hours to be equal to the Greensboro case, which aligns well with the highest demand hours for all households in the sample. For tariffs with seasonal periods, we define Winter as November - March and Summer as April - October.

⁷While it is unlikely that heat pump adoption would overlap perfectly with EV adoption [19], it is reasonable to assume some overlap [141], [142].

⁸A "smart" thermostat could learn occupancy patterns and adjust this setback range, but we do not model that possibility here.

Due to the computational demands of simulating each building's thermal performance, we tested five tariff scenarios in the Massachusetts case study, shown in Table 5.3. For the capacity distribution network tariff scenarios, we collect one third of the total network revenue requirement via a flat per-kWh charge and the other two thirds via the capacity charge.

All simulations were run using the Gurobi solver on the MIT SuperCloud [145].⁹ DE-CARB does not have support for subscription charges so we consider only demand charges for the capacity-based alternatives. For the 4-part Demand Charge, rather than a 2:1 onpeak:off-peak ratio within each seasonal period, we set a price of \$0/kW during the off-peak periods and recover each season's revenue requirement allocation entirely through the onpeak demand charge. This is intended to encourage pre-heating and pre-cooling. For the monthly demand charge, we set the price to be the same amount in each month, which reflects how demand charges are typically implemented in the US for commercial customers.

5.5 Key Metrics

We assess each tariff design based on three metrics: aggregate peak demand (which is linearly correlated with total network cost), levelized operating cost of new equipment for electrified households, and network cost impacts for non-electrified households who do not change their consumption patterns. Levelized operating costs of new equipment directly affect the total cost of ownership, which determines whether and how much customers will save compared to internal combustion engine vehicles and fossil-fueled heating systems. Non-electrified household costs capture the distributional impacts of each tariff and indicate whether each will be acceptable from an equity perspective, especially when considering that EV adoption until now has correlated strongly with income [146]. Ideally, only electrified households should pay the incremental network costs due to increased adoption (since non-electrified household loads remain the same in our model). Yet in practice and in our simulations, distribution network tariffs include not only forward-looking costs but also sunk costs of prior investments. The implications of this design choice are discussed further in Chapters 6 and 7.

We compute the following statistics for each tariff at each 5% increment in the electrification adoption curve. We provide a brief description and equation for each.

Annual Peak: the maximum hourly demand for the aggregated load over all consumers, measured in kW:

$$max_{i\in\{1,\dots,8760\}}AggDemand_i \tag{5.17}$$

where $AggDemand_i$ is the sum of all individual demands in hour *i*.

Levelized EV/CCHP operating cost: the average cost paid by EV/CCHP households for charging/heat pump operation, calculated as the total incremental costs divided by total

⁹We acknowledge the MIT SuperCloud and Lincoln Laboratory Supercomputing Center for providing HPC resources that have contributed to the research results reported within this thesis.

incremental consumption, measured in \$/kWh:

$$\sum_{j}^{n} E_j(Cost_{E,j} - Cost_{B,j}) / \sum_{j}^{n} E_j(Cons_{E,j} - Cons_{B,j})$$
(5.18)

where E_j is 1 if the household is electrified and 0 otherwise, $Cost_{E,j}$ and $Cons_{E,j}$ are the total post-electrification cost and consumption, respectively, and $Cost_{B,j}$ and $Cons_{B,j}$ are the total pre-electrification (i.e., baseline) cost and consumption, respectively.

Distributional impact: the average change in network cost (NWCost) paid by nonelectrified households compared to a scenario of flat per-kWh network and flat per-kWh energy tariffs at 0% electrification, i.e. the "status quo" (SQ), measured in percentage change:

$$\left(\sum_{j}^{n} (1 - E_j)(NWCost_j - NWCost_{SQ,j})/NWCost_{SQ,j}\right) / \sum_{j}^{n} (1 - E_j)$$
(5.19)

Under the assumptions outlined in this chapter, we isolate the impact of distribution network tariff design (and the interaction between energy and distribution network tariffs) on electrification adoption economics and distributional effects. These help determine which tariff designs offer the best chance of achieving beneficial electrification.

Alias	Energy Tariff	Distribution Network Tariff at 0% EV	Relevant Periods
eflat_nfixed	\$0.06/kWh	Fixed: \$34/month Per-kWh: \$.03/kWh	N/a
etou_nfixed	On-peak: \$0.09/kWh Off-peak: \$0.04/kWh	Fixed: \$34/month Per-kWh: \$.03/kWh	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours
$eflat_nflat$	0.06/kWh	Per-kWh: $.06/kWh$	N/a
etou_ntou	On-peak: \$0.09/kWh Off-peak: \$0.04/kWh	Per-kWh: On-peak: \$0.09/kWh Off-peak: \$0.04/kWh	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours
eflat_ndem1	0.06/kWh	Per-kWh: \$.03/kWh Capacity: \$50/kW DC	N/a
etou_ndem1	On-peak: $0.09/kWh$ Off-peak: $0.04/kWh$	Per-kWh: \$.03/kWh Capacity: \$50/kW DC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours
eflat_ndem4	\$0.06/kWh	Per-kWh: \$.03/kWh Capacity: WN: \$20/kW DC WF: \$10/kW DC SN: \$18/kW DC SF: \$9/kW DC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours Winter: October - April Summer: May - September
etou_ndem4	On-peak: \$0.09/kWh Off-peak: \$0.04/kWh	Per-kWh: \$.03/kWh Capacity: WN: \$20/kW DC WF: \$10/kW DC SN: \$18/kW DC SF: \$9/kW DC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours Winter: October - April Summer: May - September
eflat_nsub4	\$0.06/kWh	Per-kWh: \$.03/kWh Capacity: WN: \$17/kW SC WF: \$8/kW SC SN: \$16/kW SC SF: \$9/kW SC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours Winter: October - April Summer: May - September
etou_nsub4	On-peak: \$0.09/kWh Off-peak: \$0.04/kWh	Per-kWh: \$.03/kWh Capacity: WN: \$17/kW SC WF: \$8/kW SC SN: \$16/kW SC SF: \$9/kW SC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours Winter: October - April Summer: May - September

Table 5.2: Tariffs tested in the Greensboro case study

Alias	Energy Tariff	Distribution Network Tariff at 0% EV	Relevant Periods
eflat_nflat	\$0.15/kWh	Per-kWh: \$.13/kWh	N/a
etou_ntou	On-peak: \$0.23/kWh Off-peak: \$0.11/kWh	Per-kWh On-peak: \$0.20/kWh Off-peak: \$0.10/kWh	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours
etou_ndem1	On-peak: \$0.23/kWh Off-peak: \$0.11/kWh	Per-kWh: \$0.04/kWh Capacity: \$136/kW DC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours
etou_ndem4	On-peak: \$0.23/kWh Off-peak: \$0.11/kWh	Per-kWh: \$.04/kWh Capacity: WN: \$88/kW DC WF: \$0/kW DC SN: \$73/kW DC SF: \$0/kW DC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours Winter: November - March Summer: April - October
etou_ndem12	On-peak: \$0.23/kWh Off-peak: \$0.11/kWh	Per-kWh: \$.04/kWh Capacity: \$14/kW-mo DC	On-peak: M-F 6-9am, 4-9pm Off-peak: all other hours

Table 5.3: Tariffs tested in Massachusetts case study

Chapter 6

Results

This chapter is divided into two sections. First, we present the results for the Greensboro case study. Second, we present the results for the Massachusetts case study. For the levelized cost and change in network cost metrics, we present the low (100/kW) and high (500/kW) LRMC cases side-by-side. These values represent a range of possible incremental investment costs.

6.1 Results for Greensboro Case Study

We first discuss the results for the growth in annual peak under the different distribution network tariff designs, with one sensitivity analysis on household adherence to price signals. We then analyze the levelized charging cost for EV households. Finally, we discuss distributional impacts. For each metric we group the scenarios by type of energy tariff (flat or TOU) for legibility and to highlight the interaction between energy and distribution network tariff designs.

6.1.1 Greensboro: Annual Peak

Figure 6.1 shows the annual feeder-wide peak demand for each tariff at 5% EV adoption increments, with 0-30% adoption magnified. The annual peak is identical for both LRMC cases so we do not show them separately; under our assumption of perfect rationality, only relative price differences impact each household's response, not the absolute magnitude (which the LRMC affects). For example, if we assume an on-peak tariff price of 0.20/kWh, an off-peak tariff price of 0.19/kWh will induce the same EV response as a price of 0.10/kWh in our model. The *eflat_nfixed/ eflat_nflat* and *etou_nfixed/etou_ntou* tariffs produce the same result because the distribution network tariff either reinforces or has no distortionary effect on the energy tariff price differences.

Under the flat energy tariff, vehicles charge immediately upon arrival as there is no incentive to delay charging until a later time. This yields an increase in annual peak demand as early as 5% adoption. However, because of natural heterogeneity in vehicle arrival time, aggregate charging demand is spread out over evening hours and peak demand only grows modestly. Under the TOU energy tariff, the price differential induces all EVs that arrive



Figure 6.1: Annual peak demand for the Greensboro feeder at 5% EV adoption increments for distribution network tariffs combined with flat (left) and TOU (right) energy tariffs. Top shows full adoption range and bottom shows 0-30% to highlight when peak demand growth begins under each tariff.



Figure 6.2: Aggregate average weekday demand at 50% EV adoption under the *eflat_nflat* (left) and *etou_ntou* (right) tariff scenarios with tariff prices overlayed. Under the *eflat_nflat* tariff, EV owners charge immediately upon arrival. Under the *etou_ntou* tariff, EV owners schedule charging to delay until the start of the off-peak period.

prior to 9:00 PM to wait until the off-peak period to start charging. This phenomenon is illustrated in Figure 6.2 for 50% EV adoption.

Until 25% adoption, the TOU energy tariff induces charging to shift to hours where the network has excess capacity and so does not increase annual peak demand. However, at 25% adoption and beyond, the correlated response to the off-peak energy tariff produces an EV-driven "rebound" peak that exceeds the historical early-evening peak due to non-EV demand. This effect is most severe under the fixed and per-kWh distribution network tariffs where EV owners charge at the chargers' full capacity. In contrast, capacity and subscription tariffs provide an incentive to charge at a lower capacity to limit the household's maximum instantaneous demand. In this way, even though we still observe an EV-driven peak under the capacity tariffs, the peak is reduced, as shown in Figure 6.3. The flat energy tariff scenarios do not feature the same correlated response; for a given distribution network tariff (e.g., 1-part demand charge), the flat energy tariff produces a lower annual peak compared to the TOU energy tariff at all adoption levels beyond 25%.

For example, at 50% EV adoption, the annual peak under the $eflat_ndem1$ tariff is 1,347 kW (an 8% increase over the "baseline" peak demand at 0% EV adoption) compared to 1,432 kW under the $etou_ndem1$ tariff (a 15% increase over the baseline). The $etou_ntou$ tariff, which provides no incentive for EVs to charge below their maximum capacity at the start of the off-peak period, has an annual peak of 1,601 kW at 50% EV adoption (a 28% increase over the baseline).

We also observe that adding more time periods to the capacity charge improves its ability



Figure 6.3: Example of an individual household reacting a TOU energy tariff and subscription distribution network tariff (*etou_nsub4*) versus a purely volumetric TOU tariff (*etou_ntou*) (left) and hourly demand aggregated across all households during the annual peak day at 50% EV adoption (right). Note that the scales for the vertical demand axes are not equal. Under a subscription tariff, households still delay charging until the start of the off-peak energy period, but the peak is less pronounced because households have an incentive to manage charging to stay below its contracted capacity.

to mitigate peak impacts and thus to defer network upgrades. This is because under a single capacity period (*ndem1*), the combination of EV and non-EV demand can approach the household's *annual* peak demand (across all hours) without increasing the household's network charge payment. The introduction of intraday and seasonal variation sets a unique maximum level in each period. For most households, their maximum pre-EV demand during off-peak hours is lower than for on-peak hours. This sets a lower target for each household's aggregate demand as it varies the charging capacity level during off-peak hours. When all households behave this way (though still acting independently), it produces a lower annual peak. Under the subscription charge, we lose a portion of peak demand mitigation due to the 1 kW subscription buffer. However, overall, the subscription charge performs better than fixed or per-kWh distribution network tariffs when coupled with either a flat or TOU energy tariff.

We conduct a sensitivity analysis that supposes a small portion of EV households (30%) ignore both the energy and distribution network tariff price signals and charge at full capacity upon returning home. This is designed to test how robust our tariff designs are to "irrational" consumer behavior. We examine the impact on annual peak demand, shown in Figure 6.4. As above, the annual peak values are the same in both LRMC cases, so we do not show them separately.



Figure 6.4: Annual peak demand for 100% (top) and 70% (bottom) compliance among EV owners with respect to tariff price signals. When a small number of EV owners deviate from rational behavior, the aggregate peak demand is lower than when all EV owners respond rationally in a correlated manner.

We obtain the somewhat surprising result that with TOU energy tariffs, when 30% of EV owners ignore the price signal and charge immediately upon arrival home at full capacity, the annual network peak is lower than when all EV owners comply. This reflects the fact that with 100% price responsiveness, all EVs that are plugged in charge at the start of the off-peak period. If a portion of EV owners charge immediately upon returning home rather than delaying until lower-priced hours, this introduces heterogeneity that reduces the EV-driven peak demand. This result illustrates that the benefits of capacity charges are not dependent on all consumers responding to them. Note that the effect would be similar if a portion of EV customers elected a flat energy tariff (if that were an option) and reacted to that price signal alone.

For the flat energy tariff, the results are almost identical at 100% and 70% compliance. Under the capacity tariffs, the "irrational" EV households do not manage their charging to minimize their peak demand. Yet if charging at a higher capacity for a shorter period allows them to fulfill their charging needs prior to the start of the off-peak period, this will reduce the magnitude of the "rebound" peak (at the start of the off-peak capacity charge period) as fewer vehicles will be charging then. These phenomena nearly offset to produce a similar aggregate peak for all flat energy tariff scenarios for 100% and 70% compliance.

6.1.2 Greensboro: Levelized EV Charging Cost

Figure 6.5 shows the average levelized charging costs for EV owners under the two energy tariff options and two LRMC cases. Almost all EV households can achieve the lowest theoretical charging cost. This means that they fulfill their driving needs (charging sufficiently to meet usage demand each day) and adhere perfectly to price signals. For the per-kWh tariff component, nearly all EV charging occurs during the off-peak period. For capacity tariffs, EV households manage charging so as not to exceed their pre-EV peak demand in each demand period (or stay beneath their contracted capacity in the subscription tariff). This result is likely an underestimate of charging costs due to our assumption that households can perfectly forecast their individual peak usage. The result for the subscription charge provides a more realistic estimate because customers would know their subscription level in advance and then program EV charging to stay beneath that limit.

Per-kWh distribution network tariffs perform poorly from a levelized charging cost perspective. When the network revenue requirement is collected using a flat per-kWh tariff, EV households contribute more to network cost recovery than under other tariff designs as they have no mechanism to reduce their network costs by shifting demand to off-peak hours. The TOU per-kWh distribution network tariff helps to some extent to lower the levelized costs of charging, but they remain higher than under fixed and capacity-based designs. The upward curve of the values under the high LRMC case reflects the fact that the revenue requirement (driven by annual peak demand) is growing faster than aggregate electricity consumption, which applies upward pressure on the magnitude of the network charge. The energy tariff is exogenous so remains the same at all levels of adoption. The consistent upward curve beyond 20% EV adoption is eliminated in the low LRMC case, where load growth keeps pace with the cost of network upgrades.

Under the subscription distribution network tariff, there is variation in the levelized charging cost paid by EV households, shown in Figure 6.6. This reflects the fact that some



Figure 6.5: Levelized charging costs for EV owners for low (top) and high (bottom) LRMC cases with flat energy tariffs (left) and TOU energy tariffs (right). Almost all EV households can fulfill their charging needs without increasing peak demand or charging during the on-peak per-kWh window. This indicates the high level of flexibility in residential charging activity as cars are plugged in for many hours longer than their charging needs.



Figure 6.6: Distribution in levelized costs paid by EV households at 50% EV adoption for the $etou_nsub4$ tariff scenario.

households must increase their subscription level to allow for sufficient charging time to meet their charging needs.

6.1.3 Greensboro: Difference in Network Costs for Non-EV Households

Figure 6.7 shows the change in annual network costs paid by non-EV households under each distribution network tariff design for both LRMC cases. Positive values reflect cost increases and negative values reflect savings compared to network costs at 0% EV adoption with flat volumetric energy and flat distribution network tariffs (i.e., the baseline).¹ We focus on network costs because those are non-optional in states both with and without retail choice for energy supply. A consumer whose energy tariff costs increase under a TOU plan might be able to opt out and switch to a flat per-kWh energy rate, something we expect to occur in practice, though we do not model this possibility.

An (undifferentiated) fixed distribution network tariff paired with a TOU energy tariff performs the worst at all EV adoption levels, as the majority of new network costs (that are entirely driven by EV charging) are shared equally among all households.² Annual peak demand is also highest under fixed charges (see Figure 6.1), as there is no incentive for households to manage charging to limit their maximum demand. This effect is less pronounced under a flat energy tariff whereby charging is not correlated and the peak grows

¹While EVs account for 1.8% of registered vehicles in North Carolina [147], for simplicity we assume 0% as the baseline.

²Though the costs are shared equally, this may be the least equitable outcome in the sense that non-EV households are not responsible for any of the increase in network costs. Included in Joroff's definition of energy justice is "protection from a disproportionate share of costs or negative impacts or externalities associated with building, operating, and maintaining electric power generation, transmission, and distribution systems." [148]



Figure 6.7: Change in network costs for non-EV households for the low (top) and high (bottom) LRMC cases.



Figure 6.8: Distribution in the change in network cost among non-EV households at 50% EV adoption for the *etou* nsub4 tariff scenario.

more modestly. Including a flat per-kWh charge to collect half the network revenue requirement ensures that even under capacity and fixed tariffs, EV households are contributing to cost recovery at a higher share than prior to EV adoption. Without this charge, the negative impacts on non-EV households would be even more pronounced.

The 4-part demand charge performs best, in large part because the overall revenue requirement that needs to be collected is lowest among all scenarios. While the subscription tariff results in higher revenue collection from EV households compared to 1-part and 4-part demand charges, this benefit to non-EV households is offset partially by the higher annual peak under the subscription tariff.

This metric is highly sensitive to the assumed cost of network expansion. At 50% adoption and an LRMC of \$100/kW, the average non-EV household pays at most 5% more than the baseline (with the *etou_nfixed* tariff). Under the flat energy tariff and the capacity network tariffs, non-EV households experience lower costs on average, achieving true "beneficial electrification" whereby the adoption of EVs applies downward rate pressure.

At 500/kW, only at low EV adoption levels do non-EV households see non-negative impacts. At 50% EV adoption, non-EV households pay between 9% and 54% more than their baseline costs so that the utility can recover network upgrades caused by EV charging.

We find significant variability in cost impacts among non-EV households, shown in Figure 6.8. "Peakier" households (those with a high peak demand relative to their annual consumption) will pay more under the capacity tariffs, assuming their consumption patterns do not change.

Undesirable distributional impacts for certain groups (e.g., low-income households) could be mitigated using an ex-post subsidy, but this would need to be funded by other ratepayers or through general taxation; both options would increase the financial burden on other customers [115].
6.2 Results for Massachusetts Case Study

The Massachusetts case study considers electrification of both residential heating and vehicles. CCHPs represent a significant load that is inversely correlated with temperature, especially during winter in cold regions, illustrated in Figure 6.9.



Figure 6.9: Aggregate demand by load type at 50% electrification under the $eflat_nflat$ tariff for a cold week (left) and mild shoulder season week (right). During the cold week CCHP demand accounts for 51% of total consumption, compared to 27% during the shoulder week. Note that the axes are not equal.

Like in the Greensboro case, we present the results for annual peak first, levelized operating cost (inclusive of both EV and CCHP) second, and change in network costs for non-electrified households third.

6.2.1 Massachusetts: Annual Peak

Figure 6.10 shows the annual feeder-wide peak demand for each tariff at 5% EV/CCHP adoption increments. The annual peak is identical for both LRMC cases so we do not show them separately. The slight downward dip at early adoption levels is due to the fact that the annual demand occurs in a summer month, and heat pumps cool more efficiently than central AC and window AC units. The circuit flips from a summer- to winter-peaking system at 20% adoption, beyond which there is a sharp rise in peak demand.

Similar to the Greensboro case study, the *etou_ntou* tariff performs worst at all adoption levels starting with 20% due to the correlated response of all EVs starting to charge at the start of the off-peak period with no signal to limit each household's maximum demand. The annual peak is nearly identical for the remaining tariffs. While the demand charge has a mitigating effect on correlated EV charging, the maximum feeder demand under the TOU energy tariff scenarios is driven by electric heating demand during an hour with an outdoor air temperature of -15C (5F).



Figure 6.10: Annual aggregate peak demand for the Massachusetts feeder at 5% electrification increments. Beyond very low electrification levels, the peak is driven by electric heating during extreme cold temperatures where demand charges have a negligible mitigating impact.

At extremely cold temperatures, CCHPs must consume at a high capacity to maintain the minimum comfort temperature; neither TOU energy tariffs nor demand charges are effective at shifting consumption during these times. At more moderate temperatures, CCHPs are more flexible; the TOU energy tariff induces a shift in consumption to take advantage of low-price hours by pre-heating and pre-cooling and the demand charge charge incentivizes households to operate their heat pumps at a high load factor rather than running equipment at high capacity for short periods of time.

Figure 6.11 shows a load duration curve at 50% adoption for the Massachusetts feeder compared to the Greensboro feeder for the *etou_ndem1* tariff scenario. Load duration curves show the demand at each hour of the year ranked in descending order. It is purely coincidental that both feeders have an annual peak demand of around 1400 kW. The color of the bar indicates the percentage of the demand due to EV charging. For Greensboro, among the top 25 load hours, 22 of them (88%) occur at weekdays at 9:00 PM (the start of the off-peak window), mostly during summer months. For Massachusetts, only 7 of the top 25 load hours (28%) occur during weekdays at 9:00 PM, and all top 25 hours occur during a cold snap in January. This gap illustrates how geography impacts whether peaks will be "EV-driven" or "CCHP-driven." While capacity-based distribution network tariffs help reduce daily EV peak demand regardless of geography, in cold climates, capacity charges may have only a marginal impact on the annual CCHP-driven peak, which will drive network investment.



Figure 6.11: Load duration curves at 50% electrification under the *etou_ndem1* tariff for the Greensboro (a) and Massachusetts (b) case studies. For the Massachusetts feeder, electric heating is the primary cause for high-demand hours. For the Greensboro feeder, it is EV charging.

6.2.2 Massachusetts: Levelized EV/CCHP Costs

Figure 6.12 shows the levelized cost of EV and CCHP operation for electrified households. We report a combined value that indicates the total change in cost post-electrification divided by total change in consumption, averaged across all electrified households. Similar to the Greensboro case study, we see that time-differentiated capacity tariffs achieve reductions in the levelized cost compared to the status quo. For example, at 50% adoption and 100/kW LRMC, the *etou_ndem12* tariff has a levelized cost 0.09/kWh lower than the *eflat_nflat* tariff, a reduction of 25%. As a point of reference, a 0.09/kWh difference has the same impact on operating charging costs for EVs as a 0.70/gallon difference for gasoline vehicles. Furthermore, under the low LRMC scenario, levelized costs decrease over time, which reflects the downward rate pressure as electricity consumption grows faster than network investments. While electrification is defined exogenously in our model, in reality lower electricity costs would likely accelerate electrification [19], lowering costs further in a positive feedback loop.

The 1-part demand charge performs notably worse than the time-differentiated demand charges. This occurs because electrified households pay based on their maximum demand across the whole year, which is likely to occur during extreme cold temperature periods when their CCHP systems are operating least efficiently. Time-differentiated capacity charges in which customers pay based on their maximum demand in each period lessen the influence of high winter heating demand on annual costs. We note that this time differentiation would be less effective if a larger portion of network costs were collected in the winter period (for the ndem4 tariff) or in the winter months (for the ndem12 tariff) rather than collecting an equal share of the revenue requirement in each season or implementing a single monthly capacity charge.

³This assumes the US average EV and gasoline efficiencies [22].



Figure 6.12: Levelized cost of EV and CCHP operation at 5% electrification adoption increments for the low (left) and high (right) LRMC cases.

6.2.3 Massachusetts: Difference in Network Costs for Non-Electrified Households

Figure 6.13 shows the change in network costs for non-electrified households. Under the low LRMC case, non-EV households experience cost reductions under all tariffs at almost all electrification levels. In this case, the growth in annual peak demand (and associated network investment) is offset and surpassed by load growth due to EVs and CCHPs. For the high LRMC case, this phenomenon is only true at low adoption levels when the annual peak demand is stagnant. Non-electrified households face steep rises in costs under scenarios where households can reduce their network contributions by shifting the timing of EV and CCHP consumption. Unlike in the Greensboro case study, where these actions helped avoid network costs increases, in the Massachusetts case study the demand charge has a negligible effect on annual peak compared to the *eflat nflat* tariff. The 1-part demand charge is the most cost-reflective of the three demand charge options. Electrified households contribute at the highest share to incremental investment costs due to an increase in their annual individual peak demand driven by CCHP heating. With time-differentiated capacity tariffs, the annual peak is unaffected but electrified households contribute less towards the revenue requirement because they can reduce demand during non-winter periods. This shifts a larger portion of incremental investment costs to non-electrified households compared to the 1-part demand charge.

In this chapter, we presented the results from two realistic case studies addressing the long-term impacts of distribution network tariff design on network investments, electrification costs, and distributional outcomes for non-electrified households. The next chapter discusses the implications of our findings.



Figure 6.13: Change in network costs for non-electrified households at 5% EV/CCHP adoption increments for the low (left) and high (right) LRMC cases. For the low LRMC case, cost reductions are achieved for all tariffs through 50% EV adoption. For the high LRMC case, this is only true at low adoption levels before network costs steeply rise.

Chapter 7

Discussion

In this chapter, we first provide a brief assessment of the results presented above. Next, we discuss tradeoffs in tariff design and implications of our findings. Finally, we discuss limitations of our modeling approach.

7.1 Assessment of Model Results

Tables 7.1 and 7.2 provide a summary of the key metrics in the Greensboro and Massachusetts case studies, respectively, at 50% electrification. Note that 50% adoption in concentrated areas of the grid can happen earlier than statewide 50% adoption. In each table, colors indicate how well each tariff performs (green being the best, red being the worst). Colors are assigned independently in each column to reflect that the two LRMC cases do not interact.

Scenario	Energy Tariff	Network Tariff	Annual Peak (kW)		Levelized Charging Cost (\$/kWh)		Change in Network Cost for non-EV homes (%)	
			\$100/kW	\$500/kW	\$100/kW	\$500/kW	\$100/kW	\$500/kW
eflat_nfixed	Flat	Fixed	1413		\$0.08	\$0.09	-1%	22%
eflat_nflat	Flat	1-part Per-kWh	1413		\$0.11	\$0.12	-5%	18%
eflat_ndem1	Flat	1-part Demand Charge	1347		\$0.08	\$0.09	-3%	11%
eflat_ndem4	Flat	4-part Demand Charge	1341		\$0.08	\$0.09	-4%	10%
eflat_nsub4	Flat	4-part Subscription Charge	1360		\$0.10	\$0.10	-6%	10%
etou_nfixed	TOU	Fixed	1601		\$0.07	\$0.09	5%	54%
etou_ntou	TOU	2-part TOU Per-kWh	1601		\$0.09	\$0.11	4%	52%
etou_ndem1	TOU	1-part Demand Charge	1432		\$0.07	\$0.08	0%	26%
etou_ndem4	TOU	4-part Demand Charge	1403		\$0.07	\$0.08	-2%	20%
etou_nsub4	TOU	4-part Subscription Charge	1469		\$0.08	\$0.09	0%	30%

Table 7.1: Key metrics in the Greensboro case study for each tariff at 50% EV adoption for low and high LRMC cases

The results in Tables 7.1 and 7.2 indicate a trade-off in assessment criteria. For example, in the Greensboro case study, the *eflat_nflat* tariff performs well for the change in network costs for non-electrified homes but poorly in levelized operating costs as EV and CCHP

Scenario	Energy Tariff	Network Tariff	Annual Peak (kW)		Levelized EV/CCHP Operating Cost (\$/kWh)		Change in Network Cost for non-electrified homes (%)	
			\$100/kW	\$500/kW	\$100/kW	\$500/kW	\$100/kW	\$500/kW
eflat_nflat	Flat	1-part Per-kWh	1353		\$0.25	\$0.31	-27%	20%
etou_ntou	TOU	2-part TOU Per-kWh	1725		\$0.23	\$0.33	-17%	61%
etou_ndem1	TOU	1-part Demand Charge	1424		\$0.21	\$0.27	-17%	41%
etou_ndem4	TOU	4-part Demand Charge	1410		\$0.19	\$0.24	-10%	52%
etou_ndem12	TOU	Monthly Demand Charge	1417		\$0.19	\$0.22	-5%	61%

Table 7.2: Key metrics in the Massachusetts case study for each tariff at 50% EV adoption for low and high LRMC cases

households have no opportunity to save money by shifting the timing of their consumption. The opposite is true for the *etou_nfixed* tariff, under which levelized operating costs are low but new network investments are shared equally among all households, resulting in the highest in network costs for non-electrified homes among all tariffs.

Capacity-based tariffs (demand and subscription charges) offer a compromise, providing a reduction in levelized charging cost compared to the per-kWh tariffs while shifting costs to non-electrified households by only a modest amount compared to the worst-performing fixed tariff. Capacity tariffs also mitigate annual peak compared to tariffs under which households have no incentive to limit the instantaneous demand of their EVs or CCHPs. This effect is more pronounced in the Greensboro case study in which peaks are EV-driven. In the Massachusetts case study (where peaks are CCHP-driven during extremely cold hours) capacity charges have only a small impact on annual peak demand at high levels of adoption. However, capacity tariffs still achieve significant reductions in levelized costs for EVs and CCHPs compared to per-kWh tariffs.

In the Greensboro case study, scenarios with flat energy tariffs broadly perform better than those with TOU energy tariffs for annual peak (and associated network investment costs). Despite this result, we do not believe regulators should attempt to retain flat energy tariffs, which we discuss below.

Our results also demonstrate the role of the long-run marginal cost in determining whether electrification will have an upward or downward effect on tariff prices. In service territories with high marginal network expansion costs, regulators may prefer tariffs that better protect non-electrified households even if this means higher levelized costs for new electrified load. In contrast, where marginal investment costs are low, regulators may prefer tariffs that reduce the operating costs of EVs and CCHPs to accelerate electrification, confident that the impact on non-electrified households will be beneficial. In the low-LRMC scenario, reducing costs of ownership for EVs and CCHPs (a priority for many US states) is wholly compatible with the pursuit of broader energy affordability goals.

7.2 Tariff Design Considerations

Our results suggest that there is no "one size fits all" solution for tariff design in all geographies. Approaches that effectively mitigate EV-driven rebound peaks in the Greensboro case study are not as effective when we consider the concurrent electrification of heating and transportation in a cold climate. In areas like the southeast US with mild winter temperatures where there is already high electric heating penetration [140], a time-differentiated capacity tariff where households have an incentive to minimize their peak demand at all hours appears to balance stakeholder interests while mitigating peak demand growth. On the other hand, in cold climates where heating-driven peaks are expected, a subscription charge may be counterproductive; regulators may wish instead to provide even stronger incentives than modeled in this thesis for households to pre-heat at high demand during the mid-day and overnight hours. This could mitigate CCHP-driven early morning and evening peaks. For example, they could create a "super off-peak" energy tariff period (i.e., \$0/kWh) or offer peak time rebates during extremely cold days.

As regulators design new distribution network tariffs, they should consider not only geographic factors (i.e. winter- or summer-peaking systems) but also the unique drivers for consumer adoption of EVs and CCHPs. While electricity cost is a factor in EV purchases, consumers rank upfront purchase price and public charging availability as much more significant adoption barriers [149]. Furthermore, Bushnell et al. [20] show that gasoline prices have a larger effect (four to six times) in California on EV adoption compared to electricity prices, and even under the flat volumetric tariffs prevalent today, electricity enjoys a significant operating cost advantage over gasoline in almost all US states [21]. For these reasons, it may be acceptable to design a tariff that does not reduce the levelized charging cost for EVs significantly (e.g., only a modest price differential between peak and off-peak per-kWh charges or recovering a higher percentage of the network cost via a per-kWh charge).

In contrast to EVs, Davis [19] finds CCHP adoption is strongly correlated with electricity prices. In states like Massachusetts and California where natural gas is the dominant space heating fuel [8], current (high) electricity prices render heat pump adoption uneconomic [27] even before equipment and installation costs are considered.¹ Maine, which has achieved the highest CCHP adoption rate among cold-weather states with over 100,000 systems installed, also has the highest percentage of homes heating with propane or fuel oil (65%) [8], which are expensive relative to piped gas in other parts of the country. Maine's space heating expenditures per household are sixth highest among all US states (6th) [8]. Reducing the operating costs for heat pumps is therefore essential to achieving decarbonization targets in states where gas is primarily used for heating. In the absence of explicit subsidies (e.g., Central Maine Power's seasonal heat pump rate), allocating a higher portion of network cost recovery to fixed or time-differentiated capacity charges (which in our Massachusetts case study reduces the operating costs by as much as 32% compared to the status quo) is one potential solution to this challenge.

A tariff that effectively both encourages high-demand pre-heating/pre-cooling for CCHP systems and mitigates EV rebound peaks may not be possible while meeting other tariff design objectives (especially simplicity). For some regulators, the distinct operating characteristics of EVs and CCHPs may suggest designing separate tariffs for each technology. While some utilities have offered EV-specific rates (with consumption measured by the ve-

¹Addressing the negative externalities of emissions directly (i.e., via a price on a carbon) would provide incentives to switch from gas; its social cost with a carbon price is higher than its private cost without a carbon price [150].

hicle or charger), this has not been attempted at scale for CCHP systems, and it remains unclear if it is technically possible without the use of a secondary meter. Furthermore, designing technology-specific rates would violate long-held norms in rate design and risk exacerbating distributional impacts if only households with either EVs or CCHP systems would be able to reduce their costs by shifting the timing of consumption. Tariffs should be technology-agnostic, non-discriminatory, and reflect expected consumer responses, including highly correlated responses due to load automation.

Finally, we note that the theoretically most efficient distribution network tariff consists of two parts: a capacity charge based on forward-looking costs and a differentiated fixed charge to recover sunk network costs. However, an income-based fixed charge proposal has faced significant opposition in California [151], [152], and we do not believe this to be a pragmatic path forward.² Instead, we recommend recovering a portion of the total revenue requirement via a flat per-kWh charge and a portion via a subscription charge³ (with the exact split determined by the regulator to balance policy objectives). This solution gives consumers more control over their costs compared to a static fixed charge, is a less drastic departure from the status quo, does not require utilities to collect income information, and has a mechanism (the flat per-kWh charge) to protect non-electrified households from increasing network costs. In the subsections below, we address additional tariff design considerations.

7.2.1 Addressing the Apparent Benefits of Flat Volumetric Tariffs

In the Greensboro case study, paradoxically, at high levels of EV adoption, the *eflat_nflat* scenario (the status quo today) achieves a lower annual peak demand than any distribution network tariff paired with TOU energy tariffs. When there is no incentive to delay charging, each vehicle charges upon arriving at home at the maximum charger capacity. Differences in arrival times introduce heterogeneity, and aggregate peak demand is lower than the correlated "snapback" or "rebound" we observe with a TOU energy tariff. This begs the question: if the status quo tariff performs so well, then why move away from it?

As Figures 6.5 and 6.12 show, the status quo tariff has one major drawback: the effect on the levelized costs of electrified devices. When households have no opportunity to save money by shifting demand, they pay the highest price among all scenarios at almost all electrification levels. While this may not be a significant barrier to EV adoption [149], households are unlikely to purchase CCHP systems if they cannot achieve a modest payback period on their investment [19], especially in retrofit applications where they are supplementing or replacing an existing heating system that may not have yet reached the end of its life. This requires moving away from flat volumetric tariffs, under which the payback period is infinite in some states, even with lucrative purchase incentives.

For each distribution network tariff, we observe better annual peak and distribution impact performance for the flat energy tariff. Yet flat energy tariffs may be unrealistic for three key reasons. First, in states with retail choice, retailers only control the energy tariff

²Opposition to the IGFC was not uniform. There are ideological objections to income related fixed charges. Rooftop PV interests objected because they expected that a high fixed charge would reduce the value of net metering. There were also general objections to "radical" changes and support for continuity in rate design.

³Below, we explain why we prefer subscription charges over demand charges.

component of the bill, and their procurement costs are tied to wholesale electricity prices. This creates an incentive for retailers to design energy tariffs that encourage customers to consume at hours with low wholesale prices regardless of the conditions on the local distribution grid. Regulators have no oversight over the design of energy tariffs by retail suppliers.

Second, in states without retail choice, there are other considerations besides those related to the distribution network that impact tariff design decisions, for example reflecting the short-run marginal cost of generation and enabling the interconnection of additional renewable energy [153]. Reflecting these conditions in the energy tariffs requires a time-varying signal, which may not align with local network congestion.

Third, flat energy tariffs represent an unstable equilibrium. While they produce a favorable global outcome (lower network costs), EV drivers will opt for TOU rates when they are available because the potential for savings is considerable, as shown by Borlaug et al. [154]. Almost all utilities offer opt-in TOU rates today. And while some utilities may believe that they can achieve demand shifting using only behavioral nudges, Bailey et al. [130] show that when financial incentives are removed, the timing of charging reverts to the original schedule (i.e., charging immediately upon returning home).

For these reasons, we do not believe that regulators should attempt to prevent timevarying energy tariffs in the interest of reducing local peak demand growth. Our results show that well-designed distribution network tariffs can mitigate some of the negative impacts of TOU energy tariffs.

7.2.2 Subscription versus Demand Charges

The Greensboro case study shows that when paired with an energy TOU tariff, a multipart demand charge is the most effective at reducing peak demand growth.⁴ This is partly the result of the perfect foresight that each household possesses when determining how to schedule its EV load. However, even if it were realistic to expect that demand charges would produce the most efficient results, we do not believe that states should adopt demand charges for residential consumers. Subscription charges (which do not perform badly on any of the key criteria in the Greensboro case study) offer implementation advantages over demand charges, as discussed below. As PUCs attempt to balance stakeholder interests in promoting electrification while addressing the problem of correlated rebound peaks, a tariff design that does not create big winners or losers may be the most palatable.

US utilities that have pushed back against time-varying rates often cite customer confusion and expected bill impacts as the reason for their opposition (see, e.g., [156]). Because utilities are penalized for customer confusion and bill shocks, utilities may oppose complicated tariff designs. Hennig et al. [119] acknowledge this fact, including "simplicity" and "implementation burden" as core criteria for network tariff assessment. There are also experimental studies that show that highly complex tariffs blunt consumer responses (e.g., [157]).

Nothing in everyday consumer spending resembles demand charges. For customers ac-

 $^{^{4}}$ This and the following subsection were adapted from a manuscript that is currently under review for publication [155].

customed to paying flat volumetric charges and unaware of their consumption at any given moment, the concept of being charged based on their maximum demand may be an unacceptably large change. Under demand charges, customers are often not shielded from risk; accidentally running multiple appliances concurrently for a few minutes could result in hundreds of dollars in incurred costs. A demand charge also suffers from the opposite problem: for knowledgeable customers who set a high kW demand early in a billing period, there may be no incentive to manage power demand for the remainder of the period, leading to higher aggregate peak demand.

Thus, while demand charges perform well in our simulations, consumers may resist them, and regulators may be averse to implementing them. We believe subscription charges overcome these problems in several ways.

First, a subscription charge has a cognitive advantage; it gets consumers' attention and makes optimization easier. If a customer must subscribe in advance and is prompted to resubscribe from time to time – e.g., with estimated savings and a default option to continue at the same level – it forces them to think about how they can minimize costs. When the demand charge gets buried in the overall tariff, consumers may not focus their attention on cost minimization.

Second, the subscription structure is similar to popular phone and internet plans, wherein customers pay for a maximum level of service that cannot be exceeded without incurring penalties. A familiarity with these types of plans will help explain the logic of subscription charges and ease the transition to new distribution network tariffs.

Third, a subscription can be implemented in a way that protects consumers from high bills. For example, smart meters can be programmed such that if demand exceeds the subscribed level for some time (e.g., 5 or 15 minutes), the meter is temporarily disconnected. This immediate feedback will help coach customers to not turn on high-power devices simultaneously or to purchase devices that make it possible to program which appliances get turned off first [158]. If meters are tripped frequently due to exceeding the subscription level and customers want to increase their subscription, they may do so for the following billing period.⁵

Fourth, a subscription offers more bill certainty, which is important for customers on tight monthly budgets. Even without perfect foresight, customers can better predict their costs using their ex-ante contracted value compared to an ex-post charge. There are several variants of subscription charges (e.g., "smart subscriptions") that could be used in a transitional period, for example starting out using a soft cap with a small penalty fee calculated as a function of when the subscription value is exceeded, eventually shifting to a hard cap [159].

Fifth and finally, having customers sign up for the maximum demand to which they want access provides information that can help utilities to plan future networks. Even if customers change or exceed their subscriptions, knowing each household's expected peak consumption (which may be significantly lower than their service connection) is a valuable input for distribution resource planning.

These benefits may help explain why several EU member states have already adopted

⁵Frequent adjustment may be penalized, though. In Spain, for example, there is one free adjustment per year. After that, customers are penalized for increasing their subscription level.

per-kW subscription charges. A 2023 ACER report on network tariffs highlights examples from Italy, Portugal, Spain, and Slovenia, among others, and recommends a "gradual move to increasingly power-based distribution tariffs to recover those costs which show correlation with contracted or peak capacity" [131]. Because of extensive retail competition, European countries tend to have electricity bills in which different cost categories (transmission, distribution, supply, etc) are broken out. This makes it easier to implement a subscription charge for just the distribution network portion of the bill. Of course, there is no reason why US states without retail competition could not disaggregate their electricity bills as well. In states and countries with retail competition, the distribution (wires) provider could help educate customers on selecting the proper subscription level.

Despite these advantages, we are not naive about the difficulty of implementing what for US consumers would be a radical departure from the rates they have been accustomed to for decades. Techniques like shadow billing – whereby consumers continue paying under the legacy tariff but see what they would have paid under the new tariff – can be helpful to both protect and educate customers during the transition period. Given the time required to develop new rates and how quickly the issue of rebound peaks may appear, states that have not yet transitioned away from flat volumetric rates should consider consider per-kW subscription charges as a component of retail rate reforms in addition to TOU and critical peak energy pricing.

7.2.3 Direct Load Control

We deliberately chose not to model direct load control, whereby a utility or third-party aggregator modulates the consumption of customer-sited devices to achieve some operational objective (e.g., minimize peak demand). From a technical perspective, we know that direct load control is effective at mitigating the rebound peak problem in distribution grids, demonstrated in the literature [160], [116], [125], [76] and several small-scale pilots over the years [161], [162]. But despite this knowledge, such programs have not been deployed at scale anywhere in the world, even in countries recognized as leaders in utility regulation. We believe three types of barriers – behavioral, technical, and regulatory – would have to be overcome to successfully roll-out centralized load control programs at scale to mitigate local distribution peaks in the US.

First (behavioral), existing load-control programs (e.g., smart thermostat demand response) have proven popular and effective at addressing resource adequacy challenges at the bulk system level. These involve very infrequent load control actions (a few hours per year) that in effect substitute for dramatic price spikes. Load control to address peak kW demands within local distribution grids has different characteristics and would require frequent utility control of household demand (perhaps daily). Consumers may not accept such frequent interventions. Of the three barriers, this is likely the easiest to overcome if there are options for consumers to override control signals.

Second (technical), resource adequacy issues at the bulk system level have been monitored in detail by Independent System Operators for decades. Efficiently applying load control for local constraints requires granular visibility of flows at the distribution level. Currently, so-called "smart grids" are in early-stage development. Price signals at the wholesale level indicate when constrained system situations occur and provide information about the value of centralized load control for resource adequacy. Utilities or retailers (in states with retail competition) are subject to these price signals. No such price signals exist currently at the distribution level. Distribution investments are typically made many years in advance and align with a long-run price signal rather than fluctuating spot market prices.

Third (regulatory), under the cost-of-service regulatory model dominant in the US, distribution utilities have an incentive to deploy capital whereas direct load control reduces the need for capital investment. Even if regulators could compel utilities to develop load control platforms, evaluation of program payments to participating customers would not be straightforward. In states with retail choice, effective load control would require the distribution utility to provide price signals, which could change daily, to retailers, who could then directly pass them through to their consumers. The development of such a regulatory framework and pricing mechanism would be highly complex and could take years to implement.

Given the complexity of these barriers, we have little reason to believe that all three will be solved quickly. The potential rapid growth in EV and CCHP penetration points to a need for simple alternatives like tariff design to reduce unnecessary investment in distribution grids. Whereas it is relatively simple for a regulatory agency to compel a utility to implement a new default rate structure, mandating load control to mitigate local grid issues – which has no historical precedent – would likely take more time and face significant opposition. Over time, the addition of targeted load control or randomization on top of well-designed tariffs could provide even more savings for customers. This is especially true in regions where CCHP-driven peaks can be mitigated with a small number of control events per year (like in our Massachusetts case study), indicated by the steepness of the load duration curve in Figure 6.11. We believe load control and tariff design are complements rather than substitutes, and we intend to address this interaction in future work.

7.2.4 Hybrid Heating Systems and Gas Network Cost Recovery

In the Massachusetts case study, CCHPs operating at low efficiency at extremely cold outdoor temperatures produce large increases in peak demand; at 50% electrification, the feeder's annual peak is between 85% and 136% higher than at 0% electrification, depending on the tariff scenario. CCHP-driven peaks have also been studied at the bulk system level in New England; Khorramfar et al. project that electrifying space heating in all residential buildings in New England will increase winter peak demand by 184% [72].

There is an opportunity for hybrid heating systems, which use both a heat pump and supplemental furnace or boiler, to mitigate this issue. Massachusetts currently requires homeowners to remove their legacy heating systems when claiming an upfront purchase rebate for a CCHP system [143]. While this approach may achieve the highest possible reduction in emissions for each home by eliminating direct fossil fuel combustion, our results indicate that Massachusetts' policy may cause unintended grid issues beyond low adoption levels that ultimately slow the pace of decarbonization. Instead, homes could use their legacy heating systems to meet heating demand during the coldest hours of the year when CCHP efficiency is lowest. Integrated thermostats can be programmed to run either the CCHP, backup system, or both in tandem depending on operating costs or outdoor temperature. Figure 7.1 illustrates this potential; it includes the annual peak results for two tariff scenarios



Figure 7.1: Annual aggregate peak demand in the Massachusetts case study for the *eflat nflat* and *etou ndem12* tariff scenarios with and without gas backup.

in Massachusetts case study with no backup ($eflat_nflat$ and $etou_ndem12$) and with gas backup ($eflat_nflat_gbu$ and $etou_ndem12_gbu$). We only allow the gas backup system to run at hours when the outdoor air temperature is below -15C (5F), which results in 2% of total annual heating demand met by gas. For the $etou_ndem12$ tariff (where the effect is slightly more significant), adding gas backup reduces annual peak demand by 14% at 50% electrification and by 17% at 100% electrification. These reductions would likely be higher if not for a "rebound" phenomenon we observe during hours where the outdoor air temperature crosses the -15C threshold. During these hours, CCHP demand behaves in much the same way as correlated EV demand responding to a change in the energy charge; this could be mitigated by randomizing the temperature threshold.

Khorramfar et al. [72] find that when CCHP systems are sized for cooling load and paired with supplemental fossil heating, region-wide residential peak demand is halved compared to an electric resistance backup scenario. Notably, building envelope improvements also have a strong downward effect on peak heating loads, pointing to the importance of pairing weatherization with heat pump installations to avoid oversizing systems and achieve operating cost reductions.

Even with hybrid systems, in states like Massachusetts where natural gas is the dominant heating fuel, heating electrification implies a sharp reduction in utilization of the natural gas network. Under existing retail natural gas rates in many states, a large portion of network costs are recovered through a flat volumetric charge (similar to electricity). As more customers adopt CCHPs and use significantly less or no gas, the sunk costs of the gas network will not change. This reduction in gas consumption implies an increase in the perunit cost to recover the revenue requirement. As gas tariffs increase, there will be a stronger incentive to adopt heat pumps, reducing gas revenue further. This cycle – sometimes called a "utility death spiral" – poses a cost recovery challenge for gas distribution utilities and an affordability challenge for customers who continue to use gas for heating.

This problem is most acute if heating electrification proceeds "randomly" and gas utilities must maintain the entire existing network throughout the transition to electric heating. If instead regulators and utilities plan to decommission parts of the distribution gas network (e.g., entire neighborhoods), it is possible to achieve operating cost savings and avoid the need for network reinforcements to replace old and/or leak-prone pipes; Lechowicz et al. [163] demonstrate that a centrally-planned, "network-aware" transition scenario achieves 55% higher carbon reductions than an unmanaged transition under the same budget. Even with hybrid heating, it may be more cost-effective to rely on other forms of backup heat (e.g., trucked propane or fuel oil with on-site storage tanks). In Walsh and Bloomberg [164], the authors calculate that a non-pipeline hybrid system achieves a lower combined revenue requirement (inclusive of the electricity and gas networks) compared to a hybrid system using the existing gas network. Yet today in many states gas utilities have accelerated mains replacement rates for safety purposes [165], which will create more stranded costs down the road if there is high CCHP penetration.

In Massachusetts, this challenge is particularly relevant after a recent regulatory order (DPU 20-80-B) that disallows cost recovery for gas infrastructure without proof that non-gas alternatives were considered. This calls for regulatory innovation to protect households that have not yet electrified. While a detailed study of gas tariff design is outside the scope of this thesis, we can envision at least two potential solutions:

- A new rate designed specifically for heat pump customers who remain connected to the gas network for supplemental heating. This rate could shift a portion of network cost recovery to a fixed charge. In this case, regulators should be careful not to set the fixed charge too high, or else it will create an incentive to disconnect from the gas network entirely (leading to higher CCHP-driven winter peaks).
- A transfer payment from electricity to gas utilities that reflects a portion of expected future revenue from heating electrification. This may be easier to implement in areas where the same parent company provides both gas and electricity.

If backup fossil heating is either not allowed or not cost-effective, energy storage (either installed behind the meter or at the distribution substation) can help mitigate CCHP-driven local and regional peaks. We did not model the use of energy storage for this purpose so cannot comment on how the costs would compare to the costs of network expansion. While it is still early in the CCHP adoption curve in Massachusetts, we believe it is important to consider how to protect vulnerable users of both the gas and electricity networks well before the "utility death spiral" problem arises.

7.2.5 Balancing Stakeholder Interests for Efficiency and Solar

Energy efficiency and residential solar advocates have historically opposed tariff reforms that reduce the volumetric component of the bill (see, e.g., [152], [151]). While efficiency

and solar deployment are both important components of economy-wide decarbonization, we don't believe they should take precedence over core rate design principles (notably cost-reflectiveness) or achieving beneficial electrification.

In many US states, residential solar customers are compensated under the net metering framework, whereby customers are charged for their net consumption (total consumed minus total exported) each bill period. When network costs are recovered volumetrically, consumers can avoid paying for the network despite continuing to rely on it during hours when their solar arrays are not generating. If distribution costs were recovered instead in part by a capacity or fixed charge and net metering continued, solar customers would not receive as much value. Yet there is no rule that solar compensation must be tied to the retail electricity rate. Rather, solar customers could be compensated for exported energy via a separate tariff. If supporting residential solar is a policy priority for states, the solar export tariff could be set to compensate at a higher value than the true avoided cost of the generated electricity.

Statewide efficiency programs typically target reductions in kWh energy consumption to alleviate energy burdens and reduce emissions. Yet as electricity generation becomes cleaner, kWh reduction will become a less effective proxy for carbon reduction. In 2020, Sacramento Municipal Utility District (SMUD), a municipal utility, became the fist utility to change their efficiency metric from kWh reduction to "avoided carbon." This reflects the fact that the most cost-effective way to reduce carbon emissions may actually require *increasing* electricity consumption (i.e., through electrification). Furthermore, several studies (e.g., [166], [167]) demonstrate that upfront rebates, federal appliance standards, and building codes may have a larger effect than the volumetric electricity rate on the uptake of energy efficient appliances.

7.3 Limitations of Methodology

In this section, we outline six key limitations in our approach.

First, we do not calibrate the levels of the network charges to actual forward-looking network costs. Rather, the tariff level is calculated using the total revenue requirement (including sunk costs). Our case study focuses on the incentives driven by tariff structures; under our assumption of rational price responsiveness, the absolute magnitude of each tariff does not impact customers' load shapes. Yet absolute magnitudes will affect distributional impacts and incentives to electrify. Furthermore, we use a simplified network model where investments grow linearly with annual peak demand. In reality, network investments would likely occur in a lumpy manner. However, a linear curve roughly approximates a utility's *average* upgrade costs over many distribution feeders [168].

Second, we treat energy prices as exogenous and do not recalibrate the timing of tariff windows (e.g., on-peak and off-peak) as load evolves. For the former limitation, energy TOU rates are typically a proxy for wholesale prices in the day-ahead or real-time markets (in ISO territories). While direct impacts of residential electrification on the wholesale market are expected to be limited initially, at high levels of adoption it becomes important to consider how the correlated response of flexible loads like EVs and CCHPs would impact wholesale electricity prices and thus affect TOU energy rates. We will investigate this topic in future work. For the latter limitation, while utilities sometimes make small changes to the start and end times of a peak period, TOU windows are remarkably "sticky." Part of the reason is that tariffs typically apply for all customers in a rate class despite considerable diversity in load profiles at different network nodes. For example, Green Mountain Power's residential TOU rate (Rate 11) has used the same on-peak window (12:00 PM – 9:00 PM on weekdays) for 9 years despite a marked shift in region-wide wholesale price patterns in that span (i.e., low mid-day prices from solar PV). However, we recognize that if rebound consumption due to EVs and CCHPs were to become the dominant driver of peak demand in an entire distribution territory, it is likely that the utility would amend the tariff to change the timing of the off-peak period. In any case, flexible loads would continue to react to relative price differences, so the risk of a correlated response would persist even if energy prices and tariff window setting were endogenous to the model.

Third, we assume all EV charging occurs at home with Level 2 chargers, even though one would expect to see a mix of charger types and locations in the future. According to a survey by Dunckley [169], 80% of charging occurs at home, with the remaining 20% at workplace and public stations. Our case study therefore represents a sort of worst-case scenario; non-residential charging would likely help reduce coincident peak demand in the evening by shifting charging activity to daytime hours, as shown by Needell et al. [77]. Furthermore, in urban areas or places with a high proportion of rental properties, it is unlikely that all residents would be able to install dedicated charging stations at home. Our case studies consider suburban residential feeders where homes have off-street parking and sufficient electric panel capacity to install a Level 2 EV charger. An important extension of our research is examining residential charging in urban settings where alternative charging methods exist. In addition, the implications of charging of delivery trucks and other fleets deserves study.

Fourth, we assume that non-EV and non-CCHP load do not respond to price. EVs and CCHPs represent a major source of load; for the Massachusetts case study, at 50% adoption EVs and CCHPs account for 39% of annual feeder-wide consumption. However, we would expect actual consumers to shift other loads (e.g., dishwashers and washing machines) to reduce costs, enabled by automation capabilities. This would mitigate peak demand growth and allow non-electrified households to reduce their network costs, improving distributional outcomes. However, we believe it is useful to consider the purely price-unresponsive case, as it isolates the distributional impacts of electrification under each tariff scenario and adheres to the perception among some consumer protection advocates that vulnerable households do not have the flexibilities in their daily schedules to respond to prices.⁶ This is the same reason we chose to apply the alternative tariffs to the entire population rather than only electrified households given the trend towards default opt-out tariffs that achieve high participation rates.

Fifth, we associate only a single EV with each house. It is likely that some multi-vehicle households will purchase a second EV before all households have their first. We will address this possibility in future work, although we expect that capacity-based tariffs will perform similarly given that each home's demand or subscription cost is based on its aggregate

⁶Faruqui and Sergici challenged this perception in a survey of 15 TOU pilot programs, which found that low-income households responded at the same or higher rates compared to medium- and high-income households [66]. When subjected to time-varying rates, low-income customers reduced their peak demand by 17% during critical peak periods; this value more than doubled when customers were given enabling technology like smart thermostats.

demand.

Sixth, our analysis focuses specifically on EV and CCHP adoption. We did not consider the potential impact of other distributed energy resources, like solar and battery storage, which may help mitigate peak demand growth but introduce a host of other complex tradeoffs, especially for compensating exported solar generation.

This chapter provided a discussion of our results and their implications for designing distribution network tariffs in practice. We also discussed limitations of our methodology, which highlight areas for future work. The following chapter offers concluding thoughts and policy recommendations.

Chapter 8

Conclusion and Policy Recommendations

Rapid electrification, through the adoption of electric vehicles and cold climate heat pumps, is required to achieve state and federal decarbonization targets. While recent federal legislation and state rebate programs reduce consumers' upfront purchase costs for EVs and CCHPs, there has been comparatively little effort to address operating costs, an important component in the total cost of ownership.

This thesis asks whether distribution network tariff reform can reduce the direct costs of electrification while protecting non-electrified households from higher bills resulting from new network investments. The central question was formulated in response to two prevailing trends: 1) Electric utilities are projecting steeply rising distribution network costs (which will be shared among all grid users) to accommodate expected load growth due to electrification; 2) Flat volumetric tariffs, which are nearly ubiquitious today in the US, result in high operating costs for CCHPs and EVs, a headwind to adoption, particularly for CCHPs.

Recent initiatives to reform network tariffs in the US do not adequately balance stakeholder interests and may create unintended negative effects. For example, some states (e.g., Missouri, Colorado, and Michigan) have implemented default TOU rates that collect both network and energy costs via a single charge. While these initiatives better reflect the underlying cost of electricity consumption and are a step in the right direction, they tend to focus on reflecting bulk system costs and ignore distribution system constraints. Without a per-kW capacity charge, these tariffs are prone to correlated responses from automated EV and CCHP consumption, which may yield higher network investments than under the status quo. California has proposed introducing an income-based fixed charge to shift recovery of sunk network costs away from the volumetric charge in an explicit attempt to improve the economics of electrification. While this approach adheres partially to the "efficient ideal" tariff design, it has faced intense backlash, offers no opportunity for customers to manage a large portion of their bill, and on its own fails to deter future network investments.

Much of the existing research on network tariff reform either focuses on cost changes at a single snapshot (imposing a condition of revenue neutrality) or ignores the potential for network investment. Among studies that address how to mitigate the impacts of electrification on distribution grids, many consider only direct load control. While load control will likely play an important role in the future, it would be a missed opportunity to ignore how tariff design, which can be deployed relatively quickly, can address the dual challenges of rapid decarbonization and energy affordability.

In contrast to prior work, this thesis considers the impact of tariff design on network investment in the long term, how rates affect incentives for customers to electrify, and the distributional impacts on households that have not yet purchased EVs and CCHP systems. We also consider network and energy tariffs that are designed independently, which is unique among a small body of similar research (see Table 3.1). We conducted two realistic case studies to test distribution network tariff performance in the face of electrification in two distinct geographies. Households react to tariff prices independently without any coordination among households or centralized control. We evaluated network tariff designs using three key metrics: annual peak demand, levelized electricity costs for EV and CCHP consumption, and change in network costs for non-electrified households.

We found that in mild climates where heating electrification is already prevalent, correlated EV charging causes new local network peaks as early as 25% adoption, a threshold we may have already crossed in some neighborhoods. In cold climates, CCHP consumption during extremely cold hours is the dominant source of peak demand growth, triggering network investments as early as 20% adoption. Our results indicate a tradeoff between reducing costs for electrified households and containing cost impacts for non-electrified households. We found that per-kWh distribution network tariffs, which several states have adopted as the default, lead to high electrification costs and lack a price signal to limit aggregated demand peaks, leading to high network investments. While fixed network charges reduce the cost of electrification, they shift costs from EV and CCHP households to non-electrified households and again lack a mechanism to mitigate peak demand. We find that per-kW capacity tariffs, which incentivize households to limit their maximum demand, are effective at reducing EV-driven peaks and operating costs for both CCHPs and EVs. When paired with a flat per-kWh charge to collect a portion of the network revenue requirement, per-kW tariffs can deliver savings for all grid users. Among capacity tariffs, demand charges perform well but may be difficult to implement. In light of these findings, we believe a subscription charge is a pragmatic compromise that performs relatively well on all dimensions.

Our results suggest that there is an opportunity, through distribution network tariff reform, to improve the economics of electrification while lowering energy burdens for households that have not yet electrified: a true win-win. We believe this outcome – beneficial electrification – should be the goal of any tariff reform effort in order to simultaneously address the urgent challenges of decarbonization and affordability. We hope this finding counteracts the narrative that a shift away from flat volumetric tariffs will necessarily harm vulnerable grid users who cannot shift the timing of electricity consumption. When viewed in the short term, any rate reform will create winners and losers because the same revenue requirement must be recovered. However, when we consider the long-term impacts, well-designed rates can reduce the need for network investment and accelerate electrification, delivering benefits for both electrified and non-electrified households. We note that the extent to which this beneficial electrification can occur is highly sensitive to the cost of network upgrades in each utility's service territory.

Finally, our results also indicate the limitations of even relatively granular distribution network tariffs coupled with TOU energy charges. We observe EV-driven rebound peaks (in response to a drop in the per-kWh price) even under demand and subscription charges. And when considering the concurrent electrification of home heating and transportation, we see a steep rise in aggregate peak demand under all tariffs. If we relax some of our tariff design principles (i.e., simplicity, non-discrimination, and existing widespread implementation), there are several alternatives to the distribution network tariffs tested in our case study. These include randomized TOU windows, daily capacity charges, curtailable contracts [170], auctions for network capacity [171], and dynamic price setting based on equilibria estimations [172]. Yet moving from the flat volumetric tariffs ubiquitous today in the US to these advanced and untested approaches would likely face significant resistance from utilities and regulators. In the near term, simple distribution network tariff designs like those implemented in Europe can be effective at limiting peak demand growth. They also offer a bridge to the complementary measures, including direct load control, which will become necessary as we reach higher levels of electrification and more volatile wholesale prices.

One of the key insights we hope to impart to regulators is that the prevailing heuristics for tariff design in the US are poorly suited for a world with highly automated demand from EVs and CCHPs. Rate design experts generally agree that a peak-to-off-peak differential of at least 2:1 is needed motivate customers to enroll in opt-in TOU rates and respond manually to price signals [34]. But when load shifting is tantamount to configuring an automated schedule in a mobile app one time, *any* price signal (even a modest one) may be sufficient, especially for opt-out whole-home rates. Without the constraint of a high daily peak/off-peak differential, regulators will have more freedom to allocate costs between different types of charges and set tariff prices in a way that improves electrification economics and protects vulnerable customers.

In summary, we recommend that PUCs separate the design of distribution network tariffs from energy tariffs and introduce a per-kW subscription charge to collect a portion of the distribution network revenue requirement. The subscription charge should be either mandatory or opt-out, and paired with a per-kWh charge to balance stakeholder interests. This combination has the potential to 1) mitigate the need for local network upgrades, especially at high electrification levels, 2) reduce costs for non-electrified households, and 3) provide low levelized costs for households adopting EVs and CCHP systems. Table 8.1 summarizes our key insights and recommendations to PUCs.

Table 8.1: Summary of key insights and policy recommendations

Key Insights

- Per-kWh TOU rates, which are gaining popularity among US regulators as replacements for flat volumetric tariffs, may perform worse than the status quo from the perspective of network investments. Under these rates, correlated EV- and CCHP-driven peaks could exceed historical peaks in the near future.
- Capacity-based distribution network tariffs are effective at mitigating these "rebound" peaks while reducing levelized electrification costs and protecting non-participating customers from high bills.
- While rate reform necessarily creates winners and losers in the short term, in the long term all grid users can benefit due to the rate-reducing potential of electrification. The extent to which this "beneficial electrification" can occur depends both on the tariff design and the long-run marginal cost of upgrading the network.
- Different service territories call for different tariff solutions. Historically, electricity cost has not not been a barrier to EV adoption. In contrast, CCHP adoption depends strongly on electricity cost. For CCHPs, in areas with high volumetric rates and gas heating, reducing the cost of operation is essential to accelerate adoption, which can apply downward rate pressure.
- While flat energy tariffs perform well from a network investment perspective by taking advantage of demand heterogeneity, there are considerations at the generation level that call for time-varying energy tariffs, which can capture both the relative differences and absolute magnitude of hourly wholesale electricity prices [153].
- EVs and CCHP have fundamentally different flexibility characteristics than other loads because of automation; a small price differential may be sufficient to incentivize load shifting.
- While in the near term, policies that mandate the removal of legacy heating systems in order to claim CCHP rebates may be effective at reducing emissions, in the long term these policies will exacerbate peak demand growth. Backup non-electric heating during the coldest handful of hours of the year can reduce network investment at only a small emissions penalty.

Policy Recommendations

- State PUCs should initiate distribution network tariff reform proceedings for each IOU in their jurisdiction to 1) separate energy tariffs from network tariffs and 2) design distribution network tariffs that include a per-kW subscription component and a per-kWh component. Both can be done under the existing cost of service regulatory framework without new legislation.
- In determining how to allocate network costs among charge types (i.e., subscription charge vs per-kWh charge), regulators should consider both incremental network investment costs and the gap between the electricity price and fuel price in each utility's service territory. In areas with low electricity prices or where EV adoption will dominate electrification-related growth, regulators should assign a higher priority to mitigating the impacts of correlated demand (along with other rate design objectives) and a lower priority to reducing the levelized cost of EV- or CCHP-related consumption.
- PUCs should start convening stakeholders to discuss the future of cost recovery for natural gas distribution networks as customers electrify their heating.

References

- U. EPA. "Sources of greenhouse gas emissions | US EPA." (2024), [Online]. Available: https://www.epa.gov/ghgemissions/sources-greenhouse-gas-emissions (visited on 03/28/2024).
- [2] International Energy Agency, "Net zero by 2050 a roadmap for the global energy sector," May 2021. [Online]. Available: https://iea.blob.core.windows.net/assets/deebef5d-0c34-4539-9d0c-10b13d840027/NetZeroby2050-ARoadmapfortheGlobalEnergySector CORR.pdf.
- M. J. S. Zuberi, J. Chambers, and M. K. Patel, "Techno-economic comparison of technology options for deep decarbonization and electrification of residential heating.," *Energy Efficiency*, vol. 14, no. 7, p. 75, Sep. 20, 2021, ISSN: 1570-6478. DOI: 10.1007/s12053-021-09984-7. [Online]. Available: https://doi.org/10.1007/s12053-021-09984-7 (visited on 11/18/2022).
- [4] O. US EPA. "Fast facts on transportation greenhouse gas emissions." (Aug. 25, 2015), [Online]. Available: https://www.epa.gov/greenvehicles/fast-facts-transportation-greenhouse-gas-emissions (visited on 03/28/2024).
- [5] E. J. H. Wilson, P. Munankarmi, B. D. Less, J. L. Reyna, and S. Rothgeb, "Heat pumps for all? distributions of the costs and benefits of residential air-source heat pumps in the united states," *Joule*, vol. 0, no. 0, Feb. 12, 2024, Publisher: Elsevier, ISSN: 2542-4785, 2542-4351. DOI: 10.1016/j.joule.2024.01.022. [Online]. Available: https://www.cell.com/joule/abstract/S2542-4351(24)00049-7 (visited on 03/15/2024).
- [6] M. Miotti, G. J. Supran, E. J. Kim, and J. E. Trancik, "Personal vehicles evaluated against climate change mitigation targets," *Environmental Science & Technology*, vol. 50, no. 20, pp. 10795–10804, Oct. 18, 2016, ISSN: 0013-936X, 1520-5851. DOI: 10.1021/acs.est.6b00177. [Online]. Available: https://pubs.acs.org/doi/10.1021/acs.est.6b00177 (visited on 04/23/2024).
- [7] D. Avery. "These 9 states banned the sale of gas-powered cars: Is yours one of them?" CNET. (2024), [Online]. Available: https://www.cnet.com/roadshow/news/states-banning-new-gas-powered-cars/ (visited on 05/08/2024).

- [8] U. EIA. "RECS: One in three u.s. households faced challenges in paying energy bills in 2015." (2019), [Online]. Available: https://www.eia.gov/consumption/residential/reports/2015/energybills/ (visited on 05/11/2023).
- [9] U. D. of Energy. "Purchasing energy-efficient residential air-source heat pumps," Energy.gov. (2024), [Online]. Available: https://www.energy.gov/femp/purchasingenergy-efficient-residential-air-source-heat-pumps (visited on 03/28/2024).
- [10] L. Aniti, EIA. major utilities' spending on the electric distribution system continues to increase, May 27, 2021. [Online]. Available: https://www.eia.gov/todayinenergy/detail.php?id=48136#:~: text=Annual%20spending%20on%20electricity%20distribution,2000%20after% 20adjusting%20for%20inflation..
- I. N. England, "ISO new england 2023 summer review," 2023. [Online]. Available: https://www.iso-ne.com/static-assets/documents/100003/2023-09-26-egoc-a3.1-iso-ne-2023-summer-review.pdf.
- [12] EIA. "Hourly electricity grid monitor." (2024), [Online]. Available: https: //www.eia.gov/electricity/gridmonitor/dashboard/electric_overview/US48/US48 (visited on 03/28/2024).
- [13] B. Plumer and N. Popovich, "A new surge in power use is threatening u.s. climate goals," *The New York Times*, Mar. 14, 2024, ISSN: 0362-4331. [Online]. Available: https://www.nytimes.com/interactive/2024/03/13/climate/electric-power-climatechange.html (visited on 03/31/2024).
- [14] Eversource, "Electric sector modernization plan: Accelerating a just transition to a reliable and resilient clean energy future," Sep. 2023. [Online]. Available: https://www.mass.gov/doc/gmacesmp-drafteversource/download?_gl=1%2Ako8zfs% 2A_ga%2ANzUwNDI5MDE3LjE2NTA5ODEyMjQ.%2A_ga_SW2TVH2WBY% 2AMTY5MzkyMDE2OS4zNi4xLjE2OTM5MjM1NzQuMC4wLjA..
- [15] N. Grid, "Future grid plan: Empowering massachusetts by building a smarter, stronger, cleaner and more equitable energy future," Sep. 2023. [Online]. Available: https://www.mass.gov/doc/gmacesmp-draftnationalgrid/download?_gl=1%2Adfgptb%2A_ga% 2ANzUwNDI5MDE3LjE2NTA5ODEyMjQ.%2A_ga_SW2TVH2WBY% 2AMTY5MzkyMDE2OS4zNi4xLjE2OTM5MjM1OTcuMC4wLjA..
- [16] M. Boiteux and P. Stasi, "The determination of costs of expansion of an interconnected system of production and distribution of electricity," Marginal Cost Pricing in Theory and Practice, Prentice-Hall Inc., Englewood Cliffs, NJ, 1964.
- [17] Kevala, "Electrification impacts study part 1: Bottom-up load forecasting and system-level electrification impacts cost estimates," California Public Utilities Commission, Energy Division, May 9, 2023. [Online]. Available: https://docs.cpuc.ca.gov/PublishedDocs/Efile/G000/M508/K423/508423247.PDF.

- [18] P. A. Office, "Distribution grid electrification model study and report," California Public Utilities Commission, San Francisco, CA, Aug. 2023. [Online]. Available: https://www.publicadvocates.cpuc.ca.gov/-/media/cal-advocateswebsite/files/press-room/reports-and-analyses/230824-public-advocatesdistribution-grid-electrification-model-study-and-report.pdf.
- [19] L. W. Davis, "The economic determinants of heat pump adoption," National Bureau of Economic Research, 2023. [Online]. Available: https://www.nber.org/papers/w31344 (visited on 10/29/2023).
- [20] J. B. Bushnell, E. Muehlegger, and D. S. Rapson, "Energy prices and electric vehicle adoption," Energy Institute at Haas, Energy Institute WP 326, Mar. 2022. [Online]. Available: https://haas.berkeley.edu/wp-content/uploads/WP326.pdf.
- [21] M. J. Coren, "Is it cheaper to refuel your EV battery or gas tank? we did the math in all 50 states.," Washington Post, Aug. 8, 2023. [Online]. Available: https://www.washingtonpost.com/climate-environment/interactive/2023/electricvehicle-charging-price-vs-gasoline/ (visited on 05/07/2024).
- [22] AFDC. "Average fuel economy by major vehicle category." (2024), [Online]. Available: https://afdc.energy.gov/data/10310 (visited on 04/23/2024).
- [23] AAA. "AAA gas prices." (Apr. 23, 2024), [Online]. Available: https://gasprices.aaa.com/ (visited on 04/23/2024).
- [24] EIA. "Average price of electricity to ultimate customers by end-use sector." (Apr. 23, 2024), [Online]. Available: https://www.eia.gov/electricity/monthly/epm_table_grapher.php (visited on 04/23/2024).
- [25] G. Data, "New construction and the future of gas in massachusetts," ZeroCarbonMA, Feb. 2024. [Online]. Available: https://static1.squarespace.com/ static/62e94d16a77e1e191eafe4ae/t/65c509b847ec46459341d78d/1707411896890/ New+Construction+and+the+Future+of+Gas+in+MA+-+2.7.24.pdf (visited on 04/23/2024).
- [26] H. Michaels and R. Nachtrieb, "Transitioning to heat pumps in cold climates: A systems dynamics analysis," in *Proceedings of the 2022 ACEEE Summer Study on Energy Efficiency in Buildings*, 2022.
- [27] S. Sergici, A. Ramakrishnan, G. Kavlak, A. Bigelow, and M. Diehl, "Heat pump-friendly cost-based rate designs," Energy Systems Integration Group, Jan. 2023. [Online]. Available: https://www.esig.energy/wp-content/uploads/2023/01/Heat-Pump%E2%80%93Friendly-Cost-Based-Rate-Designs.pdf.
- J. R. McBride, "Clean heat at what cost? economic optimization of residential space heating in massachusetts," Accepted: 2022-08-29T15:59:53Z, Thesis, Massachusetts Institute of Technology, May 2022. [Online]. Available: https://dspace.mit.edu/handle/1721.1/144617 (visited on 12/11/2022).

- [29] S. of Maine. "Governor mills, u.s. department of energy announce \$10 million federal grant to support energy efficiency and workforce investments in rural maine | office of governor janet t. mills." (Feb. 27, 2024), [Online]. Available: http://www.maine.gov/governor/mills/news/governor-mills-us-department-energyannounce-10-million-federal-grant-support-energy (visited on 03/28/2024).
- [30] EIA. "Prices and factors affecting prices." (2023), [Online]. Available: https: //www.eia.gov/energyexplained/electricity/prices-and-factors-affecting-prices.php (visited on 05/07/2024).
- [31] EIA. "Investor-owned utilities served 72% of u.s. electricity customers in 2017 u.s. energy information administration (EIA)." (Aug. 2019), [Online]. Available: https://www.eia.gov/todayinenergy/detail.php?id=40913 (visited on 05/01/2024).
- [32] I. J. Pérez-Arriaga, J. D. Jenkins, and C. Batlle, "A regulatory framework for an evolving electricity sector: Highlights of the MIT utility of the future study," *Economics of Energy & Environmental Policy*, vol. 6, no. 1, pp. 71–92, 2017, Publisher: JSTOR.
- [33] T. Kavulla, "Why is the smart grid so dumb? missing incentives in regulatory policy for an active demand side in the electricity sector," Energy Systems Integration Group, Jan. 2023. [Online]. Available: https://www.esig.energy/wpcontent/uploads/2023/01/Why-Is-the-Smart-Grid-So-Dumb-Missing-Incentives-in-Regulatory-Policy-for-an-Active-Demand-Side-in-the-Electricity-Sector.pdf.
- [34] A. Faruqui and S. Tang, "Equitable rate design; a survey of best practices," Clean Energy Leadership Institute, Sep. 20, 2023.
- [35] T. Schittekatte, D. Mallapragada, P. L. Joskow, and R. Schmalensee. "Reforming retail electricity rates to facilitate economy-wide decarbonization: Joule." (2023), [Online]. Available: https://www.cell.com/joule/pdf/S2542-4351(23)00124-1.pdf (visited on 05/02/2023).
- [36] U. DOE, "Pathways to commercial liftoff: Virtual power plants," US Department of Energy, Sep. 2023. [Online]. Available: https://liftoff.energy.gov/wp-content/uploads/2023/09/20230911-Pathways-to-Commercial-Liftoff-Virtual-Power-Plants update.pdf.
- [37] J. P. Carvallo and L. C. Schwartz, "The use of price-based demand response as a resource in electricity system planning," 2023. [Online]. Available: https://escholarship.org/uc/item/3mp147d8 (visited on 04/11/2024).
- [38] J. D. Jenkins, E. N. Mayfield, E. D. Larson, S. W. Pacala, and C. Greig, "Mission net-zero america: The nation-building path to a prosperous, net-zero emissions economy," *Joule*, vol. 5, no. 11, pp. 2755–2761, 2021, Publisher: Elsevier. [Online]. Available: https://www.cell.com/joule/pdf/S2542-4351(21)00493-1.pdf (visited on 05/10/2024).

- [39] C. Murphy, T. Mai, Y. Sun, P. Jadun, M. Muratori, B. Nelson, and R. Jones, "Electrification futures study: Scenarios of power system evolution and infrastructure development for the united states," National Renewable Energy Lab.(NREL), Golden, CO (United States); Northern ..., 2021. [Online]. Available: https://www.osti.gov/biblio/1762438 (visited on 05/10/2024).
- [40] S. Shenstone-Harris, P. Rhodes, J. Frost, E. Carlson, E. Borden, C. Lane, and M. Whited, "Electric vehicles are driving rates down for all customers," Synapse Energy Economics, Jan. 2024. [Online]. Available: https://www.synapseenergy.com/sites/default/files/Electric%20Vehicles%20Are%20Driving%20Rates% 20Down%20for%20All%20Customer%20Update%20jan%202024.pdf.
- [41] B. Wyman, The Special Law Governing Public Service Corporations, II vols. Baker, Voorhis, & Co., 1911, vol. I.
- [42] S. Borenstein and J. Bushnell, "The US electricity industry after 20 years of restructuring," Annual Review of Economics, vol. 7, pp. 437–463, Volume 7, 2015 Aug. 2, 2015, Publisher: Annual Reviews, ISSN: 1941-1383, 1941-1391. DOI: 10.1146/annurev-economics-080614-115630. [Online]. Available: https://www.annualreviews.org/content/journals/10.1146/annurev-economics-080614-115630 (visited on 05/08/2024).
- [43] RESA. "US energy choice overview: State-by-state analysis." (Apr. 18, 2022),
 [Online]. Available: https://www.resausa.org/energy-by-state/ (visited on 12/11/2023).
- [44] L. Kiesling, "Incumbent vertical market power, experimentation, and institutional design in the deregulating electricity industry," *The Independent Review*, vol. 19, no. 2, pp. 239–264, 2014, Publisher: Independent Institute, ISSN: 1086-1653.
 [Online]. Available: https://www.jstor.org/stable/24563277 (visited on 02/26/2023).
- [45] P. L. Joskow and R. Schmalensee, "Chapter 3: Cost of service regulation of electricity distribution services in the u.s.," in *Handbook of Electricity Regulation*, Edward Elgar Press (forthcoming).
- [46] C. F. Phillips Jr, "The regulation of public utilities," PUR, Arlington Va. USA, quoting Troxell, "Economics of Public Utilities, 1993. [Online]. Available: https://publicsde.regie-energie.qc.ca/projets/401/DocPrj/R-3998-2017-B-0044-Argu-Argu-2017 02 16.PDF (visited on 05/01/2024).
- [47] W. Troesken, "Regime change and corruption. a history of public utility regulation," in Corruption and reform: lessons from America's economic history, University of Chicago Press, 2006, pp. 259–282. [Online]. Available: https://www.nber.org/system/files/chapters/c9986/c9986.pdf (visited on 03/29/2024).
- [48] D. Tietjen, "Tariff development i: The basic ratemaking process," 2016. [Online]. Available: https://pubs.naruc.org/pub.cfm?id=538E730E-2354-D714-51A6-5B621A9534CB#:~:text=3-,What%20is%20%E2%80%9CCost%20of%20Service% E2%80%9D%20Regulation%3F,and%20earn%20a%20reasonable%20return (visited on 04/17/2023).

- [49] J. C. Bonbright, Principles of Public Utility Rates. Columbia University Press, Dec. 31, 1961, ISBN: 978-0-231-88861-5. DOI: 10.7312/bonb92418. [Online]. Available: https://www.degruyter.com/document/doi/10.7312/bonb92418/html (visited on 05/07/2024).
- [50] D. P. Brown and D. E. M. Sappington, *Designing Incentive Regulation in the Electricity Sector*. MIT Center for Energy and Environmental Policy Research, 2023. [Online]. Available: https://ceepr.mit.edu/wp-content/uploads/2023/11/MIT-CEEPR-WP-2023-20.pdf (visited on 05/02/2024).
- [51] P. L. Joskow, "Regulation of natural monopoly," in *Handbook of law and economics*, ser. Handbooks in economics, bk. 27, A. M. Polinsky and S. Shavell, Eds., Amsterdam ; Elsevier, 2007, ISBN: 978-0-444-51235-2.
- [52] J.-J. Laffont and J. Tirole, A theory of incentives in procurement and regulation. MIT press, 1993. [Online]. Available: https://books.google.com/books?hl=en&lr=&id=4iH4Z2wbNqAC&oi=fnd&pg= PA1&dq=A+Theory+of+Incentives+in+Procurement+and+Regulation&ots= qtnke_yRkG&sig=ShNikziwmZ69XoKklF0cxnYeDWo (visited on 05/02/2024).
- [53] G. J. Stigler, "The theory of economic regulation," *The Bell journal of economics* and management science, pp. 3–21, 1971, Publisher: JSTOR.
- [54] D. Pomerantz, "Getting politics out of utility bills: How policymakers can protect customers from being forced to fund utilities' political machines," Energy and Policy Institute, Jan. 2023.
- [55] J. E. Moreno, "FirstEnergy ex-c.e.o. and 2 others are indicted in bribery scandal," *The New York Times*, Feb. 12, 2024, ISSN: 0362-4331. [Online]. Available: https://www.nytimes.com/2024/02/12/business/firstenergy-indictment-ohiobribery.html (visited on 05/07/2024).
- [56] H. Averch and L. L. Johnson, "Behavior of the firm under regulatory constraint," *The American Economic Review*, vol. 52, no. 5, pp. 1052–1069, 1962, Publisher: JSTOR.
- [57] K. D. Werner and S. Jarvis, "Rate of return regulation revisited," 2022.
- [58] S. Cicala, "Imperfect markets versus imperfect regulation in US electricity generation," American Economic Review, vol. 112, no. 2, pp. 409–441, 2022, Publisher: American Economic Association 2014 Broadway, Suite 305, Nashville, TN 37203. [Online]. Available: https://www.aeaweb.org/articles?id=10.1257/aer.20172034 (visited on 05/08/2024).
- [59] W. J. Boyes, "AN EMPIRICAL EXAMINATION OF THE AVERCH-JOHNSON EFFECT," *Economic Inquiry*, vol. 14, no. 1, pp. 25–35, Mar. 1976, ISSN: 00952583, 14657295. DOI: 10.1111/j.1465-7295.1976.tb00374.x. [Online]. Available: https://onlinelibrary.wiley.com/doi/10.1111/j.1465-7295.1976.tb00374.x (visited on 05/08/2024).

- [60] J.-M. Glachant, H. Khalfallah, Y. Perez, V. Rious, and M. Saguan, "Implementing incentive regulation and regulatory alignment with resource bounded regulators," Working Paper, 2012, Accepted: 2012-07-11T13:37:50Z ISSN: 1028-3625. [Online]. Available: https://cadmus.eui.eu/handle/1814/22734 (visited on 04/11/2023).
- [61] P. L. Joskow, "Incentive regulation in theory and practice: Electricity distribution and transmission networks," *Economic regulation and its reform: What have we learned?*, pp. 291–344, 2014, Publisher: University of Chicago Press. [Online]. Available: https://www.nber.org/system/files/chapters/c12566/c12566.pdf (visited on 05/02/2024).
- [62] J. P. Carvallo, P. H. Larsen, A. H. Sanstad, and C. A. Goldman, "Long term load forecasting accuracy in electric utility integrated resource planning," *Energy Policy*, vol. 119, pp. 410–422, Aug. 1, 2018, ISSN: 0301-4215. DOI: 10.1016/j.enpol.2018.04.060. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421518302751 (visited on 02/17/2023).
- [63] M. Dyson and A. Engel. "The billion-dollar costs of forecasting electricity demand," RMI. (Oct. 23, 2017), [Online]. Available: https://rmi.org/billion-dollar-costs-forecasting-electricity-demand/ (visited on 05/08/2024).
- [64] E. Halper, "Amid explosive demand, america is running out of power," The Washington Post, Mar. 7, 2024. [Online]. Available: https://www.washingtonpost.com/business/2024/03/07/ai-data-centers-power/.
- [65] J. Burkhardt, K. Gillingham, and P. K. Kopalle, Experimental evidence on the effect of information and pricing on residential electricity consumption, Feb. 2019. DOI: 10.3386/w25576. [Online]. Available: https://www.nber.org/papers/w25576 (visited on 09/21/2023).
- [66] A. Faruqui and S. Sergici, "Household response to dynamic pricing of electricity: A survey of 15 experiments," *Journal of regulatory Economics*, vol. 38, no. 2, pp. 193–225, 2010, Publisher: Springer.
- [67] M. Waite and V. Modi, "Electricity load implications of space heating decarbonization pathways," *Joule*, vol. 4, no. 2, pp. 376–394, Feb. 19, 2020, ISSN: 2542-4351. DOI: 10.1016/j.joule.2019.11.011. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2542435119305781 (visited on 10/23/2022).
- [68] P. Vaishnav and A. M. Fatimah, "The environmental consequences of electrifying space heating," *Environmental Science & Technology*, vol. 54, no. 16, pp. 9814–9823, Aug. 18, 2020, Publisher: American Chemical Society, ISSN: 0013-936X. DOI: 10.1021/acs.est.0c02705. [Online]. Available: https://doi.org/10.1021/acs.est.0c02705 (visited on 10/23/2022).

- [69] P. R. White, J. D. Rhodes, E. J. H. Wilson, and M. E. Webber, "Quantifying the impact of residential space heating electrification on the texas electric grid," *Applied Energy*, vol. 298, p. 117 113, Sep. 15, 2021, ISSN: 0306-2619. DOI: 10.1016/j.apenergy.2021.117113. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261921005559 (visited on 10/24/2022).
- [70] C. Protopapadaki and D. Saelens, "Heat pump and PV impact on residential low-voltage distribution grids as a function of building and district properties," *Applied Energy*, vol. 192, pp. 268–281, Apr. 15, 2017, ISSN: 0306-2619. DOI: 10.1016/j.apenergy.2016.11.103. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261916317329 (visited on 11/18/2022).
- [71] Z. E. Lee and K. M. Zhang, "Unintended consequences of smart thermostats in the transition to electrified heating," *Applied Energy*, vol. 322, p. 119384, 2022, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261922007243 (visited on 03/07/2024).
- [72] R. Khorramfar, M. Santoni-Colvin, S. Amin, L. K. Norford, A. Botterud, and D. Mallapragada, *Cost-effective planning of decarbonized power-gas infrastructure to meet the challenges of heating electrification*, Feb. 28, 2024. arXiv: 2308.16814[cs,eess]. [Online]. Available: http://arxiv.org/abs/2308.16814 (visited on 03/15/2024).
- [73] M. Muratori, "Impact of uncoordinated plug-in electric vehicle charging on residential power demand," *Nature Energy*, vol. 3, no. 3, pp. 193–201, Mar. 2018, Number: 3 Publisher: Nature Publishing Group, ISSN: 2058-7546. DOI: 10.1038/s41560-017-0074-z. [Online]. Available: https://www.nature.com/articles/s41560-017-0074-z (visited on 06/09/2023).
- [74] S. Huang and D. Infield, "The impact of domestic plug-in hybrid electric vehicles on power distribution system loads," in 2010 International conference on power system technology, IEEE, 2010, pp. 1–7. [Online]. Available: https://ieeexplore.ieee.org/abstract/document/5666513/ (visited on 04/25/2024).
- [75] S. Elmallah, A. M. Brockway, and D. Callaway, "Can distribution grid infrastructure accommodate residential electrification and electric vehicle adoption in northern california?" *Environmental Research: Infrastructure and Sustainability*, vol. 2, no. 4, p. 045 005, 2022, Publisher: IOP Publishing. [Online]. Available: https://iopscience.iop.org/article/10.1088/2634-4505/ac949c/meta (visited on 04/24/2024).
- [76] J. Coignard, P. MacDougall, F. Stadtmueller, and E. Vrettos, "Will electric vehicles drive distribution grid upgrades?: The case of california," *IEEE Electrification Magazine*, vol. 7, no. 2, pp. 46–56, Jun. 2019, Conference Name: IEEE Electrification Magazine, ISSN: 2325-5889. DOI: 10.1109/MELE.2019.2908794.
 [Online]. Available: https://ieeexplore.ieee.org/document/8732007/figures#figures (visited on 05/10/2024).

- [77] Z. Needell, W. Wei, and J. E. Trancik, "Strategies for beneficial electric vehicle charging to reduce peak electricity demand and store solar energy," *Cell Reports Physical Science*, vol. 4, no. 3, 2023, Publisher: Elsevier.
- [78] C. Gschwendtner, C. Knoeri, and A. Stephan, "The impact of plug-in behavior on the spatial-temporal flexibility of electric vehicle charging load," *Sustainable Cities* and Society, vol. 88, p. 104263, 2023, Publisher: Elsevier.
- [79] S. Powell, G. V. Cezar, L. Min, I. M. L. Azevedo, and R. Rajagopal, "Charging infrastructure access and operation to reduce the grid impacts of deep electric vehicle adoption," *Nature Energy*, vol. 7, no. 10, pp. 932–945, Oct. 2022, Number: 10 Publisher: Nature Publishing Group, ISSN: 2058-7546. DOI: 10.1038/s41560-022-01105-7. [Online]. Available: https://www.nature.com/articles/s41560-022-01105-7 (visited on 06/08/2023).
- [80] M. Nour, J. P. Chaves-Ávila, G. Magdy, and Á. Sánchez-Miralles, "Review of positive and negative impacts of electric vehicles charging on electric power systems," *Energies*, vol. 13, no. 18, p. 4675, Jan. 2020, Number: 18 Publisher: Multidisciplinary Digital Publishing Institute, ISSN: 1996-1073. DOI: 10.3390/en13184675. [Online]. Available: https://www.mdpi.com/1996-1073/13/18/4675 (visited on 03/15/2024).
- [81] S. A. Steinbach and M. J. Blaschke, "Another source of inequity? how grid reinforcement costs differ by the income of EV user groups," 2023.
- [82] T. Yuksel and J. J. Michalek, "Effects of regional temperature on electric vehicle efficiency, range, and emissions in the united states," *Environmental Science & Technology*, vol. 49, no. 6, pp. 3974–3980, Mar. 17, 2015, ISSN: 0013-936X, 1520-5851. DOI: 10.1021/es505621s. [Online]. Available: https://pubs.acs.org/doi/10.1021/es505621s (visited on 10/28/2023).
- [83] E. Wood, B. Borlaug, M. Moniot, D.-Y. Lee, Y. Ge, F. Yang, and Z. Liu, "The 2030 national charging network: Estimating US light-duty demand for electric vehicle charging infrastructure," National Renewable Energy Laboratory (NREL), Golden, CO (United States), 2023. [Online]. Available: https://www.osti.gov/biblio/1988020 (visited on 04/24/2024).
- [84] M. B. Anwar, M. Muratori, P. Jadun, E. Hale, B. Bush, P. Denholm, O. Ma, and K. Podkaminer, "Assessing the value of electric vehicle managed charging: A review of methodologies and results," *Energy & Environmental Science*, vol. 15, no. 2, pp. 466–498, 2022, Publisher: Royal Society of Chemistry. [Online]. Available: https://pubs.rsc.org/en/content/articlehtml/2021/xx/d1ee02206g (visited on 04/24/2024).
- [85] V. Aravinthan and W. Jewell, "Controlled electric vehicle charging for mitigating impacts on distribution assets," *IEEE Transactions on Smart Grid*, vol. 6, no. 2, pp. 999–1009, 2015, Publisher: IEEE. [Online]. Available: https://ieeexplore.ieee.org/abstract/document/7024906/ (visited on 04/24/2024).

- [86] A. Dubey and S. Santoso, "Electric vehicle charging on residential distribution systems: Impacts and mitigations," *IEEE Access*, vol. 3, pp. 1871–1893, 2015, Publisher: IEEE. [Online]. Available: https://ieeexplore.ieee.org/abstract/document/7264982/ (visited on 04/24/2024).
- [87] A. Visakh and S. Manickavasagam Parvathy, "Energy-cost minimization with dynamic smart charging of electric vehicles and the analysis of its impact on distribution-system operation," *Electrical Engineering*, vol. 104, no. 5, pp. 2805–2817, Oct. 2022, ISSN: 0948-7921, 1432-0487. DOI: 10.1007/s00202-022-01511-w. [Online]. Available: https://link.springer.com/10.1007/s00202-022-01511-w (visited on 04/24/2024).
- [88] D. Patteeuw, G. P. Henze, and L. Helsen, "Comparison of load shifting incentives for low-energy buildings with heat pumps to attain grid flexibility benefits," *Applied Energy*, vol. 167, pp. 80–92, Apr. 1, 2016, ISSN: 0306-2619. DOI: 10.1016/j.apenergy.2016.01.036. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261916300162 (visited on 05/09/2024).
- [89] E. Georges, B. Cornélusse, D. Ernst, V. Lemort, and S. Mathieu, "Residential heat pump as flexible load for direct control service with parametrized duration and rebound effect," *Applied Energy*, vol. 187, pp. 140–153, Feb. 1, 2017, ISSN: 0306-2619. DOI: 10.1016/j.apenergy.2016.11.012. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261916315975 (visited on 04/01/2024).
- [90] B. Baeten, F. Rogiers, and L. Helsen, "Reduction of heat pump induced peak electricity use and required generation capacity through thermal energy storage and demand response," *Applied Energy*, vol. 195, pp. 184–195, Jun. 1, 2017, ISSN: 0306-2619. DOI: 10.1016/j.apenergy.2017.03.055. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261917302854 (visited on 04/01/2024).
- [91] S. Borenstein, "The economics of fixed cost recovery by utilities," The Electricity Journal, vol. 29, no. 7, pp. 5–12, 2016, Publisher: Elsevier. [Online]. Available: https: //www.sciencedirect.com/science/article/pii/S1040619016301130?casa_token=me_ jsVdsF9sAAAAA:ja3zdkPEaTPlBwolj8Rnah58LZFyKLQ8EsBYL9aayZVePDNw_ CxmQx10vI0MUo85-AQsMQ0fvLs (visited on 10/25/2023).
- [92] DOE. "Recovery act: Smart grid investment grant program." (), [Online]. Available: https://www.energy.gov/oe/recovery-act-smart-grid-investment-grant-sgig-program (visited on 05/07/2024).
- [93] A. Cooper, M. Shusterr, and J. Lash, "Electric company smart meter deployments: Foundation for a smart grid (2021 update)," Institute for Electric Innovation, Washington, DC, 2021. [Online]. Available: .%20https://www.%20edisonfoundation.net/-/media/Files/IEI/publications/IEI_Smart_Meter_Report_April_2021.pdf.

- U. EIA. "How many smart meters are installed in the united states, and who has them?" (2022), [Online]. Available: https://www.eia.gov/tools/faqs/faq.php?id=108&t=3#:~:text=How%20many% 20smart%20meters%20are,of%20total%20electric%20meters%20installations (visited on 10/24/2023).
- [95] E. Johnson, R. Beppler, C. Blackburn, B. Staver, M. Brown, and D. Matisoff, "Peak shifting and cross-class subsidization: The impacts of solar PV on changes in electricity costs," *Energy Policy*, vol. 106, pp. 436–444, Jul. 1, 2017, ISSN: 0301-4215. DOI: 10.1016/j.enpol.2017.03.034. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421517301787 (visited on 11/29/2022).
- [96] G. L. Barbose, S. Forrester, E. O'Shaughnessy, and N. R. Darghouth, "Residential solar-adopter income and demographic trends: 2022 update," 2022. [Online]. Available: https://escholarship.org/uc/item/5vd6w51m (visited on 04/01/2024).
- [97] T. Schittekatte, I. Momber, and L. Meeus, "Future-proof tariff design: Recovering sunk grid costs in a world where consumers are pushing back," *Energy economics*, vol. 70, pp. 484–498, 2018, Publisher: Elsevier.
- [98] P. Simshauser, "Distribution network prices and solar PV: Resolving rate instability and wealth transfers through demand tariffs," *Energy Economics*, vol. 54, pp. 108–122, 2016, Publisher: Elsevier.
- [99] Q. Hoarau and Y. Perez, "Network tariff design with prosumers and electromobility: Who wins, who loses?" *Energy Economics*, vol. 83, pp. 26–39, Sep. 1, 2019, ISSN: 0140-9883. DOI: 10.1016/j.eneco.2019.05.009. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0140988319301549 (visited on 12/11/2022).
- [100] S. Borenstein, M. Fowlie, and J. Sallee, "Designing electricity rates for an equitable energy transition," *Energy Institute Working Paper*, vol. 314, 2021.
- [101] R. Passey, N. Haghdadi, A. Bruce, and I. MacGill, "Designing more cost reflective electricity network tariffs with demand charges," *Energy Policy*, vol. 109, pp. 642–649, 2017, Publisher: Elsevier.
- [102] M. Hofmann and K. B. Lindberg, "Evidence of households' demand flexibility in response to variable hourly electricity prices-results from a comprehensive field experiment in norway," *Energy Policy*, vol. 184, p. 113 821, 2024, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421523004068 (visited on 03/07/2024).
- [103] G. Strbac and J. Mutale, "Framework and methodology for pricing of distribution networks with distributed generation," *Cent. Distrib. Gener. Sustain. Electr. Energy*, 2005.

- [104] L. Hirth, T. Khanna, and O. Ruhnau, "How aggregate electricity demand responds to hourly wholesale price fluctuations," 2023, Publisher: Kiel, Hamburg: ZBW-Leibniz Information Centre for Economics. [Online]. Available: https://www.econstor.eu/handle/10419/272048 (visited on 04/24/2024).
- [105] S. P. Burger, C. R. Knittel, I. J. Pérez-Arriaga, I. Schneider, and F. Vom Scheidt, "The efficiency and distributional effects of alternative residential electricity rate designs," *The Energy Journal*, vol. 41, no. 1, 2020, Publisher: International Association for Energy Economics.
- [106] C. Batlle, P. Mastropietro, and P. Rodilla, "Redesigning residual cost allocation in electricity tariffs: A proposal to balance efficiency, equity and cost recovery," *Renewable Energy*, vol. 155, pp. 257–266, 2020, Publisher: Elsevier.
- [107] P. Simshauser and D. Downer, "On the inequity of flat-rate electricity tariffs," *The Energy Journal*, vol. 37, no. 3, pp. 199–230, Jul. 2016, ISSN: 0195-6574, 1944-9089.
 DOI: 10.5547/01956574.37.3.psim. [Online]. Available: https://journals.sagepub.com/doi/10.5547/01956574.37.3.psim (visited on 04/24/2024).
- [108] C.-M. Bergaentzlé, P. A. Gunkel, M. Ansarin, Y. Ghiassi-Farrokhfal, and H. K. Jacobsen, *Electricity grid tariffs for electrification in households: Bridging the* gap between cross-subsidies and fairness, Oct. 18, 2022. DOI: 10.48550/arXiv.2210.09690. arXiv: 2210.09690[econ,q-fin]. [Online]. Available: http://arxiv.org/abs/2210.09690 (visited on 12/08/2022).
- [109] J. Kahn-Lang, C. Miller, A. Satchwell, and E. Present, "Building efficiency, electrification, and distributed solar PV bill savings under time-based retail rate designs," 2024. [Online]. Available: https://escholarship.org/uc/item/5mt4n753 (visited on 04/25/2024).
- [110] S. Küfeoğlu and M. G. Pollitt, "The impact of PVs and EVs on domestic electricity network charges: A case study from great britain," *Energy Policy*, vol. 127, pp. 412–424, Apr. 1, 2019, ISSN: 0301-4215. DOI: 10.1016/j.enpol.2018.12.012.
 [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421518308085 (visited on 12/11/2022).
- [111] N. Morell-Dameto, J. P. Chaves-Ávila, T. Gómez San Román, and T. Schittekatte, "Forward-looking dynamic network charges for real-world electricity systems: A slovenian case study," *Energy Economics*, vol. 125, p. 106 866, Sep. 1, 2023, ISSN: 0140-9883. DOI: 10.1016/j.eneco.2023.106866. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S014098832300364X (visited on 09/23/2023).
- [112] P. A. Gunkel, C.-M. Bergaentzlé, D. Keles, F. Scheller, and H. K. Jacobsen, Grid tariff designs coping with the challenges of electrification and their socio-economic impacts, Apr. 11, 2023. DOI: 10.48550/arXiv.2210.03514. arXiv: 2210.03514[econ,q-fin]. [Online]. Available: http://arxiv.org/abs/2210.03514 (visited on 06/13/2023).

- [113] N. Govaerts, K. Bruninx, H. Le Cadre, L. Meeus, and E. Delarue, "Signaling future or historical distribution grid costs via tariffs? a welfare analysis of long-run incremental cost pricing," *Utilities Policy*, vol. 82, p. 101 537, 2023, Publisher: Elsevier.
- [114] I. Abdelmotteleb, T. Gómez, J. P. C. Ávila, and J. Reneses, "Designing efficient distribution network charges in the context of active customers," *Applied Energy*, vol. 210, pp. 815–826, 2018, Publisher: Elsevier.
- [115] T. Schittekatte, "Distribution network tariff design for behind-the-meter: Balancing efficiency and fairness," in *Behind and Beyond the Meter*, Elsevier, 2020, pp. 341–359.
- [116] A. Nouicer, L. Meeus, and E. Delarue, "The economics of demand-side flexibility in distribution grids," *The Energy Journal*, vol. 44, no. 1, 2023, Publisher: International Association for Energy Economics.
- [117] M.-L. Arlt, D. Chassin, C. Rivetta, and J. Sweeney, "Impact of real-time pricing and residential load automation on distribution systems," *Energy Policy*, vol. 184, p. 113 906, Jan. 1, 2024, ISSN: 0301-4215. DOI: 10.1016/j.enpol.2023.113906. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421523004913 (visited on 04/01/2024).
- [118] C. Bergaentzlé, I. G. Jensen, K. Skytte, and O. J. Olsen, "Electricity grid tariffs as a tool for flexible energy systems: A danish case study," *Energy Policy*, vol. 126, pp. 12–21, 2019, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421518307444 (visited on 04/24/2024).
- [119] R. J. Hennig, D. Ribó-Pérez, L. J. de Vries, and S. H. Tindemans, "What is a good distribution network tariff?—developing indicators for performance assessment," *Applied Energy*, vol. 318, p. 119 186, 2022, Publisher: Elsevier.
- [120] A. Satchwell, J. P. Carvallo, P. Cappers, J. Milford, and H. Eshraghi, "Quantifying the financial impacts of electric vehicles on utility ratepayers and shareholders," 2023. [Online]. Available: https://escholarship.org/uc/item/6dz355d9 (visited on 04/02/2024).
- Y. Li and A. Jenn, "Impact of electric vehicle charging demand on power distribution grid congestion," *Proceedings of the National Academy of Sciences*, vol. 121, no. 18, e2317599121, Apr. 30, 2024, Publisher: Proceedings of the National Academy of Sciences. DOI: 10.1073/pnas.2317599121. [Online]. Available: https://www.pnas.org/doi/abs/10.1073/pnas.2317599121 (visited on 05/10/2024).
- [122] H. Trabish. "Hawai'i leads the way on advanced rate design with default time-of-use rates, fixed charge innovations," Utility Dive. (May 9, 2023), [Online]. Available: https://www.utilitydive.com/news/hawaii-advanced-rate-design-default-time-ofuse-fixed-charge-innovations/648994/ (visited on 09/23/2023).

- M. Fowlie, C. Wolfram, P. Baylis, C. A. Spurlock, A. Todd-Blick, and P. Cappers, "Default effects and follow-on behaviour: Evidence from an electricity pricing program," *The Review of Economic Studies*, vol. 88, no. 6, pp. 2886–2934, 2021, Publisher: Oxford University Press. [Online]. Available: https://academic.oup.com/restud/article-abstract/88/6/2886/6222168 (visited on 04/29/2024).
- [124] Y. L. Qiu, Y. D. Wang, H. Iseki, X. Shen, B. Xing, and H. Zhang, "Empirical grid impact of in-home electric vehicle charging differs from predictions," *Resource and Energy Economics*, vol. 67, p. 101 275, Feb. 1, 2022, ISSN: 0928-7655. DOI: 10.1016/j.reseneeco.2021.101275. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0928765521000609 (visited on 04/01/2024).
- [125] M. Muratori and G. Rizzoni, "Residential demand response: Dynamic energy management and time-varying electricity pricing," *IEEE Transactions on Power* systems, vol. 31, no. 2, pp. 1108–1117, 2015, Publisher: IEEE.
- [126] Y. Li and P. A. Crossley, "Monte carlo study on impact of electric vehicles and heat pumps on LV feeder voltages," in 12th IET International Conference on Developments in Power System Protection (DPSP 2014), Mar. 2014, pp. 1–6. DOI: 10.1049/cp.2014.0095.
- [127] S. E. P. Alliance, "Residential electric vehicle rates that work," Smart Electric Power Alliance, Nov. 2019. [Online]. Available: https://sepapower.org/resource/residential-electric-vehicle-time-varying-rates-thatwork-attributes-that-increase-enrollment/thank-you/.
- [128] G. M. Power, Integrated resource plan, Dec. 2021. [Online]. Available: https://greenmountainpower.com/wp-content/uploads/2021/12/2021-Integrated-Resource-Plan.pdf.
- S. D. Wong, S. A. Shaheen, E. Martin, and R. Uyeki, "Do incentives make a difference? understanding smart charging program adoption for electric vehicles," *Transportation Research Part C: Emerging Technologies*, vol. 151, p. 104123, 2023, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0968090X23001122 (visited on 04/24/2024).
- [130] M. R. Bailey, D. P. Brown, B. C. Shaffer, and F. A. Wolak, "Show me the money! incentives and nudges to shift electric vehicle charge timing," National Bureau of Economic Research, 2023.
- [131] ACER, "Report on electricity transmission and distribution tariff methodologies in europe," European Union Agency for the Cooperation of Energy Regulators, Jan. 2023. [Online]. Available: https://www.acer.europa.eu/Publications/ACER_ electricity_network_tariff_report.pdf.
- [132] A. B. Eriksen and V. Mook, "Proposed changes to the design of network tariffs for low voltage grid users in norway," The Norwegian Energy Regulatory Authority (RME), Feb. 2020. [Online]. Available: https://publikasjoner.nve.no/rme_rapport/2020/rme_rapport2020_02.pdf.
- [133] OFGEM, "Access and forward-looking charges significant code review: Final decision," May 3, 2022. [Online]. Available: https://www.ofgem.gov.uk/sites/default/files/2022-05/Access%20SCR%20-%20Final%20Decision.pdf.
- E. Wilson, A. Parker, A. Fontanini, et al., "End-use load profiles for the u.s. building stock: Methodology and results of model calibration, validation, and uncertainty quantification," NREL/TP-5500-80889, 1854582, MainId:78667, Mar. 7, 2022, NREL/TP-5500-80889, 1854582, MainId:78667. DOI: 10.2172/1854582. [Online]. Available: https://www.osti.gov/servlets/purl/1854582/ (visited on 10/10/2022).
- [135] F. H. Administration, "2017 national household travel survey," U.S. Department of Transportation, Washington, DC, 2017. [Online]. Available: https://nhts.ornl.gov.
- [136] W. Wei, S. Ramakrishnan, Z. A. Needell, and J. E. Trancik, "Personal vehicle electrification and charging solutions for high-energy days," *Nature Energy*, vol. 6, no. 1, pp. 105–114, Jan. 2021, Number: 1 Publisher: Nature Publishing Group, ISSN: 2058-7546. DOI: 10.1038/s41560-020-00752-y. [Online]. Available: https://www.nature.com/articles/s41560-020-00752-y (visited on 10/11/2022).
- [137] C. M. Domingo, T. G. San Roman, A. Sanchez-Miralles, J. P. P. Gonzalez, and A. C. Martinez, "A reference network model for large-scale distribution planning with automatic street map generation," *IEEE Transactions on Power Systems*, vol. 26, no. 1, pp. 190–197, 2010, Publisher: IEEE. [Online]. Available: https://ieeexplore.ieee.org/abstract/document/5504171/?casa_token=u_ YxXi36wdsAAAAA:eU4uc7pTgLv9pOsHXfCbo6K_H_ 0ObafHT10bfeRyC1gOvmZFNG1vYBOdBCqRd2oDK3K8S2hz4g (visited on 04/25/2024).
- [138] A. W. L. Yee, "The impact of distributed energy resources (DERs) in integrated gas-electricity energy systems," Accepted: 2018-04-27T18:10:46Z Journal Abbreviation: Impact of DERs in integrated gas-electricity energy systems, Thesis, Massachusetts Institute of Technology, 2017. [Online]. Available: https://dspace.mit.edu/handle/1721.1/115030 (visited on 12/11/2022).
- T. L. Lee, "Implications of heating electrification on distribution networks and distributed energy resources," Accepted: 2022-08-29T16:24:49Z, Thesis, Massachusetts Institute of Technology, May 2022. [Online]. Available: https://dspace.mit.edu/handle/1721.1/144976 (visited on 12/11/2022).
- [140] M. Virginia Olano. "Chart: Which states have the most heat pumps?" Canary Media. (Aug. 12, 2022), [Online]. Available: https://www.canarymedia.com/articles/heat-pumps/chart-which-states-have-themost-heat-pumps (visited on 04/20/2024).

- [141] Y. Min and E. Mayfield, "Rooftop solar, electric vehicle, and heat pump adoption in rural areas in the united states," *Energy Research & Social Science*, vol. 105, p. 103 292, 2023, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2214629623003523 (visited on 04/10/2024).
- [142] M.-k. Cha, C. L. Struthers, M. A. Brown, S. Kale, and O. Chapman, "Toward residential decarbonization: Analyzing social-psychological drivers of household co-adoption of rooftop solar, electric vehicles, and efficient HVAC systems in georgia, u.s.," *Renewable Energy*, vol. 226, p. 120382, May 1, 2024, ISSN: 0960-1481. DOI: 10.1016/j.renene.2024.120382. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0960148124004476 (visited on 05/09/2024).
- [143] M. Save. "Air source heat pumps," Mass Save. (2024), [Online]. Available: https://www.masssave.com/en/residential/rebates-and-incentives/heating-andcooling/heat-pumps/air-source-heat-pumps (visited on 05/06/2024).
- [144] N. Veilleux, "Residential ccASHP building electrification study," Cadmus, Jun. 3, 2022. [Online]. Available: https://cadmusgroup.com/wp-content/uploads/2022/06/Residential-ccASHP-Building-Electrification-Study_Cadmus_Final_060322_Public.pdf.
- [145] A. Reuther, J. Kepner, C. Byun, et al., "Interactive supercomputing on 40,000 cores for machine learning and data analysis," in 2018 IEEE High Performance extreme Computing Conference (HPEC), ISSN: 2377-6943, Sep. 2018, pp. 1–6. DOI: 10.1109/HPEC.2018.8547629. [Online]. Available: https://ieeexplore.ieee.org/document/8547629 (visited on 05/06/2024).
- [146] J. H. Lee, S. J. Hardman, and G. Tal, "Who is buying electric vehicles in california? characterising early adopter heterogeneity and forecasting market diffusion," *Energy Research & Social Science*, vol. 55, pp. 218–226, Sep. 1, 2019, ISSN: 2214-6296. DOI: 10.1016/j.erss.2019.05.011. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2214629618312258 (visited on 09/23/2023).
- [147] D. Boraks. "Electric vehicle registrations grew 54% in NC last year, but still account for tiny share of cars," WFAE 90.7 - Charlotte's NPR News Source. Section: Energy & Environment. (Feb. 24, 2023), [Online]. Available: https://www.wfae.org/energy-environment/2023-02-24/electric-vehicleregistrations-grew-54-in-nc-last-year-but-still-account-for-tiny-share-of-cars (visited on 04/18/2024).
- [148] A. Joroff, "Energy justice: What it means and how to integrate it into state regulation of electricity markets," *Envtl. L. Rep. News & Analysis*, vol. 47, p. 10927, 2017, Publisher: HeinOnline. [Online]. Available: https://heinonline.org/hol-cgi-bin/get_pdf.cgi?handle=hein.journals/elrna47§ion=133 (visited on 05/10/2024).

- [149] A. Pamidimukkala, S. Kermanshachi, J. M. Rosenberger, and G. Hladik, "Evaluation of barriers to electric vehicle adoption: A study of technological, environmental, financial, and infrastructure factors," *Transportation Research Interdisciplinary Perspectives*, vol. 22, p. 100 962, 2023, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2590198223002099 (visited on 04/11/2024).
- [150] U. EPA, "EPA report on the social cost of greenhouse gases: Estimates incorporating recent scientific advances," US Environmental Protection Agency, Washington, DC, 2022.
- [151] A. Faruqui. "The income graduated fixed charges in california will harm customers with low electric bills," PV Magazine USA. (May 8, 2023), [Online]. Available: https://pv-magazine-usa.com/2023/05/08/the-income-graduated-fixed-charges-incalifornia-will-harm-customers-with-low-electric-bills/ (visited on 11/10/2023).
- [152] J. Engstrom and L. Deehan. "New proposed income-based fixed charges in california would devalue energy conservation," CALPIRG. (Aug. 14, 2023), [Online]. Available: https://pirg.org/california/resources/new-proposed-income-based-fixed-charges-incalifornia-would-devalue-energy-conservation/ (visited on 11/10/2023).
- [153] T. Schittekatte, D. Mallapragada, P. L. Joskow, and R. Schmalensee, "Electricity retail rate design in a decarbonizing economy: An analysis of time-of-use and critical peak pricing," *The Energy Journal*, vol. 45, no. 3, Jul. 1, 2024, ISSN: 01956574. DOI: 10.5547/01956574.45.3.tsch. [Online]. Available: https://www.iaee.org/ej/ejexec/ej45-3-Schittekatte-exsum.pdf (visited on 09/20/2023).
- [154] B. Borlaug, S. Salisbury, M. Gerdes, and M. Muratori, "Levelized cost of charging electric vehicles in the united states," *Joule*, vol. 4, no. 7, pp. 1470–1485, 2020, Publisher: Elsevier.
- [155] G. Turk, T. Schittekatte, P. Dueñas Martínez, P. L. Joskow, and R. Schmalensee, "Designing distribution network tariffs under increased residential end-user electrification: Can the US learn something from europe?" MIT Center for Energy and Environmental Policy Research, 2024. [Online]. Available: https: //ceepr.mit.edu/wp-content/uploads/2024/01/MIT-CEEPR-WP-2024-02.pdf.
- [156] J. Tomich. "Missouri overhauls electric rates, raising rewards and risks for customers," E&E News by POLITICO. (Jul. 12, 2023), [Online]. Available: https://www.eenews.net/articles/missouri-overhauls-electric-rates-raising-rewardsand-risks-for-customers/ (visited on 04/11/2024).
- [157] G. D. Jacobsen and J. I. Stewart, "How do consumers respond to price complexity? experimental evidence from the power sector," *Journal of Environmental Economics and Management*, vol. 116, p. 102 716, Oct. 1, 2022, ISSN: 0095-0696. DOI: 10.1016/j.jeem.2022.102716. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0095069622000754 (visited on 10/31/2023).

- [158] Y. Mou, A. Papavasiliou, and P. Chevalier, "Application of priority service pricing for mobilizing residential demand response in belgium," in 2017 14th International Conference on the European Energy Market (EEM), IEEE, 2017, pp. 1–5. [Online]. Available: https://ieeexplore.ieee.org/abstract/document/7981860/?casa_token= 7rjp8qP5vBcAAAAA:8WL5sYVaTv-OA5O0o4plLLJrHd7GNhAuskZQYmbjN0QmCrIXkVhzUzMj1riz5fqhL9j5oaro0es (visited on 10/25/2023).
- [159] DNV-GL, "Effective and cost reflective distribution tariffs," 2019. [Online]. Available: https://www.dnv.com/Publications/effective-and-cost-reflectivedistribution-tariffs-162913.
- [160] S. Backe, G. Kara, and A. Tomasgard, "Comparing individual and coordinated demand response with dynamic and static power grid tariffs," *Energy*, vol. 201, p. 117619, Jun. 15, 2020, ISSN: 0360-5442. DOI: 10.1016/j.energy.2020.117619.
 [Online]. Available: https://www.sciencedirect.com/science/article/pii/S036054422030726X (visited on 10/30/2022).
- [161] J. Cross and R. Hartshorn, "My electric avenue: Integrating electric vehicles into the electrical networks," in 6th Hybrid and Electric Vehicles Conference (HEVC 2016), London, UK: Institution of Engineering and Technology, 2016, 12 (6 .)–12 (6 .) ISBN: 978-1-78561-294-7. DOI: 10.1049/cp.2016.0972. [Online]. Available: https://digital-library.theiet.org/content/conferences/10.1049/cp.2016.0972 (visited on 04/10/2024).
- [162] C. A. Smith, "The pecan street project: Developing the electric utility system of the future," 2009. [Online]. Available: https://repositories.lib.utexas.edu/items/9561c4c0-9e94-4edc-9cc5-d2c47e4b63b0 (visited on 04/10/2024).
- [163] A. Lechowicz, N. Bashir, J. Wamburu, M. Hajiesmaili, and P. Shenoy, "Equitable network-aware decarbonization of residential heating at city scale," in *Proceedings of* the 14th ACM International Conference on Future Energy Systems, Orlando FL USA: ACM, Jun. 20, 2023, pp. 1–13. DOI: 10.1145/3575813.3576870. [Online]. Available: https://dl.acm.org/doi/10.1145/3575813.3576870 (visited on 11/07/2023).
- [164] M. J. Walsh and M. E. Bloomberg, "The future of gas in new york state," Building Decarbonization Coalition, Mar. 2023. [Online]. Available: https://static1.squarespace.com/static/62e94d16a77e1e191eafe4ae/t/ 65e7c3623bd89b76cf1f9f43/1709687651602/BDC-The-Future-of-Gas-in-NYS+%282%29.pdf.
- [165] O. of Energy Policy and S. Analysis, "Natural gas infrastructure modernization programs at local distribution companies: Key issues and considerations," U.S. Department of Energy, Washington, DC, Jan. 2017. [Online]. Available: https://www.energy.gov/policy/articles/natural-gas-infrastructure-modernizationprograms-local-distribution-companies-key.

- [166] S. Datta and M. Filippini, "Analysing the impact of ENERGY STAR rebate policies in the US," *Energy Efficiency*, vol. 9, no. 3, pp. 677–698, Jun. 1, 2016, ISSN: 1570-6478. DOI: 10.1007/s12053-015-9386-7. [Online]. Available: https://doi.org/10.1007/s12053-015-9386-7 (visited on 05/07/2024).
- [167] S. Meyers, J. E. McMahon, M. McNeil, and X. Liu, "Impacts of US federal energy efficiency standards for residential appliances," *Energy*, vol. 28, no. 8, pp. 755–767, Jun. 1, 2003, ISSN: 0360-5442. DOI: 10.1016/S0360-5442(02)00188-3. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0360544202001883 (visited on 05/07/2024).
- [168] N. Govaerts, K. Bruninx, H. Le Cadre, L. Meeus, and E. Delarue, "Forward-looking distribution network charges considering lumpy investments," *Journal of Regulatory Economics*, vol. 59, no. 3, pp. 280–302, Jun. 2021, ISSN: 0922-680X, 1573-0468. DOI: 10.1007/s11149-021-09428-1. [Online]. Available: https://link.springer.com/10.1007/s11149-021-09428-1 (visited on 05/07/2024).
- [169] J. Dunckley, "Electric vehicle driving, charging, and load shape analysis: A deep dive into where, when, and how much salt river project (SRP) electric vehicle customers charge," *Electric Power Research Institute*, 2018.
- [170] R. J. Hennig, L. J. de Vries, and S. H. Tindemans, "Risk vs. restriction—an investigation of capacity-limitation based congestion management in electric distribution grids," *Energy Policy*, vol. 186, p. 113 976, 2024, Publisher: Elsevier.
 [Online]. Available: https://www.sciencedirect.com/science/article/pii/S030142152300561X (visited on 05/08/2024).
- [171] N. Morell-Dameto, J. P. Chaves-Ávila, T. G. San Román, P. Dueñas-Martínez, and T. Schittekatte, "Network tariff design with flexible customers: Ex-post pricing and a local network capacity market for customer response coordination," *Energy Policy*, vol. 184, p. 113 907, 2024, Publisher: Elsevier. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0301421523004925 (visited on 02/23/2024).
- [172] S. Huang, Q. Wu, H. Zhao, and C. Li, "Distributed optimization-based dynamic tariff for congestion management in distribution networks," *IEEE Transactions on Smart Grid*, vol. 10, no. 1, pp. 184–192, Jan. 2019, Conference Name: IEEE Transactions on Smart Grid, ISSN: 1949-3061. DOI: 10.1109/TSG.2017.2735998.